Security Testing of Human-interactive

Systems



Gelei Deng

College of Computing & Data Science

A thesis submitted to the Nanyang Technological University in partial fulfillment of the requirements for the degree of Doctor of Philosophy

2024

Statement of Originality

I hereby certify that the work embodied in this thesis is the result of original research, is free of plagiarised materials, and has not been submitted for a higher degree to any other University or Institution.

29-Dec-2023

Date

Gelei Deng

Supervisor Declaration Statement

I have reviewed the content and presentation style of this thesis and declare it is free of plagiargism and of sufficient grammatical clarity to be examined. To the best of my knowledge, the research and writing are those of the candidate except as acknowledged in the Author Attribution Statement. I confirm that the investigations were conducted in accord with the ethics policies and integrity standards of Nanyang Technological University and that the research data are presented honestly and without prejudice.

26-Jan-2024

Date

Asst Prof Tianwei Zhang

Authorship Attribution Statement

Please select one of the following; *delete as appropriate:

This thesis contains material from 4 papers published in the follow-

ing peer-reviewed journal(s) / from papers accepted at conferences

in which I am listed as an author.

Please amend the typical statements below to suit your circumstances if (B) is selected.

Chapter 3 is published as Gelei Deng, Yuan Zhou, Yuan Xu, Tianwei Zhang, Yang Liu, An investigation of byzantine threats in multi-robot systems, in Proceedings of the 24th International Symposium on Research in Attacks, Intrusions and Defenses (RAID 21). https://dl.acm.org/doi/abs/10.1145/3471621.3471867

The contributions of the co-authors are as follows:

- I was the lead author, I wrote the manuscript drafts and conducted all experiments.
- Prof Tianwei Zhang guided the research work research direction and revised the manuscript drafts. Prof Yang Liu supervised the research process.
- The rest of the co-authors participates in the discussion.

Chapter 4 is published as Gelei Deng, Guowen Xu, Yuan Zhou, Tianwei Zhang, Yang Liu. On the (In) Security of Secure ROS2. In Proceedings of the 2022 ACM SIGSAC Conference on Computer and Communications Security (ACM CCS 2022). https://doi.org/10.1145/3548606.3560681

The contributions of the co-authors are as follows:

- I was the lead author, I wrote the manuscript drafts and conducted all experiments.
- Prof Tianwei Zhang guided the research work research direction and revised the manuscript drafts. Prof Yang Liu supervised the research process.
- Dr. Guowen Xu helped with the design of the cryptographic defense solution; Dr. Yuan Zhou helped with the model checking methodology design.

Chapter 5 is published as Gelei Deng, Zhiyi Zhang, Yuekang Li, Yi Liu, Tianwei Zhang, Yang Liu, Guo Yu, Dongjin Wang. NAUTILUS: Automated RESTful API Vulnerability Detection. In 32nd USENIX Security Symposium (USENIX Security

23). https://www.usenix.org/conference/usenixsecurity23/presentation/ deng-gelei

The contributions of the co-authors are as follows:

- I was the lead author, I wrote the manuscript drafts and conducted all experiments.
- Prof Tianwei Zhang guided the research work research direction and revised the manuscript drafts. Prof Yang Liu supervised the research process.
- Dr. Yuekang Li helped with the paper writing as the corresponding author. Mr. Yi Liu assisted the experiments.
- The rest of the authors are industrial collaboration partners, providing industriallevel test environments and real-world vulnerabilities.

Chapter 6 is accepted as Gelei Deng, Yi Liu, Yuekang Li, Kailong Wang, Ying Zhang, Zefeng Li, Haoyu Wang, Tianwei Zhang, Yang Liu. MASTERKEY: Automated Jailbreak Across Multiple Large Language Model Chatbots. In The Network and Distributed System Security Symposium (NDSS) 2024. https://www. usenix.org/conference/usenixsecurity23/presentation/deng-gelei

The contributions of the co-authors are as follows:

- I was the lead author, I wrote the manuscript drafts and conducted experiments. Mr. Yi Liu was the co-first author, wrote the manuscript and designed the evaluation methodology. Ms. Zefeng Li helped with the experiments.
- Prof Tianwei Zhang guided the research work research direction and revised the manuscript drafts. Prof Yang Liu supervised the research process.
- Prof. Yuekang Li, Prof Kailong Wang, Dr. Ying Zhang and Prof Haoyu Wang helped with the paper writing.

ITU NTU NTU NTU NTU NTU NTU NTU NTU UN ITU Gelei Deng J NT ITU NTU NTU NTU NTU NTU NTU NTU

26-Dec-2023

Date

Gelei Deng

Acknowledgements

As I reflect on the fourth year of my PhD journey, I am filled with a deep sense of gratitude and fulfillment. This journey has been one of significant progress, learning, and the forging of lasting friendships. It has been an honor and a privilege to work alongside individuals who have greatly contributed to shaping my doctoral voyage and my personal growth.

First and foremost, I extend my heartfelt thanks to my supervisor, Prof. Tianwei Zhang, for his invaluable guidance, unwavering support, and profound wisdom. His mentorship has been a guiding light throughout my research journey. When we started as a small group, the direction of our work was not clear, but his dedication to excellence and commitment to nurturing my potential have been instrumental in navigating these uncertainties. Over four years, we have made significant strides, carving out a unique path in our field. I am equally grateful to my advisor, Prof. Yang Liu, whose insights and expertise have greatly enriched my research experience. His perspectives and constructive feedback have been pivotal in broadening my horizons, not just academically but also professionally. Prof. Liu's guidance has been crucial in helping me identify and seize good work opportunities, and for that, I am deeply thankful.

My appreciation also goes to two of my research collaborators, Yi Liu and Dr. Yuekang Li, who has become an assistant processor now. Our close collaboration over the past four years has led to many successful results, and I truely appreciate their kindness and help during the years of collaborations. The insightful discussions and generous assistance enriched this work in countless ways. I extend my heartfelt thanks to research fellows Dr. Guowen Xu, Dr. Yuan Zhou, Dr. Yuan Xu, and Dr. Hao Ren. Their contributions were indispensable, and their support was crucial in realizing many of my projects. It is always a pleasant to discuss with these kind and intelligent people, not only about the research but also the wisdom of life. I am thankful for the camaraderie and support from fellow Ph.D. students in the same research group - Xingshuo Han, Xiaoxuan Lou, Kangjie Chen, Dikai Liu, Guanlin Li, Qinghao Hu, and Meng Zhang. Their companionship and collaborative spirit have greatly enriched my research experience. Also, I would like to say thank you to the lab administrative assistants, Dr. Shi Ling and Ms. Zolynn, for their meticulous handling of administrative matters, making every process smoother and more manageable.

My deepest appreciation is reserved for my wife, Ms. Xinxue Gu, whose unwavering support and understanding have been my cornerstone. Her love and encouragement have been a continuous source of inspiration throughout these demanding yet exhilarating years.

These four years have been a blend of busyness, exhaustion, excitement, and discovery. The journey of exploring new frontiers in cybersecurity has been challenging yet immensely rewarding. The process of unearthing new knowledge, tackling complex problems, and contributing to the field has been a truly enriching experience. There were sadness, frustration in the past, but the results are promising and exciting.

To all those mentioned and to those whose names may have been unintentionally omitted, I extend my deepest thanks for your indispensable contributions to this chapter of my life. The passion, love, and growth I have experienced throughout this journey have been made possible by each of you, and for that, I am eternally grateful.

Contents

Acknowledgements				
Li	st of	Figures xv	ii	
Li	st of	Tables xi	x	
A	ostra	\mathbf{ct}	1	
1	Intr	oduction	3	
	1.1	Motivation	4	
	1.2	Main Work	6	
	1.3	Contribution of the Thesis	7	
	1.4	List of Materials Related to the Thesis	9	
	1.5	Outline of the Thesis	0	
2	Rela	ated Work 1	1	
	2.1	Human-Interactive Systems	1	
	2.2	Robotic Systems	2	
		2.2.1 Robot Operating System	2	
		2.2.2 Multi-Robot Systems	3	
	2.3	Human Interactive Web Applications	3	
		2.3.1 Web Applications	4	
		2.3.2 AI-driven Chatbots	4	
	2.4	System Testing	4	
		2.4.1 Fuzzing	5	
		2.4.2 Static Analysis	5	
		2.4.3 Formal Verification	6	
ગ	Idor	tifying Byzantino Bisks in Multi-Bobot Systems	7	
0	3.1	Introduction 1	7	
	3.9	Background and Threat Model	.1	
	0.4	3.2.1 Robot Operating System	,1)1	
		3.2.2 Workflow of Robot Tasks	,1)1	
		3.2.2 Wolknow of Hobot Tasks	יד פס	
		5.2.5 Multi-10000 Systems	4	

4	Sec	uritv I	nvestiga	tion of Robot Operating System 2		57
	3.9	Concl	usion		••••	. 56
		3.8.2	Related	Works		. 54
		3.8.1	Counter	measures		. 53
	3.8	Discus	ssions and	l Related Works	••••	. 53
		3.7.2	Evaluat	ion Results		. 52
		3.7.1	Experin	nental Setup		. 51
	3.7	Real-v	world Eva	luation		. 51
			3.6.2.3	Task Forwarding Attack		. 50
			3.6.2.2	Map Merging Poisoning Attack		. 47
			3.6.2.1	Task Assignment Control Attack		. 45
		3.6.2	Evaluat	ion Result		. 45
		3.6.1	Experin	nental Setup		. 43
	3.6	Evalu	ation			. 43
		3.5.3	Attack 3	3: Task Forwarding Manipulation		. 43
			3.5.2.2	Blocking group merging		. 42
			3.5.2.1	False global map generation		. 42
		3.5.2	Attack 2	2: Map Merging Poisoning		. 42
				ration systems		. 41
			3.5.1.2	Fake bidding information in decentralized	l explo-	
				systems		. 41
			3.5.1.1	Fake location or status information in cen	tralized	- 5
		3.5.1	Attack	1: Task Assignment Control		. 40
	3.5	Multi-	-robot Sv	stem Risk Analysis		. 40
		3.4.2	Metrics			. 39
	J.T	3 4 1	Worklos	ad Descriptions		. 35
	34	An M	BS Work	load Suite	• • • •	. 94 35
			3343	Requirement Checking	• • • •	. 55 34
			J.J.4.1 ススオク	Message Mutation Stratogy		. ა∠ აა
		ა.ა.4	Require	Overview	• • • •	
		<u>99</u> 4	3.3.3.2 Domini	Byzantine message extraction		. 30 20
			3.3.3.1	Data-flow graph construction	• • • •	. 30
		3.3.3	Data-flo	w Analysis	• • • •	. 30
			3.3.2.2	Task-specific Requirements	• • • •	. 29
			3.3.2.1	General Requirements		. 28
		3.3.2	Require	ment Specifications		. 28
		3.3.1	Overvie	W		. 27
	3.3	A Me	thodology	to Characterize Byzantine Risks in MRS .	• • • •	. 27
		3.2.4	Threat 1	Model		. 24
			3.2.3.2	Communications in Multi-Robot Systems		. 23
			3.2.3.1	Multi-Robot Workloads		. 22

	4.1	Introduction	8
	4.2	Background	0
		4.2.1 Robot Operating System	0
		4.2.2 Data Distribution Service	1
		4.2.3 DDS Security	2
		4.2.4 Secure ROS2	3
	4.3	Threat Model	4
		4.3.1 System Assumptions	4
		4.3.2 Adversary's Capabilities	4
	4.4	Methodology of Investigating ROS2	5
		4.4.1 ROS2 Abstraction and Modeling	6
		4.4.2 Model Construction	9
		4.4.3 Model Checking	'1
		4.4.4 Discussion	3
	4.5	Security Vulnerabilities in ROS2	'4
		4.5.1 V1: Permission File Replacement	'4
		4.5.2 V2: Outdated Node Service	6
		4.5.3 V3: Default Mis-configuration	7
		4.5.4 V4: Permission File Inference	7
		4.5.5 Discussion	8
	4.6	Vulnerability Exploitation	9
		4.6.1 Simulation Setup	9
		4.6.2 Simulation Evaluation	0
		4.6.3 Physical Evaluation	2
	4.7	A General Defense Solution	2
		4.7.1 Design Rationale	3
		4.7.2 Methodology Description	4
		4.7.3 Implementation	6
		4.7.4 Security Evaluation	7
		4.7.4.1 Theoretical Analysis	7
		4.7.4.2 Formal Verification	8
		4.7.4.3 Empirical Validation	8
		4.7.5 Efficiency Evaluation	9
		4.7.5.1 Performance Evaluation	9
		4.7.5.2 Resource Consumption Evaluation 9	1
	4.8	Related Works	3
	4.9	Conclusion	4
5	Hur	nan-Interactive Testing of RESTful API Service 9	5
	5.1	Introduction	0
	5.2	Background	9
		5.2.1 Key Concepts	9 1
		5.2.2 RESTful API Vulnerabilities	1

	5.3	Runni	ing Example
	5.4	Desig	n $\ldots \ldots \ldots$
		5.4.1	Overview
		5.4.2	Specification Annotation
			5.4.2.1 Operation Annotation $\dots \dots \dots$
			5.4.2.2 Parameter Annotation $\ldots \ldots \ldots$
			5.4.2.3 Annotation Updates
			5.4.2.4 Annotation Primitives
		5.4.3	Exploration Stage
		5.4.4	Exploitation Stage
			5.4.4.1 Payload-based Mutation
			5.4.4.2 Payload and Oracle
	5.5	Imple	mentation and Evaluation
		5.5.1	Evaluation Setup
		5.5.2	Vulnerability Detection (RQ1)
		5.5.3	$Coverage (RQ2) \dots \dots$
		5.5.4	Ablation Study (RQ3) 121
		5.5.5	Real-world Vulnerabilities (RQ4) 122
			5.5.5.1 Vulnerability Identification
	5.6	Relate	ed Work
	5.7	Concl	$usion \dots \dots$
	0.1		
6	Ant	tomate	ed Jailbreaking of Large Language Model Chatbots 129
6	Aut 6.1	t <mark>omat</mark> e Introd	ed Jailbreaking of Large Language Model Chatbots 129 luction 130
6	Aut 6.1 6.2	tomate Introd Backg	ed Jailbreaking of Large Language Model Chatbots 129 luction 130 round 133
6	Aut 6.1 6.2	tomate Introd Backg 6.2.1	ed Jailbreaking of Large Language Model Chatbots 129 luction 130 ground 133 LLM Chatbot 133
6	Aut 6.1 6.2	tomate Introd Backg 6.2.1 6.2.2	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134
6	Aut 6.1 6.2	tomate Introd Backg 6.2.1 6.2.2 6.2.3	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135
6	Aut 6.1 6.2	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135
6	Aut 6.1 6.2 6.3	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135Usage Policy (RQ1)136
6	Aut 6.1 6.2 6.3	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138
6	Aut 6.1 6.2 6.3 6.4	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140
6	Aut 6.1 6.2 6.3 6.4 6.5	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho	ed Jailbreaking of Large Language Model Chatbots129luction130round133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140
6	Aut 6.1 6.2 6.3 6.4 6.5	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140Design Insights141
6	Aut 6.1 6.2 6.3 6.4 6.5	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1 6.5.2	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135npirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY141Design Insights141Time-based LLM Testing143
6	Aut 6.1 6.2 6.3 6.4 6.5	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1 6.5.2 6.5.3	ed Jailbreaking of Large Language Model Chatbots129luction130ground133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140odology of Revealing Jailbreak Defenses141Time-based LLM Testing143Proof of Concept Attack148
6	Aut 6.1 6.2 6.3 6.4 6.5 6.6	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1 6.5.2 6.5.3 Metho	ed Jailbreaking of Large Language Model Chatbots129luction130round133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135npirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140odology of Revealing Jailbreak Defenses141Time-based LLM Testing143Proof of Concept Attack148odology of Crafting Jailbreak Prompts150
6	Aut 6.1 6.2 6.3 6.4 6.5 6.6	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1 6.5.2 6.5.3 Metho 6.6.1	ed Jailbreaking of Large Language Model Chatbots129luction130pround133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135npirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140odology of Revealing Jailbreak Defenses141Time-based LLM Testing143Proof of Concept Attack148odology of Crafting Jailbreak Prompts150Design Rationale150
6	Aut 6.1 6.2 6.3 6.4 6.5 6.6	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1 6.5.2 6.5.3 Metho 6.6.1 6.6.2	ed Jailbreaking of Large Language Model Chatbots129luction130round133LLM Chatbot133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140odology of Revealing Jailbreak Defenses141Time-based LLM Testing143Proof of Concept Attack148odology of Crafting Jailbreak Prompts150Design Rationale150Workflow152
6	Aut 6.1 6.2 6.3 6.4 6.5 6.6	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1 6.5.2 6.5.3 Metho 6.6.1 6.6.2 6.6.3	ed Jailbreaking of Large Language Model Chatbots129luction130round133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140odology of Revealing Jailbreak Defenses141Time-based LLM Testing143Proof of Concept Attack148odology of Crafting Jailbreak Prompts150Design Rationale150Workflow152Dataset Building and Augmentation152
6	Aut 6.1 6.2 6.3 6.4 6.5 6.6	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1 6.5.2 6.5.3 Metho 6.6.1 6.6.2 6.6.3 6.6.4	ed Jailbreaking of Large Language Model Chatbots129luction130round133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135npirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140odology of Revealing Jailbreak Defenses141Time-based LLM Testing143Proof of Concept Attack148odology of Crafting Jailbreak Prompts150Design Rationale152Dataset Building and Augmentation152Continuous Pre-training and Task Tuning154
6	Aut 6.1 6.2 6.3 6.4 6.5 6.6	tomate Introd Backg 6.2.1 6.2.2 6.2.3 An Er 6.3.1 6.3.2 Overv Metho 6.5.1 6.5.2 6.5.3 Metho 6.6.1 6.6.2 6.6.3 6.6.4 6.6.5	ed Jailbreaking of Large Language Model Chatbots129huction130round133LLM Chatbot133LLM Jailbreak134Jailbreak Defense in LLM135mpirical Study135Usage Policy (RQ1)136Jailbreak Effectiveness (RQ2)138iew of MASTERKEY140odology of Revealing Jailbreak Defenses141Time-based LLM Testing143Proof of Concept Attack148odology of Crafting Jailbreak Prompts150Design Rationale152Dataset Building and Augmentation152Continuous Pre-training and Task Tuning154Reward Ranked Fine Tuning154

		$6.7.1 \text{Experiment Setup} \dots \dots$
		6.7.2 Jailbreak Capability (RQ3) 158
		6.7.3 Ablation Study (RQ4)
		6.7.4 Cross-language Compatibility (RQ5) 160
	6.8	Mitigation Recommendation
	6.9	Related Work
		6.9.1 Prompt Engineering and Jailbreaks in LLMs
		6.9.2 LLM Security and Relevant Attacks
		6.9.3 Vulnerability Analysis for Traditional Web Applications 165
	6.10	Conclusion
7	Con	clusion and Future Work 165
	7.1	Conclusion
	7.2	Future Work

Bibliography

171

List of Figures

1.1	Illustration of commonly used testing methodologies	4
3.1	Application pipeline for a typical robot task.	22
3.2	An example of Byzantine threats in a surveillance application.	26
3.3	Workflow of the proposed methodology.	26
3.4	Workflow of $W1$	36
3.5	Workflow of W2	37
3.6	Workflow of W3	38
3.7	Workflow of $W4$	38
3.8	Workflow of W5	39
3.9	Simulated environment for workloads $W1 - W4$	44
3.10	Task assignment control attack against $W1$	45
3.11	Task assignment control attack against $W2$	46
3.12	Map accuracy of $\mathbf{W2}$ with 3 robots under the task assignment con-	
	trol attack.	46
3.13	Task assignment control attack against $W3$	47
3.14	Map merging poisoning attack against W2	48
3.15	Map accuracy of $W2$ with 3 robots under the map merging poison-	
	ing attack.	48
3.16	Map merging poisoning attack against $W3$.	48
3.17	False map generation attack against W4	49
3.18	Blocking group merging attack against W4.	49
3.19	Visualized attack effects for W5.	50
3.20	Task forwarding attack against W5.	50
3.21	Physical environment for real-world evaluation.	51
3.22	Task assignment control attack against W1, W2, and W3 (Physical).	52
3.23	Map merging poisoning attack against W2 and W3 (Physical)	52
41	BOS2 DDS architecture with the DCPS protocol	61
4.2	Methodology Overview	66
4.2 4.3	An example of identifying key functions in an BCL code snippet	00
т.0	(rcl publisher init).	68
4.4	Partial diagram of the CSP $\#$ model for the ROS2 system (node <i>i</i>	00
	publishes/subscribes to topic j)	69
		-

4.5	Unauthorized publishing/subscription through the vulnerabilities of V1 (0) and V2 (2).	75
4.7	Mitigating vulnerabilities with our defense.	87
4.8	Encryption/Decryption time cost under different frequencies and message sizes.	90
4.9	Communication latency of four implementations with various mes- sage lengths and frequencies.	91
4.10	Communication latency of four implementations with one publisher and various numbers of subscribers.	92
4.11	Communication latency of four implementations with various numbers of publisher-subscriber pairs.	92
4.12	Resource consumption of four implementations	93
5.1	The OpenAPI specification of BuddyPress APIs* \ldots	100
5.2	RESTful API vulnerability categories	101
5.3	The running example (CVE-2021-21389)	104
5.4	Overview of NAUTILUS	104
5.5	Annotation updates on the running example*	106
5.6	The mapping between the request types and the required payload- s/oracles	114
5.7	The vulnerabilities and their types uncovered by different tools on the evaluation benchmarks.	118
5.8	The performance of NAUTILUS, NAUTILUS-NO-ANNOTATION, NAU- TILUS-OPERATION-ONLY, and NAUTILUS-PARAMETER-ONLY on	102
5.9	The maximum sequence length, number of auto-generated annota- tions, number of vulnerabilities, and number of annotation-related	123
	vulnerabilities of NAUTILUS.	124
6.1	A jailbreak attack example.	134
6.2	An example of time-based blind SQL injection	143
6.3	Abstraction of an LLM chatbot with jalbreak defense.	143
6.4	The proposed LLM time-based testing strategy.	144
6.5	Overall workflow of our proposed methodology	152
6.6	Average Query Success Rate Across LLM Chatbots for MASTERKEY,	150
	MASTERKEY-INO-FINETUNE, and MASTERKEY-INO-KEWARD	129

List of Tables

5.1	Parameter generation strategy (\bigcirc : True \bigcirc : False)	109
5.2	Benchmark Applications	117
5.3	Selected RESTful API vulnerabilities identified by NAUTILUS. For	
	Vulnerability Identification Tools: \checkmark : this tool can identify the	
	vulnerability, X : this tool cannot identify the vulnerability, $-$: this	
	tool is not designed to uncover the type of vulnerability. For Manual	
	Annotation: \checkmark : the annotations are updated by human experts	
	to provide guidance on operation and parameter generation, \boldsymbol{X} : no	
	manual inputs into the specification annotations.	118
5.4	False positives / true positives identified by different tools on the	1.00
	evaluation benchmarks	120
5.5	Performance of NAUTILUS against RESTLER, ZAP and w3at in	
	terms of both the average endpoint coverage and detected vulnera-	
	and highlight statistically significant results in hold (We calculate	
	the average increased number by $(\# of \text{NAUTILUS})-(\# of baseline)$	191
	# of baseline .).	141
6.1	Usage policies of service providers	136
6.2	Number and ratio of successful jailbreaking attempts for different	
	models and scenarios.	139
6.3	LLM Chatbot generation token count vs. generation time (second),	
	formatted in mean (standard deviation)	141
6.4	Experimental results of time-based LLM testing. Time formatted in	
	mean (standard deviation). Unit: Second	145
6.5	Performance comparison of each baseline in generating jailbreak	
	prompts in terms of query success rate. Values in bold indicate	
	the best-performing metrics in their respective categories	157

Abstract

In an era where technology and human interaction are increasingly intertwined, human-interactive systems, such as robotics, web services, and artificial intelligence, play a pivotal role in our daily lives. From multi-robot systems managing complex tasks to large language model chatbots transforming human-machine communication, these systems are integral to modern society's functionality. However, ensuring the security of these systems poses a formidable challenge. Unlike traditional systems, human-interactive systems operate in environments with vast and unpredictable input/output spaces, making conventional security testing methods like fuzzing insufficient.

This thesis addresses the critical and complex issue of conducting effective security testing on human-interactive systems. It tackles the unique challenges posed by the extensive and dynamic nature of these systems' interaction with both their environment and users. The research encapsulates four comprehensive studies, each targeting a different facet of human-interactive system security, yet collectively contributing to a broader understanding and enhancement of these systems' security.

The first study delves into the Byzantine threats in Multi-Robot Systems (MRSs), revealing the intricate and expanded attack surface that arises from their collaborative nature. A novel methodology specific to the Robot Operating System (ROS) is introduced, demonstrating how traditional security approaches can be adapted and applied to these complex systems.

In the realm of robotic operating systems, the second study focuses on ROS2, highlighting the vulnerabilities inherent in its security module, Secure ROS2 (SROS2). This research not only identifies critical security flaws but also proposes an innovative defense mechanism, showcasing the need for and application of advanced security measures in these systems. The third study shifts the focus to RESTful APIs, which are fundamental to web services yet are prone to overlooked vulnerabilities. The introduction of NAU-TILUS, an advanced tool for detecting API vulnerabilities, underscores the importance of specialized security approaches in dealing with the nuanced and diverse nature of human-interactive systems.

Finally, the thesis addresses security concerns in Large Language Model (LLM) chatbots. Through the development of Jailbreaker, a comprehensive framework, the research provides insights into the complex nature of security threats in AI-driven human interaction systems, highlighting the need for robust and adaptive security strategies.

Overall, this thesis presents a novel and holistic approach to security testing in human-interactive systems, emphasizing the need for specialized methods to address their unique security challenges. By bridging the gap between traditional security testing methods and the dynamic nature of these systems, this research significantly advances the field of system security in the context of human-machine interaction.

Chapter 1

Introduction

The recent decade marks a transformative era in which technology and human life are becoming ever more entwined. Central to this transformation are humaninteractive systems [1-3] — a broad category encompassing robotics [4-6], web services [7], artificial intelligence [8–10] and beyond. These systems have swiftly transitioned from being mere technological novelties to becoming vital cogs in the machinery of our daily lives and the broader societal infrastructure. For instance, multi-robot systems efficiently handle complex tasks. Large language model chatbots [11], such as ChatGPT [9] has revolutionized human-machine communication, demonstrating the deep integration of these systems into various facets of modern living.

As these systems grow in complexity and importance, their security emerges as a critical concern. Traditional security measures, designed for more static and predictable environments, struggle to cope with the dynamic and often unpredictable nature of human-interactive systems. The conventional tools and methodologies for security testing, such as fuzzing [12–14], are increasingly seen as inadequate for these advanced systems. This inadequacy stems from several factors. Firstly, the complexity of human-interactive systems surpasses that of previous systems, with their ability to handle undetermined inputs and outputs leading to vast and unforeseen variations, thus posing a significant challenge in maintaining their integrity and reliability. Secondly, these systems grapple with both security and safety issues, necessitating assurances that they not only fulfill their intended tasks but also operate without harming people or compromising other users' security.



FIGURE 1.1: Illustration of commonly used testing methodologies

The distinctive nature of these systems demands a fundamental shift in our approach to their security. Traditional methods are no longer sufficient; there is an urgent need for innovative and specialized security testing strategies. These strategies must be capable of comprehensively understanding and adeptly adapting to the complexities and fluid nature of human-interactive system environments. In response to this critical gap, this thesis is committed to exploring and establishing advanced methodologies for effective security testing of human-interactive systems across various dimensions.

1.1 Motivation

While some existing works have effectively addressed the complexities in traditional systems like robotics and web services, the ever-increasing intricacy of these technologies continues to challenge conventional security methodologies. The burgeoning complexity of traditional systems like robots and APIs, coupled with the advent of novel technologies such as Large Language Models (LLMs), necessitates a reevaluation and enhancement of existing security methodologies. In addressing these challenges, this thesis aims not only to adapt and improve traditional solutions but also to extend their applicability to newer, more complex systems.

Several methodologies are commonly used in system security testing, as shown in Figure 1.1. These include two primary testing scenarios: whitebox testing, where the system's internal mechanisms are known, and blackbox testing, where only inputs and outputs are accessible, with system runtime information hidden. For blackbox scenarios, fuzzing [12, 14, 15] is often employed to deduce the system's

execution logic, analyzing test results to refine test case generation. In whitebox scenarios, with access to runtime information, strategies like instrumental-based fuzzing are applicable for higher precision and better test case generation. When systems can be formally described, formal analysis techniques such as model checking [16, 17] and theorem proving [18, 19] are used. These strategies aim to uncover misimplementations, including bugs, exceptions, logical errors, and security vulnerabilities.

These traditional testing strategies have shown effectiveness in certain contexts but are increasingly inadequate for the intricate and dynamic environments of advanced human-interactive systems. For instance, in the realm of robotics and web services, the expanding functionalities and interconnectivity demand a more nuanced approach to security testing. Formal verification, while comprehensive, often becomes impractical for large-scale, dynamic systems due to its intensive computational requirements. Fuzzing, on the other hand, though beneficial in uncovering vulnerabilities, struggles to navigate the complex and unpredictable input/output scenarios presented by these systems. Therefore, enhancing and adapting these traditional methodologies to suit the evolving landscape of human-interactive systems is a critical focus of this research.

In the case of newer technologies like LLMs, the challenges are fundamentally different. These systems, typified by black-box models, defy traditional security testing approaches due to their opaque nature and continuous learning capabilities. The traditional methods are not equipped to handle the unpredictability and evolving nature of these models, making it crucial to develop innovative security testing approaches. This thesis seeks to bridge this gap by applying and modifying traditional methodologies, such as fuzzing and formal verification, in novel ways to address the unique challenges posed by LLMs and other advanced systems.

By extending and refining traditional security methodologies, this research aims to make them compatible with both existing and emerging systems. The approach is twofold: firstly, to enhance the efficacy of these methodologies in complex and dynamic environments like robotics and APIs; and secondly, to adapt and apply these methods in new contexts, such as the security testing of LLMs, where traditional approaches have previously been ineffective. This dual approach ensures that the security solutions developed are not only robust but also versatile, capable of addressing the diverse range of challenges presented by the current and future landscape of human-interactive systems. The ultimate goal is to establish methodologies that can effectively secure these systems, thus ensuring their reliable and safe integration into the fabric of modern society.

1.2 Main Work

This thesis is anchored in addressing the evolving security challenges of humaninteractive systems through a series of focused studies, each targeting a specific aspect of these systems. The overarching aim is to develop and refine security methodologies that are both robust and adaptable, capable of meeting the unique demands of these increasingly complex systems. Four distinct yet interconnected studies form the core of this thesis, each contributing to a comprehensive understanding and enhancement of system security in the context of human-machine interaction.

<u>Byzantine Threats in Multi-Robot Systems (MRSs)</u>: The first study delves into the security of Multi-Robot Systems (MRSs), particularly focusing on Byzantine threats where certain robots may be unreliable or compromised. This research proposes a novel methodology tailored for the Robot Operating System (ROS), which includes three innovative steps: requirement specification using signal temporal logic, attack surface determination via data-flow analysis, and attack identification employing requirement-driven fuzzing. This approach not only identifies new types of attacks but also tests their impact in both simulated environments and real-world MRS scenarios. The study's findings significantly advance our understanding of the security dynamics in MRSs and demonstrate the necessity of specialized security measures for these complex systems.

<u>On the (In)Security of Secure ROS2 (SROS2)</u>: Taking one step further, this study examines the security of ROS2, the most popular open-source robotics programming and control platform. We particularly focus on its native security module, Secure ROS2 (SROS2). Through a systematic analysis from multiple perspectives, this research identifies critical vulnerabilities in SROS2. It also proposes a defense solution based on private broadcast encryption, enhancing the security of ROS2. The study's experimental setup, including simulation and physical multirobot testbeds, illustrates the practical implications of these vulnerabilities and the effectiveness of the proposed solutions. This research underscores the need for continuous evaluation and enhancement of security mechanisms in robotic operating systems.

<u>NAUTILUS: Automated RESTful API Vulnerability Detection</u>: This work focuses on web services, particularly RESTful APIs, which are fundamental yet often vulnerable components of these services. NAUTILUS, an advanced automated tool, is introduced for uncovering API vulnerabilities. This tool incorporates a novel specification annotation strategy, enabling it to generate meaningful operation sequences and uncover vulnerabilities that require the execution of multiple API operations in a specific order. The effectiveness of NAUTILUS is demonstrated through extensive testing on various RESTful services, revealing its capability to detect significantly more vulnerabilities compared to existing tools. This study highlights the importance of specialized approaches in addressing the security challenges of web services.

<u>MasterKey: Automated Jailbreak Across Multiple Large Language Model Chatbots</u>: The final study addresses the security concerns in LLM chatbots, particularly focusing on "jailbreak" attacks. The research introduces Jailbreaker, a comprehensive framework that offers insights into these attacks and their countermeasures. By employing a time-sensitive approach to reverse-engineer the defensive strategies of prominent LLM chatbots and developing an automatic generation method for jailbreak prompts, this study marks a significant advancement in understanding and mitigating jailbreak threats. The success of Jailbreaker in bypassing existing defense mechanisms and its high success rate in automated jailbreak generation underscore the need for more robust defenses in the realm of LLM chatbots.

Together, these studies provide a multifaceted view of the security landscape in human-interactive systems, each contributing to the broader goal of enhancing the security and reliability of these systems in our increasingly interconnected world.

1.3 Contribution of the Thesis

This thesis makes several significant contributions to the field of human-interactive system testing.

1. Enhancing Fuzzing Techniques for MRS Security: We advance the security of MRSs through an innovative adaptation of fuzzing techniques. This approach is specifically designed to address Byzantine threats in MRSs, a critical concern in autonomous driving systems. It incorporates requirement specification using signal temporal logic, data-flow analysis, and requirement-driven fuzzing, providing a comprehensive framework for identifying and mitigating complex security risks in these distributed systems. This contribution is particularly notable for its practical application in enhancing the security of MRSs against sophisticated and evolving threats.

2. Formal Verification in complex systems: We creatively introduce strategies to perform formal verification on complex real-world systems. In particular, we apply formal verification to ROS2's security module, SROS2. This contribution is pivotal in identifying inherent vulnerabilities within a critical, industrial-level system.

3. Cryptographic Defense Strategy for ROS2: The development and implementation of a cryptographic defense strategy for ROS2, based on private broadcast encryption, is another major contribution of this thesis. This innovative approach enhances the security framework of ROS2, demonstrating a practical solution to safeguard industrial robotic systems against sophisticated threats.

4. Human-in-the-Loop Fuzzing for RESTful APIs: We introduce humanin-the-loop fuzzing strategy for RESTful API systems. Our solution NAUTILUS incorporates OpenAPI specification annotations, allowing for more effective and logical operation sequences in RESTful API testing. This approach significantly improves the capability to detect complex vulnerabilities that require multiple API operations in sequence, a notable advancement over traditional fuzzing techniques. The framework also empowers human experts to interact with and refine the testing process, bridging the gap between automated testing and human expertise.

5. Jailbreaking Large Language Models with Testing: We propose "MasterKey," a comprehensive framework for understanding and countering jailbreak attacks. This contribution is substantial in revealing and circumventing the defense mechanisms of LLM chatbots. It employs a novel method inspired by time-based SQL injection attacks for reverse engineering chatbot defenses, allowing for an unprecedented understanding of these systems' security. Additionally, it introduces an effective technique for automatically generating jailbreak prompts, significantly enhancing the capability to test and strengthen the defenses of LLM chatbots against sophisticated attacks.

To summarize, this thesis collectively advances the field of human-interactive system testing through its significant contributions. It demonstrates an innovative adaptation of fuzzing techniques to enhance security in Multi-Robot Systems (MRSs), especially against Byzantine threats. The application of formal verification to complex systems, particularly ROS2's SROS2, identifies critical vulnerabilities in industrial-level systems. The introduction of a cryptographic defense strategy for ROS2 further fortifies security in robotic systems. The development of the NAU-TILUS framework for RESTful APIs using human-in-the-loop fuzzing marks an evolution in API security testing. Finally, the "MasterKey" framework for Large Language Model (LLM) chatbots showcases novel methods for understanding and countering jailbreak attacks, enhancing the robustness of LLM security. Each of these contributions, while distinct in their focus, collectively represents a significant stride in securing increasingly complex and critical human-interactive systems.

1.4 List of Materials Related to the Thesis

The thesis mainly contains the materials from the following papers.

- Gelei Deng, Yuan Zhou, Yuan Xu, Tianwei Zhang, and Yang Liu. 2021. An Investigation of Byzantine Threats in Multi-Robot Systems. In *Proceedings* of the 24th International Symposium on Research in Attacks, Intrusions and Defenses (RAID '21), 2022.
- Gelei Deng, Guowen Xu, Yuan Zhou, Tianwei Zhang, and Yang Liu. 2022. On the (In)Security of Secure ROS2. In Proceedings of the 2022 ACM SIGSAC Conference on Computer and Communications Security (CCS '22), 2022.
- Gelei Deng, Zhiyi Zhang, Yi Liu, Yuekang Li, Tianwei Zhang, Yang Liu, Guo Yu, and Dongjin Wang. In 32nd USENIX Security Symposium (USENIX Security '23), 2023.

• Gelei Deng, Yi Liu, Yuekang Li, Kailong Wang, Ying Zhang, Zefeng Li, Haoyu Wang, Tianwei Zhang, and Yang Liu. In *The 31st Network and Distributed System Security Symposium (NDSS '24)*, 2024.

1.5 Outline of the Thesis

The remainder of this thesis is organized as follows:

Chapter 1 provides an overview of this thesis, including motivations, main work, and contributions.

Chapter 2 reviews related works, elaborating on the motivations behind this research.

Chapter 3 focuses on Byzantine threats in Multi-Robot Systems (MRSs), introducing innovative fuzzing techniques for enhanced security.

Chapter 4 discusses the application of formal verification in ROS2 and the development of a cryptographic defense strategy for its security module.

Chapter 5 presents NAUTILUS, a novel framework for human-in-the-loop fuzzing in RESTful API systems, significantly improving vulnerability detection.

Chapter 6 introduces MASTERKEY, a comprehensive approach for understanding and mitigating jailbreak attacks in Large Language Model chatbots.

Chapter 7 summarizes the findings, discusses the implications of the research, and outlines future work directions. This research contributes to advancing the security and robustness of human-interactive systems, enhancing their reliability and safety in complex environments.

Chapter 2

Related Work

In this section, we introduce the related works and necessity the background information.

2.1 Human-Interactive Systems

Human-Interactive Systems [2, 3, 20] are at the nexus of cutting-edge computing and intricate human dynamics, encompassing a vast spectrum of evolving applications. These systems signify a shift towards more complex and interactive technological landscapes, where the integration of human elements into computing platforms is central. As these systems grow in sophistication and user engagement, they present unique challenges in terms of development, implementation, and particularly, system testing.

The field of robotics, especially in human-robot interactions [21, 22], has advanced significantly, pushing the boundaries of how machines perceive, understand, and respond to human inputs. This complexity in interaction demands rigorous and innovative testing methodologies to ensure reliability and safety. Similarly, web applications [7, 23] have transformed from static pages to multifaceted platforms. Their complexity lies not only in the backend technology but also in the frontend user experience, requiring comprehensive testing strategies that cover a wide array of user scenarios. The rapid progression in artificial intelligence, with the advent of large language models (LLMs) [9, 24, 25], has given rise to sophisticated AI-driven

chatbots. These systems, while enhancing user interaction, introduce complexities in natural language processing and behavioral prediction, complicating the testing process. Other systems, such as Virtual reality (VR) systems [20, 26] have expanded their scope beyond entertainment, venturing into practical and industrial domains [27, 28]. The immersive nature of VR introduces unique challenges in testing, particularly in ensuring user safety and system stability in diverse, often unpredictable, virtual environments.

As human-interactive systems continue to advance, their increasing complexity and the need for nuanced user interaction pose significant challenges for system testing. Ensuring the reliability, safety, and user-friendliness of these systems requires innovative testing approaches that can adapt to the dynamic nature of human interaction within complex technological frameworks. The evolution of these systems underscores the need for continuous development in testing methodologies, matching the pace of technological innovation and the depth of human-computer integration.

2.2 Robotic Systems

As a subset of human-interactive systems, robotic systems have seen immense growth, seamlessly integrating into various aspects of modern life. They epitomize the fusion of mechanical engineering, computer science, and human-machine interaction, revolutionizing industries from manufacturing to healthcare. With advances in AI, robotics now extend beyond traditional automation, engaging more directly and intelligently with human environments [29].

2.2.1 Robot Operating System

The Robot Operating System (ROS) [4] has established itself as a pivotal framework in the field of robotics, offering a comprehensive and robust environment for developing a wide range of robotic applications. Renowned for its open-source nature, ROS boasts a user-friendly interface, an extensive library of resources, and a strong, collaborative global community. By 2023, it has become the predominant framework for robotics development, acclaimed for its versatility and ease of use. In response to evolving needs, particularly in terms of security and multi-robot communication, ROS2 [5] was developed as an advanced iteration of ROS. This new version places a significant emphasis on enhancing security features, incorporating elements such as encryption, authentication, and access control. Additionally, ROS2 is designed to facilitate seamless communication among multiple robots, enabling the effective formation and operation of multi-robot systems (MRSs).

Both ROS and ROS2 excel in simplifying the intricacies of robotic programming. They provide indispensable tools for tasks such as hardware abstraction, device control, and algorithm implementation, making them essential in both research and industrial applications. Their widespread adoption is evidenced by their use in various platforms, from the Dji Matrice 200 drone [30] to the PR2 humanoid robot [31], underscoring their significant impact and utility in the robotics community.

2.2.2 Multi-Robot Systems

Multi-Robot Systems (MRSs) [32–34] are a dynamic area within robotics, facilitating collaborative and distributed problem-solving. These systems have gained prominence in tackling large-scale and complex tasks that single robots cannot achieve efficiently. MRSs are characterized by their flexibility and adaptability, making them ideal for diverse applications like environmental exploration, collaborative manufacturing, and swarm intelligence. The coordination in MRSs, whether through centralized or decentralized schemes, plays a critical role in their effectiveness, impacting aspects like task allocation, communication efficiency, and system robustness.

2.3 Human Interactive Web Applications

Web applications have become increasingly complex and interactive, reflecting the ongoing evolution of human-interactive systems. As these applications develop, they not only offer enhanced functionalities but also foster more intricate interactions with users. This complexity and interactivity are central to modern web applications, shaping user experiences and expectations in the digital world.

2.3.1 Web Applications

Web applications, an integral part of modern internet usage, have evolved to offer dynamic and sophisticated user experiences. This evolution is marked by the increasing importance of APIs, particularly RESTful APIs, which are fundamental in connecting different web services and applications. RESTful APIs enable seamless interactions and data exchange between various online platforms, playing a crucial role in the functionality and scalability of web applications. They facilitate a more connected and integrated web experience, allowing applications to communicate efficiently and effectively with each other.

2.3.2 AI-driven Chatbots

AI-driven chatbots, particularly those powered by large language models (LLMs), have transformed the landscape of human-computer interaction within web applications. LLMs, with their advanced natural language processing capabilities, enable chatbots to conduct more nuanced and contextually aware conversations. This advancement has made AI-driven chatbots an essential feature in many web applications, enhancing user engagement, providing personalized assistance, and automating complex interactions. The incorporation of these chatbots marks a significant step towards creating more interactive and intelligent web environments.

2.4 System Testing

System testing is essential in the technological realm, especially for ensuring the reliability and security of various technologies in the rapidly evolving landscape of human-interactive systems. Traditional testing solutions, while effective for conventional systems, often fall short when applied to the complexity inherent in human-interactive systems. This section delves into key methodologies and approaches in system testing, focusing on their application, evolution, and particularly their limitations in addressing the unique challenges posed by these complex systems.

2.4.1 Fuzzing

Fuzzing [12, 35], a cornerstone in system testing, is exceptionally effective for uncovering vulnerabilities in complex systems, including robotic platforms and web services. This method employs the generation of a vast array of random inputs to induce unexpected behaviors or crashes, thereby exposing potential security weaknesses. Traditional fuzzing strategies can be broadly categorized into two types: whitebox and blackbox approaches. Whitebox fuzzing [15], often more thorough, analyzes the internal structures and workings of the application, leveraging this insight to generate more effective test cases. On the other hand, blackbox fuzzing [13], not reliant on internal data, tests the system from an external perspective, simulating the actions of an end-user or an attacker without knowledge of the system's internal mechanics.

Despite its adaptability to various system architectures and complexities, fuzzing encounters limitations in the realm of human-interactive systems where the input spaces are extensive and unpredictable. This unpredictability can significantly hamper the efficacy of fuzzing in thoroughly identifying vulnerabilities within such environments. The diverse and dynamic nature of human-interactive systems' inputs, often influenced by user behaviors and external interactions, challenges traditional fuzzing methods. This leads to the necessity for more nuanced or hybrid testing approaches that can better navigate the complex landscape of these systems, ensuring more comprehensive security and reliability testing.

2.4.2 Static Analysis

Static analysis serves as a complementary approach to fuzzing in system testing. It involves examining the code of a system without executing it, aiming to uncover potential flaws or vulnerabilities. This approach is beneficial for identifying certain types of errors early in the development process. However, its effectiveness is limited in dynamic, human-interactive systems where runtime behaviors and user interactions significantly influence system performance. Static analysis may not adequately capture the complex interactions or the evolving nature of these systems, thus missing critical vulnerabilities that only manifest during actual operation.

2.4.3 Formal Verification

Formal verification, utilizing mathematical methods, is a rigorous approach employed to ensure the correctness and security of systems against specific properties or specifications. This technique is particularly valuable in critical domains like robotic operating systems, where safety and reliability are paramount. By applying formal verification to systems in real-world scenarios, it becomes possible to identify and methodically address security vulnerabilities [36].

However, formal verification encounters significant challenges when applied to humaninteractive systems, largely due to their inherent complexity and the necessity for more flexible, adaptive approaches. One major challenge arises from the sheer complexity of these systems, which often makes it infeasible to verify every aspect thoroughly. Another challenge is the dynamic nature of these systems; they often evolve rapidly based on user interactions and environmental changes, outpacing the static capabilities of formal verification. This rapid evolution and complexity make formal verification less suitable for ensuring the ongoing security and correctness of such dynamic systems. The need for approaches that can adapt to and accommodate these rapid changes is therefore crucial in the context of system testing for human-interactive systems.
Chapter 3

Identifying Byzantine Risks in Multi-Robot Systems

Multi-Robot Systems (MRSs) show significant advantages to deal with complex tasks efficiently. However, the system complexity inevitably enlarges the attack surface and adds difficulty in guaranteeing the security and safety of MRSs. Thus, we present an in-depth investigation about the Byzantine threats in MRSs, where some robot is untrusted. We design a practical methodology to identify potential Byzantine risks in a given MRS workload built from the Robot Operating System (ROS). It consists of three novel steps (requirement specification using signal temporal logic, attack surface determination via data-flow analysis, attack identification using requirement-driven fuzzing) to thoroughly assess MRS workloads. We use this fuzzing method to inspect five typical MRS workloads from past works and the ROS platform, and identify three novel kinds of attacks that can be launched with five attack strategies. We conduct comprehensive experiments in the Gazebo simulator and a real-world MRS with three TurtlBot3 robots to validate these attacks, which can remarkably decrease the system's performance, or even cause task failures.

3.1 Introduction

The robotics technology is becoming more popular and ubiquitous in our society. A variety of intelligent and autonomous robots were designed to significantly improve

our work efficiency and quality of life. With the increased complexity of tasks and performance demands, Multi-Robot Systems (MRSs) have gained ever-growing attention. Multiple mobile robots are interconnected with each other within the same environment. They collaboratively work on an enormous task, which is hard to achieve by a single robot. Due to these benefits, MRSs have been developed for myriad scenarios and applications, such as precision agriculture [37–40], minefield mapping [41–45], search and rescue [46–48].

The significance of MRSs calls for special attention to security, as the system complexity can increase the attack surface. Past works have demonstrated that a single robot device is vulnerable to a plethora of attacks from different components, including sensors [49–52], actuators [53, 54], motion controller [55], Robot Operating System [56, 57] and applications [15, 49, 55]. These vulnerabilities enable an external adversary to easily intrude into the robot and take full control of the robot, resulting in Byzantine faults in an MRS. A Byzantine fault describes a condition of a distributed system where some components may fail and there is a lack of sufficient information to identify such failures [58]. In a distributed MRS, if one robot is compromised, it has the potential to affect other robots and even threaten the entire system, due to their close communication and collaboration. For instance, in 2021 January, a drone swarm got crashed during a light show in Chongqing, China. Up to 100 drones lost controls and hit into a building due to a small bug in the mainframe control [59].

Such Byzantine problem has been extensively studied in traditional distributed systems. However, it is relatively less explored in the context of MRSs. Prior work [60–63] considered the Byzantine resilience in MRSs from a theoretical perspective. They simply treat each robot as a dot without considering the specific workloads, robots' capabilities and physical constraints. Some other works [64] fuzzes the inter-robot communication based on Dolev-Yao threat model [65], yet similarly, they do not model the physical environment of robots. Hence, it is infeasible to apply these solutions and findings to the real-world Multi-Robot scenarios and tasks in a practical way.

We are particularly interested in two unsolved questions: given the design or implementation of an MRS, (1) how can we identify the potential Byzantine threats and the optimal attack strategy? (2) how much damage can a Byzantine robot bring to the entire MRS? Addressing these two questions is challenging. First, an MRS can execute a variety of workloads with distinct characteristics and user requirements. The inter-robot communication and collaboration can be implemented with various mechanisms. So it is hard to design a unified method for vulnerability identification and assessment. Second, during the operation, a malicious robot has very high freedom to affect other robots and the entire system in unexpected ways. They exchange different types of messages, and each message has a large input space for the robot to tamper with. This makes it difficult to comprehensively search for potential attacks.

In the following of this chapter, we provide the *first* practical study towards the Byzantine threats in MRSs based on the Robot Operating System (ROS) [66]. ROS is the most popular open-source platform for robot app development. It provides thousands of packages to achieve various functions, compatible with mainstream robot devices. This platform has benefited the robotics research, as well as the development of commercial products in industry, e.g., Dji Matrice 200 drone [30], PR2 humanoid [31], and ABB manipulators [67]¹.

We design a requirement-driven fuzzing methodology, which can automatically analyze a given MRS workload and identify the potential Byzantine risks. We consider a threat model where only one robot in an MRS is malicious. Our method can be easily extended to the case with multiple Byzantine robots. We assume the Byzantine robot can arbitrarily compromise any type of messages sent from it at any time. Then our fuzzing strategy has three steps to discover the potential risks. (1) *Requirement specification*: we formulate the requirements for the normal operation of an MRS with Signal Temporal Logic (STL). This includes the general requirements (safety, mechanism, performance) as well as task-specific requirements. (2) Data-flow analysis: we dynamically generate the data-flow diagram at the level of node operations by simulating the workload. Through analyzing this diagram, we extract the parameters and communication messages controllable by the Byzantine robot, which form the fuzzing input space. (3) Requirement-driven fuzzing: we mutate the messages from the identified input space and check whether the requirements are violated. Different mutation strategies (dropping, content modification, etc.) are considered for different types of messages.

We build an MRS workload suite, which incorporates standard implementations of common MRS workloads and coordination schemes from the past literature [68–85]

¹ROS is used as a communication wrapper in ABB manipulators.

and ROS platform. Using our requirement-driven fuzzing methodology, we uncover three new forms of Byzantine attacks with five attack strategies in these existing workloads. (1) Task assignment control attack: the Byzantine robot can manipulate the messages of location, robot status or task bidding to compromise the task assignment process. (2) Map merging poisoning attack: the Byzantine robot can decrease the quality of generated map by falsifying the explored map messages. (3) Task forwarding manipulation attack: the Byzantine robot can tamper with the transmitted task information to mislead other robots to perform wrong jobs.

We perform extensive experiments using the Gazebo simulator [86] to validate the effectiveness of these attacks. They can significantly decrease the performance of the entire system, or even cause system crash. Moreover, we deploy these workloads in a real-world environment and MRS consisting of three TurtleBot3 UGVs [87], and successfully achieve the discovered attacks.

In summary, we make the following contributions:

- We design a novel requirement-driven fuzzing methodology to identify Byzantine threats and the corresponding strategies for distributed MRS workloads.
- We introduce and opensource a first-of-its-kind MRS workload suite, consisting of different standard workloads and coordination schemes. They can be deployed in simulators as well as physical robots for performance evaluation, security assessment and other purposes as well.
- We discover three new forms of Byzantine attacks in existing common MRS workloads.
- We perform evaluations in both simulation and real-world environments, and the real-world experiments confirm that the consequences observed in simulated environments are realistic.

The rest of this chapter is organized as follows. Section 3.2 introduces the background of ROS, MRS workloads and our threat model. Section 3.3 presents our novel fuzzing method for Byzantine threat identification. We describe our MRS workload suite in Section 3.4, followed by the discovered attacks in Section 3.5. Sections 3.6 and 3.7 demonstrate our evaluations in a simulator and physical environment, respectively. We discuss possible countermeasures and related works in Section 3.8, and conclude in Section 3.9.

3.2 Background and Threat Model

3.2.1 Robot Operating System

ROS is the most popular robotic platform for robot research and development. It has been widely adopted in the research community, as well as industry, e.g., Dji Matrice 200 drone [30] and PR2 humanoid [31]. This platform provides full-stack open-source services to ease the development of robotic workloads. First, it offers a set of core libraries as the low-level middleware. These libraries are deployed between robot apps and hardware to support runtime execution, such as abstracting hardware, passing messages and managing devices. Second, it provides thousands of high-level packages for various functions [88]. Developers can integrate these packages to build a robot workload. For the rest of this work, we focus on the Multi-Robot workloads implemented from the ROS platform. Our methodology and tool can be extended to other robotic platforms and implementations as well.

3.2.2 Workflow of Robot Tasks

The workflow of a task running on a robot can be represented as a Directed Acyclic Graph of actions (actionDAG), where each node represents a certain action, and edges represent the dependencies of the actions in this task. Figure 3.1 shows the structure of a standard robot task. It consists of three fundamental stages: (1) *Perception*: the robot extracts estimated states of the environment and the device from raw sensor data. It uses the Localization node to determine the device position, and CostmapGen node to model the device's surroundings. (2) *Planning*: the robot determines the long-range actions. It uses the Path Planning node to find the shortest path, and Exploration node to search for accessible regions. (3) *Control*: the robot processes the execution actions and forwards these motions to the actuators. It uses Path Tracking to produce velocity commands following the planned path, and Motor Driver to transfer the velocity command to specific actuators.



FIGURE 3.1: Application pipeline for a typical robot task.

3.2.3 Multi-Robot Systems

In an MRS, a number of robots with the same type (homogeneous) or different types (heterogeneous) work together to complete one workload. Such collaboration mode can bring two benefits over single-robot systems. First, since a robot is mainly designed with one specific functionality, the incorporation of multiple robots can address complex tasks that can never be achieved by one robot. Second, the computation capability and power capacity of a robot are limited. Hence, an MRS can significantly increase the working efficiency and operation duration. Due to these advantages, MRSs have been practically adopted in many scenarios.

3.2.3.1 Multi-Robot Workloads

We present a categorization of common MRS workloads in our daily life.

Navigation. This type of workloads is a fundamental capability of mobile robot systems, widely applied in house cleaning [89], warehouse delivery [90, 91], surveillance [92] and patrolling [6, 93]. It can be abstracted as determining the robot's own position and moving towards a predefined destination. To achieve this goal, in Figure 3.1, the CostmapGen node creates a costmap of the robot's surroundings, and Localization estimates the robot's position. Based on such information, Path Planning generates an optimal collision-free path to the destination. Path Tracking follows this path and outputs the best action. The final velocity command is sent to the actuators. During moving, each robot needs to frequently interact with the environment, recognize surrounding objects or other robots, and possibly recalculate the path.

Exploration. In this type of workloads, the robots are expected to spread in an unknown area to achieve the maximal coverage, and collect as much information as possible. Typical examples of exploration include map building [94] and rescue [95]. Generally, the goal of each robot is to keep reaching new locations that are never touched by other robots. In Figure 3.1, the localization node executes the Simultaneous Localization and Mapping (SLAM) algorithm to infer the robot's position in absence of a map. Then, Exploration selects an unexplored position as the destination and sends the goal to Path Planning. By repeating this process of costmap update and exploration, the map of the environment will be expanded, until the entire area has been explored.

Formation. A swarm system is a special MRS which consists of a large number of simple robots with local sensing and communication capabilities. These robots interact with each other to produce complex swarm behaviors. Formation is one typical workload for swarm systems, where the robots try to maintain certain physical arrangements or patterns. There are two typical swarm behaviors in a formation task. The first type is aggregation/dispersion. Aggregation refers to the behavior where robots from different locations gather together in one spot. In contrast, dispersion is to move the robots from one spot to fully cover a certain area. The second type of swarm behaviors is pattern formation: robots need to adjust their locations to create a global shape, varying from simple geometry [96] to more complex shapes, e.g., alphabetical letters [97].

Antagonism. An MRS can also be implemented for the purpose of antagonism, e.g., robotic soccer and robot combat. For example, in a soccer game, the robots in one team are instructed to compete with the opponent team to score the goals and defending the opponent robots. Such MRSs are usually implemented in a closed monitored environment and less prone to attacks. So we do not discuss the security vulnerabilities of these workloads.

3.2.3.2 Communications in Multi-Robot Systems

Since robots in an MRS collaborate on the workload, they need to frequently exchange messages. In general, robots share information by either broadcasting or one-to-one communication via wireless networks. To efficiently control the entire system, there are typically two coordination schemes in modern MRSs.

Centralized scheme [68, 71]. In this design, a centralized entity is introduced to coordinate all the robots in an MRS. This entity can be a local edge gateway, a remote cloud server, or even a powerful robot inside the system. It collects information from the robots, makes decisions, and sends the instructions to different robots.

Decentralized scheme [73, 78]. This design eliminates the centralized entity, so each robot can communicate with others directly. Every robot retrieves information from the environment and other robots and makes decisions by itself. Robots exchange or broadcast messages frequently to make the entire system reach consensus. This decentralized scheme exhibits a higher level of autonomy.

It is worth noting each workload can be implemented by either the centralized or decentralized scheme. They may have different efficiency for different workloads. In Section 3.4, we will review and analyze the real-world MRS implementations for different workloads and coordination schemes.

3.2.4 Threat Model

We consider an MRS where a number of robots collaboratively work on one workload. We focus on the Byzantine threats in this system. Particularly, we assume one robot is malicious and fully controlled by the adversary, which attempts to compromise the entire MRS. There are several reasons that make this assumption realistic. (1) The ROS middleware lacks basic security mechanisms for the authentication and encryption of the communication between nodes, and thus suffers from many security issues, e.g., plaintext communication, lack of authentication or authorization [56, 98], and denial-of-service vulnerability [57]. A remote adversary can easily leverage these vulnerabilities to break into the robot and control it to perform arbitrary malicious behaviors. (2) A lot of function packages in the ROS platform contain exploitable software vulnerabilities [57, 99]. According to the Robot Vulnerability Database [100, 101], 17 robot vulnerabilities and 834 bugs (e.g., no authentication, uninitialized variables, buffer overflow) were discovered in the function packages of 51 robot components, 37 robots and 34 vendors in the ROS platform. Most of them are still unpatched. Exploitation on real-world ROS package vulnerabilities was reported [102], and red teaming strategies on ROS applications were studied in [103]. These software vulnerabilities also enable the adversary to intrude into the robot and take full control of it. (3) The ROS platform is open for everyone to upload and share their function packages. Unfortunately, it does not perform any security check over the submitted code. Hence, an adversary can publish malicious function packages for other users to download. Based on the ROS2 Robotic Systems Threat Model [104]: "third-party components releasing process create additional security threats (third-party component may be compromised during their distribution)".

The Byzantine robot tries to affect the completion of the workload by sending malicious messages to other robots or the centralized controller. It can also refuse completing the tasks assigned to it. This can bring disastrous consequences due to the following two facts. First, robots are closely interconnected with each other following either the centralized or decentralized scheme. The stability and integrity of the entire MRS highly depend on the reliability of the inter-robot communications. Second, there is a severe lack of Byzantine-resilient mechanisms in existing MRS designs and implementations. Developers do not consider Byzantine defenses because it is challenging to have a satisfactory solution due to the variety and complexity of messages exchanged between robots. Deploying such defenses can also decrease the system performance as a ratio of robots are not trusted. These two facts exacerbate the severity of Byzantine threats in MRSs.

The Byzantine robot can also have other means to interfere with the system. For instance, it can alter the environmental states or perform sensor spoofing attacks to indirectly affect the decisions of other robots. It can also conduct physical damages (e.g., path blocking, collision) to compromise the system. Those attack vectors are not considered in our work, as they are less stealthy and could be easily detected by the ground monitoring systems.

Note that we do not consider the case where the benign robots can detect the existence of the Byzantine robot via monitoring the surrounding environments. The reason is that due to the limit of communication and sensing ranges, a robot cannot obtain the global information of the environment or the past behaviors of other robots. The absence of such information makes it hard for a benign robot to identify whether its neighbors are anomalous from their current behaviors and



FIGURE 3.3: Workflow of the proposed methodology.

states. How to detect the Byzantine threat from benign robots will be an interesting future work.

An example of Byzantine threats.

Figure 3.2 shows an example of Byzantine threats in a surveillance workload, where three robots perform the navigation task given by the control station in real time. Each task consists of a sequence of planned checkpoints that the robot needs to follow. We assume that robot 1 becomes the Byzantine adversary and sends falsified path data to the control station. This can cause the station to mistakenly believe some untouched checkpoints have been passed, and then replan new tasks for benign robots which will exclude these checkpoints. Then these spots will never be navigated, and the workload cannot be completed.

3.3 A Methodology to Characterize Byzantine Risks in MRS

In this section, we present our novel methodology to automatically and comprehensively identify possible Byzantine threats in MRSs.

3.3.1 Overview

Given an MRS workload, our goals are to (1) identify whether its implementation is Byzantine-resilient, *i.e.*, functioning well when some robot is malicious; (2) if not, produce the optimal attack strategies to compromise the system.

As described in Section 3.2.4, a Byzantine robot can tamper with arbitrary messages in an arbitrary way to interfere with the entire MRS. To thoroughly identify potential Byzantine risks in a workload, we propose to use the fuzzing strategy [12]. However, there are several design challenges to apply fuzzing to our scenario. First, traditional fuzzing bug-oracles are designed to mainly detect system crashes, rather than abnormal system states in our case (e.g., robots are stuck in the idle status permanently). To address this issue, we introduce a bug oracle which is aware of MRS states via Signal Temporal Logic (STL) formulas with robustness semantics [105, 106]. The STL formulas describe the requirements the MRS should satisfy during its operation. Our method constantly monitors if the formulas are violated when fuzzing the target workload. Second, traditional fuzzing techniques cannot generate mutated communication messages due to the large input space of the MRS workload, with numerous types and formats of messages. To handle this limitation, we propose to leverage dynamic data-flow analysis to extract the critical inter-robot communication messages, which can significantly reduce the input space for fuzzing.

Figure 3.3 shows the workflow of our methodology. We adopt the STL to specify the requirements that the MRS should follow (Section 3.3.2). With these requirements, our method performs dynamic data-flow analysis (Section 3.3.3) and requirement-driven fuzzing (Section 3.3.4) to extensively evaluate the workload and output the possible attacks. Below we introduce the mechanism of each step in detail.

3.3.2 Requirement Specifications

During the execution, an MRS should satisfy various requirements to guarantee safety and task completion. These requirements can be divided into general ones (e.g., safety) and task-specific ones (e.g., navigation coverage, map accuracy). We adopt STL to formulate these two kinds of requirements for attack identification.

We briefly describe some basic concepts of STL, while more details can be referred to [106]. Let \Box , \Diamond , and U be the temporal operators "always", "eventually", and "until", respectively. Given a variable set X, its value at time t is denoted as X(t). Then a signal w over X is a time sequence $(t_0, X(t_0)), \ldots, (t_n, X(t_n))$, where $t_0 = 0$ and $t_i < t_{i+1}$. For our case, the variable set of a robot is the position x, velocity v, acceleration a, and the set of detected obstacles O. The syntax of an STL formula φ over X can be defined as: $\varphi := \top \mid \mu \equiv f(X(t)) > 0 \mid \neg \varphi \mid \varphi_1 \lor \varphi_2 \mid \varphi_1 U_{[a,b]}\varphi_2$, where \top means True, \neg is the negation operator, and \lor is the disjunction operator. $\mu \equiv f(X(t)) > 0$ is called an atomic STL formula, where $f : X \to R$ is a realvalued function related to a property, e.g., the distance function (e.g., the minimal distance between a robot and its surroundings) for safety consideration. We use these notations to describe some representative requirements for an MRS.

3.3.2.1 General Requirements

First, we consider some general requirements which are suitable for various Multi-Robot workloads.

Safety. The most important requirement is collision avoidance. At any time instance, a robot should keep a safe distance d_s from obstacles, including other robots in the same environment. Let d(t) be the minimal distance between the robot and obstacles at time t, then the STL formula for safety is $\varphi_1 \equiv \Box(d(t) \ge d_s)$.

Mechanics. Due to the physical limitations, the speed and acceleration of a robot cannot exceed the boundaries. Suppose the maximal speed and acceleration of a robot are v_{max} and a_{max} , respectively, then the STL formula is $\varphi_2 \equiv \Box (0 \leq v(t) \leq v_{\text{max}} \& |a(t)| \leq a_{\text{max}})$.

Energy saving. Due to the limited battery capacity, a robot is expected to reach its destination x_g before power exhaustion. Hence, suppose the battery power

at time t is E(t), the STL formula can be written as $\varphi_3 \equiv (\Diamond || x(t) - x_g ||_2 \leq \epsilon) \& ((E(t) > 0) U (|| x(t) - x_g ||_2 \leq \epsilon))$, where ϵ is a predefined tolerance for task completion.

Execution time. For an arbitrary workload, each robot is expected to complete the assigned task as soon as possible within a given time budget T. Hence, the STL formula for timeliness can be written as $\varphi_4 \equiv \Diamond_{[0,T]} || x(t) - x_g ||_2 \leq \epsilon$.

3.3.2.2 Task-specific Requirements

In addition to the above general requirements, there are also some specific requirements for different workloads. We describe three examples as below.

Navigation requirement. In a navigation workload, the robots in the system are required to complete a set of navigation tasks, such as going through a set of waypoints or regions. However, due to the unexpected and dynamic changes in the environment, not every task can be completed safely (e.g., some spots are occupied by accident). Hence, each system is given some tolerances for task completion. Given a set of navigation tasks $\{x_g^1, x_g^2, \ldots, x_g^K\}$ and the minimal task completion rate ω , the STL formula for the navigation requirement can be written as $\varphi_5 \equiv$ $\langle (\wedge_{i \in \{i_1, \ldots, i_k\}} || x(t) - x_g^i ||_2 \leq \epsilon) U k/K \geq \omega$, where \wedge denotes the conjunction operator.

Exploration requirement. For exploration, robots are instructed to collect as much information as possible with a shorter moving distance. To evaluate the completion of an exploration task, the MRS is required to cover $w \in [0, 1]$ of the ground truth map within a given time duration T. Hence, the STL formula for this requirement can be written as $\varphi_6 \equiv \Diamond_{[0,T]} M(t)/M \ge w$, where M(t) is the area of the explored map at t.

Formation requirement. In a formation workload, robots in the system should coordinate with each other to form a predefined formation. The system first determines a set of positions for these robots to occupy. Hence, we can illustrate the following requirement: given a formation with a set of vertexes $\{p_1, p_2, \ldots, p_m\}$, the system should assign each vertex to one robot exclusively, and then each robot moves to its corresponding destination within a given time budget T. Let $\sigma(r_1, r_2, \ldots, r_m)$ be a permutation of the robots $\{r_1, r_2, \ldots, r_m\}$ and $\sigma(i)$ is required to move to p_i . The STL can be described as $\varphi_7 \equiv \Diamond_{[0,T]} \wedge_{i \in \{1,2,\dots,m\}} \|x_{\sigma(i)}(t) - p_i\|_2 \leq \epsilon$.

3.3.3 Data-flow Analysis

After deriving the requirements, we need to identify the input space for fuzzing. According to our threat model, the Byzantine robot can send arbitrary malicious messages to other robots or the centralized controller. We propose to use dataflow analysis [107] to identify the critical messages that could possibly violate the requirements. This can be achieved with the following two steps automatically:

3.3.3.1 Data-flow graph construction

In ROS applications, the task in each robot consists of multiple computation nodes that perform different functions (Figure 3.1). The communication between those nodes (either inside one robot or across different robots) is implemented in a publisher-subscriber mode. Message topics are many-to-many named buses which describe the states of robots or environment. A node can subscribe to a topic if it wants to receive relevant data, or publish data to a topic. Therefore, given an MRS workload, we first construct the corresponding data-flow graph [108] to include all nodes and the types of messages flowing among them.

3.3.3.2 Byzantine message extraction

The next step is to identify the critical messages that can be manipulated by the Byzantine robot. To achieve this, we label each inter-node communication based on its source node and highlight the nodes controlled by the Byzantine robot. Messages sent by these nodes are potential targets for falsification. We exclude messages transmitted within a single robot and focus solely on inter-robot communication for fuzzing. Although the extraction of Byzantine messages is a one-time effort, accurately identifying the message topics for mutation is essential for launching the fuzzing process. This is where the construction of the dataflow graph becomes indispensable. By constructing the data-flow graph, we can precisely locate the topics to fuzz, ensuring that our fuzzing efforts are effectively directed at the most impactful areas. This approach underscores the continued importance of the data-flow graph construction in our methodology.

We inspect all the Multi-Robot packages from the ROS platform [66] and discover six common types of messages as the targets of the Byzantine attacks, elaborated as below:

M1: Odometry. This type of messages typically stores the estimation of the robot's instant velocity and position in the environment. This message is important for robots to adjust their motion to avoid collision and complete motion tasks. For instance, in a navigation scenario, the centralized controller needs to collect robots' exact positions from their odometry messages and calculate the corresponding paths for them.

M2: Robot status. Robots in an MRS need to frequently broadcast their current statuses (e.g., "active", "idle") for the system to properly allocate the tasks in time. Some MRS workloads may introduce more statuses to better coordinate the robots. For instance, in a coordinated exploration task, robots can stay at the "verification" status when they are moving in the explored regions. The map information sent at this status may be used to increase the map accuracy. The exploration tasks are preferably assigned to the robots at the "idle" status, and then to those at the "verification" status. This can maximize the utilization of all the robots in the MRS.

M3: Map. Most robot workloads need the map information during the execution, whether it is known (navigation) or unknown (exploration). In ROS, a map message is generally represented as the occupancy status of each cell in the target region. A typical map message in the *nav_msgs* package contains a variable *MapMetaData*, which includes the information of the width and height of the map in terms of the number of cells, the resolution of each cell and the origin of the map, and a vector variable *data*, which describes the occupancy probability of each cell in the map. The accuracy of the exchanged map information can heavily affect the allocation and execution of subsequent tasks.

M4: Reward. In some MRS workloads, each robot calculates the reward of performing one specific task and broadcasts the value to the entire system. A new task is thus allocated to the robot with the highest reward. As a result, the

reward values can significantly affect the allocation decisions, then the efficiency and completion of the entire workload.

M5: Task/Goal. This type of messages contains the current task to be completed. In a centralized system, these messages are sent from the centralized controller to each robot for task assignment. In a decentralized system, robots broadcast those messages until the task assignee receives the task information. A Byzantine robot participating in the propagation of such messages can tamper with the tasks or goals and mislead the assignee to perform wrong jobs.

M6: Path. This type of messages contains the trajectory of a robot from the current location to the destination. In some workloads (e.g., decentralized exploration [73]), each robot has the capability and responsibility of calculating its own path based on the given goal. In the applications where individual robots do not have the computation capability to generate paths independently, the paths are calculated by the centralized controller and sent to the robots.

3.3.4 Requirement-driven Fuzzing

Our next stage is to perform requirement-driven fuzzing over the MRS workload. We mutate all possible critical messages identified in Section 3.3.3.2, and monitor if they lead to any violations of the requirements specified in Section 3.3.2.

3.3.4.1 Overview

Algorithm 1 details our requirement-driven fuzzing procedure. For each requirement in terms of the STL formula, our method repeatedly conducts the following steps for each message: (1) identifying the message data type and performing mutations according to the mutation strategy designed for the data type (Line 5); (2) replacing the original message with the mutated one, executing the workload in the simulator, and recording the state sequence of the system execution (Line 6). Particularly, the adversary has the right not to perform the assigned tasks, which is also considered during the simulation; (3) computing the robustness of the recorded state sequence (Line 7); (4) if a violation is detected, storing the simulation configurations and continuing with the next message (Lines 8 - 10); otherwise, if the robustness of the mutated message is smaller, replacing the old message with the mutated one and updating the corresponding robustness (Lines 11 - 13). After all the messages in the input space are fuzzed, we summarize the mutations that can lead to requirement violations. A new round of fuzzing will start if testing time is allowed.

Algorithm 1: Requirement-driven Fuzzing
Input: A simulator SIM , the set of message types for mutation M , fuzzing
space $\Pi_{msg\in M} Input(msg)$, a set of STL formulas Φ , a fuzzing
time-limit $ au$
Output: V: requirement robustness related to each message; M_v : the
corresponding messages causing violations.
1 for each requirement $\varphi \in \Phi$ do
2 Initialize the values of the input messages $M = \{msg_0, msg_1, \dots, msg_n\},\$
$V(msg_i) = +\infty, \ M_v = \emptyset;$
3 while $total_time \ i \ \tau \ do$
4 for each msg in M do
5 $mutated_msg = MUTATE(msg, Input(msg));$
$6 \qquad state_seq = SIM.SIMULATE(mutated_msg);$
7 $v = req_checker(\varphi, state_seq, 0); /* Compute the robustness of$
φ with respect to $state_seq$ */
\mathbf{s} if $v < 0$ then
9 $M_v = M_v \cup \{msg\};$
10 $M = M \setminus \{msg\};$
11 else if $v < V(msg)$ then
12 $msg \leftarrow mutated_msg;$
$13 \qquad \qquad \qquad V(msg) = v;$
14 end
15 end
16 if $M = \emptyset$ then
17 Generate a new set of <i>M</i> randomly, initialize <i>V</i> , and repeat Lines 4
18 end
19 end
20 end

3.3.4.2 Message Mutation Strategy

For a given requirement with the STL formula φ , the corresponding robustness degree v of the system can be calculated at the end of the workload execution. The mutation strategy aims to minimize the system robustness degree by varying the messages sent by the Byzantine robot, and ideally result in a system requirement violation. For different types of messages, we provide a couple of possible mutation strategies.

The first strategy is to drop the messages. The Byzantine robot can pretend to "forget" sending critical messages to the corresponding receiver, or broadcasting them to the entire system. This is one kind of Denial-of-Service attack in MRS.

The second strategy is to randomly change the values contained in the messages within the legal range. For the numerical type (e.g., odometry, reward), the Byzantine robot can change the message to a random value within the legal range. For the categorical type (e.g., robot status), the Byzantine robot can change it to a different category. Similarly, for the message type of task/goal, the adversary can alter the content to a random task pre-defined in the workload.

The third strategy is specifically designed for the map message. A map message is a 2-dimensional metric, which provides a much larger fuzzing input space than other types. So a fully random mutation strategy is inefficient. Instead, we consider two new mutation methods: (1) the Byzantine robot replaces the target map with an empty one or a fully-occupied one, to check the system's Byzantine-resilience in extreme cases; (2) the Byzantine robot randomly picks a region with a distance of l from its current position, where l is a pre-defined hyper-parameter based on the map size.

3.3.4.3 Requirement Checking

Give an STL formulas φ and a sequence of system states $state_seq$, the requirement checking process $req_checker(\varphi, state_seq, t)$ returns the robustness degree of φ over w at time instant t, which describes how far w is from satisfying or violating φ at t [106]. The robustness can be computed as follows. First, the robustness of the atomic STL $\mu \equiv f(X(t)) > 0$ with respect to $w(X) = X(0), X(1), \ldots, X(n)$ at time instant t can be computed as $\rho(\mu, w, t) = f(w(X))[t] = f(X(t))$. Based on the syntax of STL, we have $\rho(\neg \varphi, w, t) - \rho(\varphi, w, t), \rho(\varphi_1 \lor \varphi_2, w, t) = \max\{\rho(\varphi_1, w, t), \rho(\varphi_2, w, t)\}, \rho(\varphi_1 U_{[a,b]}\varphi_2, w, t) = \max_{t' \in [t+a,t+b]} \max\{\rho(\varphi_1, w, t), \rho(\varphi_2, w, t)\}$

 $\min\{\rho(\varphi_2, w, t'), \min_{t'' \in [t,t']} \rho(\varphi_1, w, t'')\}$. Hence, the robustness of an arbitrary STL formula can be computed by applying the above the computation recursively. More details about the robustness degree functions can be found in [106].

Based on the definition of robustness degree, $\rho(\varphi, w, t) < 0$ means that the signal violates φ at t. Hence, for each STL-based requirement $\varphi \in \Phi$ derived from the requirement specification step (Section 3.3.2), the system robustness degree ρ on the selected requirement is calculated with the system state sequence obtained from the simulator. If we detect an execution whose ρ is smaller than 0, i.e., a requirement violation, we store the system configurations and mutated message; otherwise, we guide the mutation to the direction that decreases the robustness. If no requirement violation is detected at the end of fuzzing, the lowest system robustness together with the corresponding mutated message and system configurations are returned as output.

3.4 An MRS Workload Suite

To extensively evaluate the effectiveness of our method and understand the Byzantine vulnerabilities in MRSs, we select five typical MRS workloads as our testbed, which cover a variety of coordination schemes and application domains discussed in Section 3.2.3. These workloads are identified from prior literature and existing packages in the ROS platform. Each workload can support an arbitrary number of robots running end-to-end tasks including perception, planning and motion control. They are ready to be deployed to a ROS simulator (e.g., Gazebo [86]) or physical robot devices.

To our best knowledge, this is the first-of-its-kind workload suite for Multi-Robot applications based on the ROS platform. We open-source this MRS workload suite², and expect it can contribute to the robotics community for other purposes as well (e.g., performance evaluation and characterization, MRS hardware and software co-design). We give detailed descriptions of these workloads in this section, followed by their security assessment in the next section.

3.4.1 Workload Descriptions

W1: Centralized Navigation [68–70]. The goal of this workload is to efficiently complete a surveillance task by coordinating multiple robots to navigate

²https://github.com/GeleiDeng/RAID_2021_MRS_Fuzzing

to different target locations. Figure 3.4 shows the workflow of this workload. A centralized controller server (e.g., Ground Control Station (GCS)) is introduced to manage the communication. Specifically, given a sequence of goal positions, the controller server generates the corresponding collision-free paths and sends the path information to different robots. Then each robot follows the designated path to reach its destination. During moving, a robot needs to avoid collisions with obstacles and other robots. Meanwhile, it also needs to frequently send two types of messages to the controller server: (1) odometry messages containing the robot's current location; (2) status messages denoting whether the robot has arrived at the destination, or in the "idle" status waiting for further commands. The controller server takes such information to update the paths and assign a new task to the robot which is in the "idle" status and closest to the target location.



W2: Centralized Exploration [71, 72]. In this scenario, an MRS is deployed to build a map of an unknown area. This workload is implemented in a centralized manner, where a GCS is used to manage the robots for exploration. Figure 3.5 shows the workload of this implementation. During the task, the GCS runs the exploration stack and identifies the frontiers of potential areas to be explored. It then assigns the frontiers to available robots by calculating the exploration cost and utility [78], and generates the paths from the corresponding robots' current positions to the frontiers. Each robot frequently exchanges three types of information with the GCS: (1) odometry messages denoting the robot's current position, (2) status messages denoting whether the robot is in the exploration or idle status, and (3) map messages containing the exploration results. The GCS merges the maps from different robots. It assigns new area for the robot which has finished its current exploration. The above process is repeated until the entire map is established.



W3: Decentralized Exploration with Bidding [73–77]. This workload achieves the same function as W2, but in a decentralized manner. Robots talk to each other and adopt the bidding algorithm to reach agreement for frontier assignment. Figure 3.6 shows the workflow. During exploration, robots keep exchanging two types of messages: (1) a map message containing the local map maintained by the robot, and (2) a bidding message containing the robot's gains of exploring different frontiers. When a robot receives messages from other robots, it first merges their newly explored maps to its local one. Then it compares its gain with other robots' and selects the frontier where it has the highest gain. It declares to the system that it will explore this frontier, and then starts the task. These steps are repeated until the entire area is explored. At last, robots merge their local maps again to generate the final map.



W4: Decentralized Exploration with Group Merging [78–81]. This workload is similar to W3. The difference is that robots are highly distributed without knowing the relative position of each other. A robot cannot broadcast to all the robots: It can only talk to the robots which move into its sensing range. The workflow is shown in Figure 3.7. In this scenario, the group merging algorithm is adopted to achieve task allocation. Specifically, when two robots meet, they exchange the map information and verify whether their maps can be merged together. Robots with confirmed joint map regions form one group and share the explored maps via the wireless network until they move out of the communication range. At the end of the task, all the robots will share the same map information for the entire area.



W5: Swarm formation [82–85]. This workload is designed for a swarm system. It controls the swarm robots to achieve aggregation, dispersion and line formation. It is applied to InchBot [109], a novel swarm microrobotic platform that contains highly modular two-wheel mini robots with wireless sensing and communication capability. Figure 3.8 shows the workflow. Each robot obtains the relative positions of other robots within the communication range through the wireless sensor network. They achieve the aggregation or dispersion behaviors by maintaining a pre-defined average distance with others within the sensing range. In the line formation task, each robot moves to an ideal location to form a line with its adjacent robots, while maintaining a pre-defined distance in a similar way as dispersion. The formation commands are given by a controller outside the system, and then robots broadcast the commands to others within the sensing range.



3.4.2 Metrics

Our workload suite also provides a set of metrics to measure the performance and efficiency of Multi-Robot workloads. They can be classified into the following two categories.

General metric. We measure the *execution time*, which is the total time spent in completing the workload. A severe attack can significantly increase the execution time and cause Denial-of-Service damage. An extreme case is that the execution

time can be infinity as the workload can be never completed. This is a workloadindependent metric and can be used to describe different scenarios.

Task-specific metrics. In addition to the above general metric, different workloads can also have diverse measurements for the performance and efficiency. (1) For a navigation workload, we measure the *navigation rate*, i.e., the percentage of the destination spots reached by the robots. We expect the system to cover as many desired destinations as possible. (2) For an exploration workload, we adopt the *map quality* to quantify the performance of the execution [110]. It directly reflects how well a map can be constructed by the robots. (3) For a formation workload, we introduce *formation similarity*, which denotes the similarity between the planned and actual patterns. Specifically, we adopt an Euclidean distance-based similarity measure between the two formation spaces:

$$s = e^{-\frac{\sum_{i=1}^{m} \|x_i - p_i\|_2}{m}}$$

where p_i is the ideal position of the *i*-th robot in the formation, x_i is its actual position, and *m* is the number of robots in the system.

3.5 Multi-robot System Risk Analysis

We leverage the proposed risk identification methodology in Section 3.3 to assess the five workloads described in Section 3.4. The five workloads are implemented and fuzzed in the simulation environment based on Algorithm 1. System requirement violations together with the mutated messages are recorded, and the root causes of those violations are manually inspected. We finally identify seven Byzantine risks in these implementations and characterize them into three classes of attacks, as described below.

3.5.1 Attack 1: Task Assignment Control

Multi-Robot Systems efficiently achieve the ultimate goals by breaking down the high-level task into sub-tasks and appropriately assigning them to qualified robots. This task assignment process involves a series of calculations to maximize the overall system gain based on the task, environment and current status of each robot. It can be performed in a centralized controller or distributed to individual robots depending on the coordination scheme.

If the Byzantine robot can manipulate the messages related to task assignment, it can compromise the assignment decisions and affect other robots. We identify two strategies to realize this attack, targeting different schemes and messages.

3.5.1.1 Fake location or status information in centralized systems

Typical centralized systems assign tasks to the most appropriate robots by considering their positions and statuses. For instance, in the workloads **W1** and **W2**, tasks are assigned to available robots closest to the target positions. The Byzantine robot can send fake location and status information to the controller, causing it to make wrong assignments. Specifically, in the navigation workload **W1**, the Byzantine robot can lie to the GCS that it is the closest to the target positions. Then it can intercept all the navigation tasks that are supposed to assigned to other robots. In the centralized exploration workload **W2**, idle robots have higher priority to get assignments. The Byzantine robot can send the *idle* status to the GCS, even it still has uncompleted tasks. Such messages can also mislead the GCS to assign more tasks to the Byzantine robot, while ignoring the correct candidates. More seriously, the malicious robot can occupy these tasks without finishing them. This can significantly decrease the completion degree of the workload.

3.5.1.2 Fake bidding information in decentralized exploration systems

The task assignment mechanism can be attacked in a decentralized MRS as well. For instance, in the workload **W3**, robots bid for the exploration task by calculating the overall gain based on their current positions and statuses. A Byzantine robot can easily steal tasks from others by broadcasting fake bidding information with extreme high gain values. Then it can just keep these tasks uncompleted to affect the system performance.

3.5.2 Attack 2: Map Merging Poisoning

In exploration workloads, the final map is generated by merging local maps from all the robots. Unfortunately, existing map merging packages in the ROS platform (e.g., [111]) adopt the common map merging algorithms [112] without verifying the correctness of the input map data. Hence, a Byzantine robot can keep sending false map information to the map merging function to affect the exploration process. We identify two strategies against the workloads **W2**, **W3** and **W4** based on this attack.

3.5.2.1 False global map generation

In the workloads **W2** and **W3**, the adopted map merging packages [111, 112] from the ROS platform commonly assemble maps by generating the union of occupancy maps submitted by each robot and then performing noise reduction. A Byzantine robot can compromise this algorithm by sending wrong map data where empty cells are replaced by occupied cells. Then the final merged map gives wrong information for those cells.

3.5.2.2 Blocking group merging

In the decentralized exploration workload **W4**, robots exchange the map information after the confirmation of joint map areas. Such algorithm can mitigate the random false map generation attack to some extent, as the falsified map might be verified and corrected by other robots who have explored the area. However, the Byzantine robot can still craft a partially fake map to block the grouping process. For instance, it can just introduce false map information to the cells which are significantly far away from its current position, and unlikely to be explored by other robots. In this case, the faked data will not be verified by other robots and are merged directly. As a result, the faked information will block the merging of maps from benign robots.

3.5.3 Attack 3: Task Forwarding Manipulation

In some systems where task information is transmitted through robots without the centralized controller, a Byzantine robot can manipulate the task information such that the subsequent robots will receive and conduct wrong tasks. We identify one such attack that is applicable to the swarm formation workload **W5**.

Man-in-the-Middle attack. In a swarm system, robots are assumed to have limited sensing and communication ranges. Hence, a robot cannot send commands to all the robots directly. Therefore, it first issues the formation task to a random robot within the communication range. This robot then forwards the task to other robots it can talk to. The task messages are then propagated via the sensory network and reach every robot in the system. A Byzantine robot inside the propagation chain can change the task messages, causing some robots inside the network to execute false commands.

3.6 Evaluation

In this section, we conduct simulation experiments to validate the Byzantine risks identified in Section 3.5. Evaluations with physical experiments can be found in the next section. More simulation and physical experimental results and video recordings are listed online³.

3.6.1 Experimental Setup

We select the Gazebo simulator [86] with Rviz [113] for the simulation of the MRS with five workloads. Gazebo is the mainstream open-source simulator that can accurately reflect the physical characteristics of robots. We configure Gazebo to simulate a group of robots with rigid body and workload environments. Rviz is a 3D visualization tool for ROS applications. It can display message contents with different ROS topics, and provide APIs for users to publish desired messages to the related topics. We use Rviz to visualize the 2D information from both the

³https://geleideng.github.io/RAID_2021_MRS_Byzantine/

simulator and robot applications and publish navigation/exploration goals to the workloads.

We simulate the workloads W1 - W4 in a $14 \times 14m^2$ square room, which is further separated into multiple smaller compartments (Figure 3.9). We implement a homogeneous MRS with the TurtleBot3 robots [87]. The number of robots varies from 3 to 5 for each workload. Each robot is equipped with a 2D Lidar sensor covering a maximum sensing range of 10 meters to detect the surroundings. For the formation workload **W5**, we simulate a system with 10 to 20 InchBots [109] on an open surface.





(A) 3D view of the room in Gazebo



FIGURE 3.9: Simulated environment for workloads W1 – W4.

We consider two baselines for comparisons with our attack. (1) Normal: all the robots in the MRS are benign and follow the received instructions to complete the tasks. (2) Idle: there exists a Byzantine robot in the MRS. It stays idle without requiring any tasks or sending messages. This represents the simplest Byzantine attack which can degrade the system performance to some extent. For each workload in each case, we assume the GCS or the benign robots have the ground truth of the completion status of the tasks. So they can determine the time to stop the tasks.

All the simulations are conducted on a Unbuntu 16.04 laptop equipped with an Intel i7-9750H CPU and 32GB RAM. We adopt the ROS Kinetic version for all the MRS workloads. Each experiment below is repeated for 10 times and the average result is reported.



FIGURE 3.10: Task assignment control attack against W1.

3.6.2 Evaluation Result

3.6.2.1 Task Assignment Control Attack

To launch this attack against **W1**, **W2** and **W3**, we randomly select one robot as the Byzantine robot and falsify the task assignment messages sent from it.

Figure 3.10 shows the results for the navigation workload W1. For the *idle* situation, the Byzantine robot increases the mission time by 26.1%, 23.0% and 11.5% for an MRS of 3, 4 and 5 robots respectively (Figure 3.10a). Due to the idle robot in the system, the performance of an MRS with r robots will be the same as that of an MRS with only r - 1 robots. When the Byzantine robot performs our discovered attack, it obtains all the tasks but never completes them. Then the mission completion time is infinity while the navigation rate is zero, resulting in task failures.

Figure 3.11 shows the results of the centralized exploration workload W2. A malicious idle robot can increase the mission time by 18.2%, 12.9% and 4.2% for the three MRSs, respectively. If it performs the task assignment control attack, then the performance degradation will be much larger (55.5%, 36.4% and 30.8%). Different from W1, the Byzantine robot cannot cause failures in W2. The reason is that the GCS generates and assigns multiple frontiers for exploration simultaneously. Each robot can only require one task at one time. Hence, the Byzantine robot cannot steal all the tasks. The exploration task will be finally completed by the benign robots. The Byzantine robot can affect the optimal assignment process to cause longer delay. We further analyze the effects of the task assignment attacks







FIGURE 3.12: Map accuracy of **W2** with 3 robots under the task assignment control attack.

on the process of map construction. Figure 3.12 shows the change of map accuracy for **W2** with three robots. The Byzantine robot starts to send malicious messages at 40s. After that, the map accuracy grows at a slower speed than the original scenario, delaying the task completion.

The performance of the bidding-based decentralized exploration workload W3 is shown in Figure 3.13. Similarly, in the *idle* situation, the Byzantine robot only increases the mission completion time. However, different from W2, only one frontier can be assigned to a robot via the bidding algorithm each time in W3. Hence, with the task assignment control attack, the Byzantine robot can occupy all the frontiers to be explored but does not conduct the jobs, causing the failure of the exploration task.

Summary: We observe that the task assignment control attacks can cause task failures for the entire MRS in W1 and W3. For W2, the workload can be completed due to the Byzantine robot's incapability of occupying all jobs. But it can



FIGURE 3.13: Task assignment control attack against W3.

still significantly degrade the performance of the entire system.

3.6.2.2 Map Merging Poisoning Attack.

We implement this type of attacks in the exploration workloads **W2**, **W3** and **W4**, respectively. The Byzantine robot sends falsified map to the GCS or other robots in the *attack* situation.

Figure 3.14 shows the performance of the centralized exploration workload W2. Similarly, compared with the *normal* situation, the *idle* situation only affects the mission duration time. However, in the *attack* situation, the global map generated at the GCS is poisoned by the falsified map from the Byzantine robot. Therefore, GCS fails to generate correct paths for the robots to follow. This will cause an immediate system failure. Figure 3.15 illustrates the map accuracy during the workload execution. Without an attack, the map accuracy grows gradually to saturation. When the attack occurs at 40s, the correct map generated previously is poisoned, so the accuracy of the merged map will immediately drop to close to zero and will never arise anymore.

Figure 3.16 shows the performance of the bidding-based decentralized exploration workload W3. The *idle* situation is similar with the previous workloads and attacks. Under the *attack* situation, each robot can function well even with the existence of a Byzantine robot as the exploration process relies on the local map which is correct for the benign robots. So the total mission completion time is not significantly affected. However, after the exploration is completed, the maps from these robots cannot be merged together due to the poisoned map information from









FIGURE 3.16: Map merging poisoning attack against W3.

the Byzantine robot. Hence, the workload can be treated as a failure without any maps produced.

For the group merging-based decentralized exploration workload W4, we consider two strategies. (1) The Byzantine robot conducts a simple false map generation attack (section 3.5.2.1). Figure 3.17 shows the corresponding results. The falsified



FIGURE 3.17: False map generation attack against W4.



FIGURE 3.18: Blocking group merging attack against W4.

map from the Byzantine robot cannot be merged into the exploration cluster, which increases the overall execution time for other benign robots to explore. But the quality of the final map is unchanged. (2) The Byzantine robot introduces a partially fake map to block the grouping process (Section 3.5.2.2). It can firstly form a group with several robots within its communication range and poison the map via its fake map information, causing the rest of the robots not able to join the group. As a result, the mission duration increases, and the maps cannot be correctly merged (Figure 3.18).

Summary: Based on the above analysis, we conclude that both centralized (W2) and decentralized (W3 and W4) exploration workloads are vulnerable to the map merging poisoning attack. Even though a decentralized workload can mitigate some simple attacks, a smarter adversary can still craft fake maps to cause task failures.



FIGURE 3.19: Visualized attack effects for W5.



FIGURE 3.20: Task forwarding attack against W5.

3.6.2.3 Task Forwarding Attack.

Finally, we consider this type of attack against the swarm formation workload (W5). The Byzantine robot can control the system formation behaviors by forwarding false task information. Figure 3.19 compares the formations of robots in the *normal* (a), *idle* (b) and *attack* (c) situations, when 10 robots are instructed to generate a line formation. Different from previous workloads, the *idle* robot can also affect the workload completion (i.e., formation). This is because even though the idle Byzantine robot does nothing, other robots can observe it and make formation decisions based on its wrong position. Hence, the idle robot increases the mission completion time and decrease the formation similarity. In the *attack* situation, the Byzantine robot has more severe impact on the formation since it can mislead other robots proactively to perform a wrong formation. Quantitative results are presented in Figure 3.20, where the formation similarity is calculated based on Section 3.4.2.

Summary: Systems that allow robots to propagate task information as middleman are vulnerable to this task forwarding attack. The selected swarm formation



FIGURE 3.21: Physical environment for real-world evaluation.

workload W5 is a typical example.

3.7 Real-world Evaluation

To fully validate the Byzantine threats, we implement an MRS and deploy the attacks against different workloads in the physical world.

3.7.1 Experimental Setup

Our testing environment is a $2.5 \times 5m^2$ maze. We adopt three Turtlebot3 devices to form a Multi-Robot System. Each robot is equipped with a Raspberry Pi 3 chip [114] as the on-board processor, and a 360-degree 2D laser scanner [115] for SLAM. The ROS core nodes are deployed on a Ubuntu 16.04 server connected to the robots through the wireless network. Figure 3.21 shows the environment with the corresponding 3D view from Gazebo and 2D view from Rviz.

Due to the limited physical space and number of robots, we only implement the navigation workload **W1**, exploration workloads **W2** and **W3**. We believe the conclusions will be applied to the other two workloads too.

For the centralized workloads **W1** and **W2**, the GCS nodes performing the path planning and map merging tasks are running on a server connected to the robots directly. For the decentralized workload **W3**, the processing nodes (path planning, mapping, etc.) of each robot are deployed on a server due to the limited computation capabilities of the on-board processor. To simulate the restricted communication range in the MRS, two robots are forbidden to share information



FIGURE 3.22: Task assignment control attack against W1, W2, and W3 (Physical).



FIGURE 3.23: Map merging poisoning attack against W2 and W3 (Physical).

if their distance is beyond a threshold (1m in our experiments). We launch the task assignment control attacks against **W1**, **W2** and **W3**, and the map merging poisoning attacks against **W2** and **W3**. We only compare the *normal* and *attack* situations.

3.7.2 Evaluation Results

Task assignment control attack. Figure 3.22 shows the impact of this attack against three workloads. For W1, We observe the task can never be completed with a navigation rate of zero. For W2, the map can be finally constructed much longer time (27.7% increase in mission duration). The workload W3 cannot be completed since the exploration tasks are not assigned to benign robots. These results are in general consistent with the simulation results in Section 3.6.
Map merging poisoning attack. Figure 3.23 shows the attack results against the two exploration workloads W2 and W3. W2 fails to complete, as the map at GCS is poisoned by the falsified data from the Byzantine robot. For W3, the robots can still perform and complete the exploration tasks. However, the final map cannot be correctly merged due to the falsified map sent by the Byzantine robot. These also match the simulated results in Section 3.6.

3.8 Discussions and Related Works

3.8.1 Countermeasures

MRS developers focus more on the development of motion algorithms to guarantee motion safety and task achievement, while ignoring the severity of Byzantine threats. To our best knowledge, there are no practical defense solutions deployed in current MRSs. We discuss two possible countermeasures that can help to alleviate the discovered Byzantine risks from Section 3.5. We expect that they can be adopted to enhance the security of MRSs in the near future.

The first direction is to implement message checking in MRSs. Messages sent by a robot imply their physical status, which should comply with some system rules. For instance, the distance between two positions of a robot recorded at two consecutive timestamps should be shorter than the maximum speed of the robot times the period duration. When a Byzantine robot launches the task assignment control attack in the navigation workload (Section 3.5.1), this system rule will be violated and GCS can detect the anomaly. We can design the corresponding rules for each type of communication messages, and enforce the rule checking in every robot in real-time. However, the Byzantine robot may realize such rules and carefully craft malicious messages that are not recognizable, but can still affect the system. How to design robust rules to reduce such possibilities is challenging but important as future work.

The second direction is to apply consensus protocols together with new coordination schemes to protect MRSs. A resilient consensus protocol [116] was introduced in swarm workloads such as formation control, flocking, and sensor fusion to detect Byzantine agents. Its main idea is that the system can be resilient to a number of F non-cooperative nodes by actively verifying information with neighbors as long as the network connectivity of the system is above (2F + 1). Strobel et al. [63] leveraged the blockchain technology to detect and exclude Byzantine robots in a swarm system. Those methods require the system to have very large number of robots with high connectivity, which may not be realistic in some practical scenarios. Besides, the system's efficiency will be sacrificed since some messages may only contribute to the information verification instead of the actual workload. In the future, we will consider to design more efficient and comprehensive communication schemes and consensus protocols for various types of MRSs.

3.8.2 Related Works

Byzantine faults in Multi-Robot Systems. Byzantine faults in MRSs were first discussed and modeled as a convergence problem of robot networks, i.e., a set of robots are required to asymptotically reach the same but prior unknown location. Bouzid et al. [117] proved the necessary and sufficient conditions to achieve convergence under Byzantine attacks in *Obvious Robot Networks*, where robots cannot recall past computations and can only move in one-dimensional space. Bouzid et al. [60] extended the mathematical theory to two other swarm systems based on the ATOM model [118] and CORDA model [119]. Auger et al. [61] developed a certified framework to prove the convergence of robot networks using the COQ proof assistant. Molla et al. [62] designed deterministic algorithms to identify the lower bounds of time and memory for solving the dispersion problem on a ring of robots. Zikratov et al. [120] proposed a trust management framework to identify malicious Byzantine entities in multi-agent systems.

These prior works mainly focused on the theories of Byzantine faults with very simple robotic functionalities and tasks. In contrast, our work presents the first *practical* study about the Byzantine threats in real-world implementations based on the Robot Operating System framework. We investigate the impact of Byzantine attacks on the complex workloads (e.g., navigation, exploration) with different coordination schemes. We also evaluate the discovered Byzantine attacks with both accurate simulations and physical experiments. These are never achieved in previous works. **Detecting vulnerabilities in robotic systems.** Pogliani et al. [121] designed a new methodology to perform data flow analysis and discover vulnerabilities in the source code of industrial robot software. Recently, researchers applied the fuzzing technique to study the security and safety of robotic systems and Autonomous Vehicles (AVs). For instance, CPFuzz [14] was designed to find the safety violations in cyber-physical systems. RVFuzzer [15] fuzzes the configuration parameters and environmental factors to identify input validations bugs in robotic vehicles. PGFuzz [13] is a policy-guided fuzzing framework to discover any policy violations in the control programs of robotic vehicles. Fuzz testing for AVs usually focuses on a single vehicle [122–125]. For example, [122] illustrated the application of fuzzing to test the vehicle's CAN bus; Li *et al* [123] proposed a testing framework, AV-FUZZER, to find safety violations of an autonomous driving system.

Different from the above works, our fuzzing method focuses on MRSs. For instance, PGFuzz fuzzes the input controller command and environmental variables to discover the potential vulnerabilities. It targets one single robot of specific kind, and the temporal logic formulas are extracted from the specification documents. In contrast, our work focuses on the interaction of multiple robots in a collaborative workload, and do not rely on the specification documents. Moreover, we propose different requirements for the secure and safe operation of an MRS in the STL formulas. We also identify the critical messages as the fuzzing space, and different strategies to mutate these messages for testing. This method is effective to identify Byzantine threats in an MRS implementation.

Other attacks against robotic systems. Past works have demonstrated that a variety of robotic components are vulnerable, and prone to different types of attacks. For instance, sensor spoofing attacks can spoof the sensor data (e.g. GPS [126–130], Lidar data [131, 132], optical images [51], gyroscopic data [52, 133, 134]) to cause the robots to make wrong decisions. An adversary can also tamper with the controller input (e.g., configuration or calibration parameters, perceived states), making the robot instable, halt, or rush to wrong directions [55]. Moreover, recent works disclosed many known cyber security issues in the ROS framework, such as plaintext communication, lack of authentication or authorization [56], and denialof-service vulnerability [57]. Finally, malware based on machine learning techniques [102] is also developed to maximize the attack impacts on ROS applications. While the cyber-attacks against individual robots have been studied, research about the security of Multi-Robot Systems is still at an early stage. This drives us to investigate and evaluate different attack strategies and their damages on various MRS workloads.

3.9 Conclusion

In this section, we perform an investigation towards the Byzantine threats in MRS workloads from the ROS platform. We propose a requirement-driven fuzzing methodology, which can automatically analyze potential Byzantine risks in an MRS workload. We build an MRS workload suite containing common implementations of typical Multi-Robot workloads and coordination schemes to evaluate our method. We identify three new forms of Byzantine attacks with five attack strategies, which are further validated by simulation and real-world experiments. We expect this study can raise the awareness of robotics researchers and developers about the severity of MRS Byzantine threats, and design new solutions to enhance the security and safety of existing MRSs.

Chapter 4

Security Investigation of Robot Operating System 2

Robot Operating System (ROS) has been the mainstream platform for research and development of robotic applications. This platform is well-known for lacking security features and efficiency for distributed robotic computations. To address these issues, ROS2 is recently developed by utilizing the Data Distribution Service (DDS) to provide security support. Integrated with DDS, ROS2 is expected to establish the basis for trustworthy robotic ecosystems.

In this research, we systematically study the security of the current ROS2 implementation from three perspectives. By abstracting the key functions from the ROS2 native implementation, we first formally describe the ROS2 system communication workflow and model it using a concurrent modeling language. Second, we verify the model with some key security properties through a model checker, and successfully identify four security vulnerabilities in ROS2's native security module: Secure ROS2 (SROS2). To validate these flaws, we set up simulation and physical multi-robot testbeds running different real-world workloads developed by Open Robotics and Amazon AWS Robotics. We demonstrate that an adversary can exploit these vulnerabilities to totally invalidate the security protection offered by SROS2, and obtain unauthorized permissions or steal critical information. Third, to enhance the security of ROS2, we propose a general defense solution based on the private broadcast encryption scheme. We run different workloads and benchmarks to show the efficiency and security of our defense. Our findings have been acknowledge by ROS2 official, and the suggested mitigation has been implemented in the latest SROS2 version.

4.1 Introduction

The robotics technology is playing an important role in the intellectualization of industry and our daily life. Its development is accelerated by the Robot Operating System (ROS). As the most popular robotic platform, ROS provides great ease for developing and managing robotic devices and applications [135]. However, ROS has its limitations by design. It lacks basic security features, leaving ROS-based systems extremely vulnerable [99, 136, 137]. Besides, it is not suitable for multirobot systems (MRS) in real-time processing. All the robots have to connect to one master node for communication, which makes the system inflexible and inefficient.

To address these problems, ROS2 is developed as an upgrade to ROS. ROS2 uses the Data Distribution Service (DDS) as the communication middleware instead of the traditional master-based communication method, which brings two main advantages. First, DDS allows participants to work in a distributed fashion, which efficiently extends the ROS2-based applications to various multi-robot scenarios [138–143]. Second, ROS2 develops its native security tool, SROS2, on top of DDS's built-in security modules. SROS2 provides many security features which are missing in ROS, such as network traffic encryption, authentication and access control. With these benefits, ROS2 rapidly gains huge popularity. An increased number of IT and robotic companies adopt ROS2 to develop their robotic products (e.g., Amazon Robomaker [144], iRobot [145], etc.)

While ROS2 aims to provide better protection than ROS, there are still unsolved security concerns about it. (1) The security of the new features in ROS2 is not thoroughly verified. These features may contain loopholes, which could be exploited to cause severe security, privacy and safety hazards. (2) The multi-robot scenario supported by ROS2 can bring new security challenges. A large quantity of heterogeneous robots from different parties and locations can be coordinated by the cloud service (e.g., Amazon RoboMaker) to complete complex tasks, which could potentially enlarge the attack surface of the entire system. With the fast adaption of ROS2, a comprehensive study on its security is urgently needed.

We present the first systematic investigation about the security of SROS2 with the following contributions. First, we design a method based on the model checking technique [146] for ROS2 verification (Section 4.4). Modeling every detail of the ROS2 system can be extremely challenging, because it involves multiple layers with thousands of functions, and the corresponding model can contain a huge number of states that may cause the state explosion issue [147]. To overcome this problem while accurately modeling the system, we leverage the code property graph [148] to represent the ROS2 client library and its DDS middleware implementation, and efficiently identify the key functions involved in inter-robot communication. We further eliminate the non-related components from the key function CPG representations, and analyze them to abstract the events describing the ROS2 inter-node communication workflow. Based on this, we model two key ROS2 components (nodes and DDS participants) and the communication environments as processes driven by those events. We formulate a set of desired security properties based on the official ROS2 Robotic Systems Threat Model [149], and leverage a model checker to automatically identify vulnerabilities that can lead to violations of these properties.

Second, with the aforementioned methodology, we successfully identify four vulnerabilities existing across multiple ROS2 versions, which can invalidate the SROS2 security mechanisms (Section 4.5). By exploiting those vulnerabilities, the adversary can (1) bypass access control to send arbitrary malicious messages to unauthorized ROS2 nodes, (2) receive confidential messages from unauthorized topics, or (3) extract sensitive information about the system security settings. We validate the exploitability and practicality of those vulnerabilities using four real-world workloads developed by Open Robotics [29] and Amazon AWS Robotics [144] through both simulation and physical experiments (Section 4.6). We confirm that a single malicious actor can easily terminate the entire system, mislead other benign robots to crash, and steal users' private information. These vulnerabilities have been reported to Open Robotics, the official maintainer of ROS2, and acknowledged by them. Following our suggestion, temporary mitigation methods have also been integrated into the ROS2 testing version.

Third, to thoroughly address the implementation flaws, we propose a general defense solution customized for ROS2 (Section 4.7). Patching these vulnerabilities separately requires careful modifications of the ROS2 source code to re-design the SROS2 access control functions, which can be a tremendous and tedious task. Instead, we propose to adopt the private broadcast encryption (PBE) primitive [150] to fundamentally address the security flaws in the SROS2 design. Our solution guarantees to provide secure access control as PBE is proved to have key indistinguishability under chosen-ciphertext attacks (IK-CCA). It can work with ROS2 as an individual security module without additional infrastructure support or modification of the ROS2 source code. We implement our solution as a lightweight Python library that can be imported directly by ROS2 applications. We deploy various workloads in our physical testbed to show that our solution can mitigate the discovered vulnerabilities with acceptable performance and resource overhead. We have open-source our solution on our submission website [151] to benefit the robotics community.

4.2 Background

4.2.1 Robot Operating System

Robot Operating System (ROS) adopts a *node*-based structure, where each node is an independent process that executes certain functions. A typical robot application comprises many nodes distributed in one or multiple robot devices. These nodes exchange messages with each other to finish the task cooperatively. The node communication follows a publish-subscribe mode through a *topic*: each node can publish *messages* with a customized data structure to a topic, and all the nodes subscribed to that topic will receive the messages.

With more emerging scenarios, the design of ROS exhibits two fundamental drawbacks. First, ROS is not suitable for distributed MRS development. All the network traffics must go through a *master* node, and every robot needs to keep continuous network connection with this node. This makes the master node a single-point-offailure and performance bottleneck. Second, ROS lacks the basic security mechanisms, and contains many security loopholes. While new security modules were developed by the community to patch these issues, they are not widely adopted in real-world applications. Up to now, the latest official ROS distributions have not included those extensions yet, making the majority of ROS-based systems vulnerable to various attacks [99, 136, 137, 152].



FIGURE 4.1: ROS2 DDS architecture with the DCPS protocol.

To thoroughly solve these issues, Open Robotics [29] proposed the new Robot Operating System 2 (ROS2) in 2014. ROS2 has the similar client library and userlevel API structure as ROS, so previously developed ROS applications can be easily migrated to the ROS2 platform. At the network transport layer, ROS2 adopts the Data Distribution Service (DDS) protocol [153], which has the distributed communication capability and built-in security modules. Therefore, ROS2 enjoys all the functionalities from the original ROS, with new support for distributed computing, better performance and security enhancements. With the increased number of packages and projects migrating from ROS to ROS2, ROS2 is expected to establish the basis for the future robotic ecosystems.

4.2.2 Data Distribution Service

DDS is a mature middleware protocol adopted in ROS2 for real-time connectivity. It supports a publish-subscribe protocol called Data-Centric Publish-Subscribe (DCPS) [154]. The basic structure of DCPS is illustrated in Figure 4.1. A global data space is created to contain all the data objects (i.e., DDS topics). These DDS topics are similar as the topic objects in ROS, and can be accessed by DDS processes. A process that publishes or subscribes to a topic is called a *participant*. The communication between participants are regulated by a series of configurable parameters that control the behaviors of DDS, namely Quality of Service (QoS). ROS2 interacts with DDS by calling the abstract DDS APIs (Figure 4.1). The userland code defines the function logic in the app, e.g., how the nodes communicate with others through topics, and how the received messages are processed. The code is then interpreted by the ROS2 Client Library (RCL) to form the node-based communication structure. This structure is further processed by the ROS2 DDS Middleware (RMW) to generate the corresponding DDS structure and configuration parameters. Finally, the DDS configurations are passed to the DDS APIs to build the DDS system structure. With these steps, ROS2 nodes and DDS participants establish a one-to-one relationship¹. At runtime, when an ROS2 node tries to publish a message, ROS2 translates such behavior into a series of DDS API calls, and the actual communication is achieved through DDS. In this process, ROS2 works as a middleware and does not handle protocol-level details.

4.2.3 DDS Security

DDS has its native security specification [156] that adds security protections by defining a series of Service Plugin Interfaces (SPIs). The DDS SPIs provide five security features: authentication, access control, cryptography, logging and data tagging. They can be enabled and configured through the QoS parameters. ROS2 adopts the first three features from DDS as summarized below:

Authentication. This plug-in uses the Public Key Infrastructure (PKI) [157]: each participant has a public-private key pair and an x.509 certificate that binds its public key to its name. Through the PKI, a DDS participant can verify other participants' identity by checking their certificates. Each x.509 certificate must be signed by (or have a signature chain to) one trusted Certificate Authority (CA), which is typically set up by the robotic system owner.

Access control. This plug-in defines and enforces restrictions on the DDS-related capabilities of a given domain participant. It requires two XML files per domain participant, signed by the CA. (1) A *governance file* specifies the domain properties, e.g., if the domain can be joined by other participants, if it can be discovered in the network, etc. (2) A *permission file* specifies the permissions of the domain

¹For performance optimization, ROS2 maps multiple nodes to one participant if these nodes share the same configurations. The design rationale is disclosed in [155].

participant. It declares if a participant can publish or subscribe to specific topics. This permission file is used to configure the access control policies for system participants.

Cryptography. This plug-in declares the cryptography-related operations, e.g., encryption, decryption, signature, etc. Both the authentication and access control plug-ins utilize these primitives to achieve their functions. By default, enabling this plug-in will encrypt all the DDS network traffics using the established Advanced Encryption Standard in Galois Counter Mode (AES-GCM) [158].

4.2.4 Secure ROS2

ROS2 builds its security mechanisms based on the DDS security specification. The system owner declares the security configurations in the ROS2 userland code, which will be interpreted and passed to the DDS security plug-ins. This set of security features are collectively named "Secure ROS2" (SROS2).

Specifically, SROS2 provides command line integration [159] to enable the SROS2 features. It includes a key generation tool that helps the system owner act as the CA and generate the certificate/key files for the nodes in the system. SROS2 standardizes the security file formats, and specifies how the system owners should distribute those files to the robots. These files need to be put in a specific *keystore* folder following the pre-defined structure and naming rules, so that they can be loaded by SROS2 and passed to DDS as QoS parameters. Enabling SROS2 features brings the following security mechanisms to the system:

Traffic encryption. In the default settings of ROS and ROS2, traffics between nodes are in plaintext. Once SROS2 is enabled, the messages are encrypted by the DDS cryptography plug-in.

Access control. SROS2 enforces access control on the nodes by restricting the underlying DDS participants' capabilities. The system owner provides the governance and permission files for all the nodes. Then each node can only publish/subscribe to the topics declared in its corresponding permission file.

Topic information protection. In ROS, topic-related information is public and can be retrieved by the built-in RCL tool (i.e., rostopic [160]), which brings

privacy concerns. SROS2 restricts users from reading such information from unauthorized topics, thus protecting the privacy of topics and relevant nodes.

4.3 Threat Model

4.3.1 System Assumptions

We consider a distributed MRS where a number of robots collaboratively work on one workload under the guidance of a centralized Ground Control Station (GCS). The system is developed with ROS2 and fully secured by the SROS2 modules. We assume all the configurations are set correctly with the following properties: (1) There exists one physical controller serving as the system owner of the MRS. It defines the system functions through userland codes, and also defines the access control policies for each robot that joins the system. Robot users only have local privilege to control their own robots. (2) A trusted CA is controlled by the system owner and generates unforgeable digital certificates for all the nodes within the MRS. These certificates are distributed to robots by the system owner securely. The system owner has the capability of remotely updating the certificate files stored on the robots at runtime. (3) Network traffic is properly encrypted by the DDS cryptography plug-in. (4) The system owner correctly implements the Mandatory Access Control (MAC) [161] policies by creating the permission files following the SROS2 standards [159].

4.3.2 Adversary's Capabilities

Following previous works on robotic security [162–164], we assume that one robot in the MRS is malicious and fully controlled by the adversary. This assumption is realistic due to several reasons: (1) there are software vulnerabilities and bugs in the robotic applications [100], which can be exploited by the adversary to intrude into the system and take full control of a robot. (2) ROS2 has its open-source platform that allows developers over the world to upload and share their function packages [88]. Unfortunately, there is no security check on the submitted code, and an adversary can publish malicious packages for other developers to download [165]. (3) Many cloud providers offer cloud-robotic services (e.g., AWS RoboMaker [144], Google Cloud Robotics [166]) to deploy robotic applications across the cloud and local robots. Robots from different parties and locations will be connected and coordinated by the cloud to complete the tasks. It is highly possible that some party is malicious and introduces an adversarial robot into the system, which tries to attack other robots via the interaction with the cloud.

The adversarial robot attempts to invalidate the SROS2 security features (especially the access control mechanism) and execute malicious operations in the MRS. These include (1) retrieving restricted information from unauthorized topics, (2) retrieving private node and topic configuration information, and (3) sending malicious messages to unauthorized topics.

The adversarial robot can perform arbitrary operations locally. However, it has the following limited capabilities when communicating with other actors in the system due to the presence of the SROS2 security mechanisms. (1) Due to the presence of SROS2, it can only communicates with the GCS and other robots by publishing and subscribing to relevant ROS2 topics using the functions defined by the ROS2 client library with valid security files. (2) It cannot forge digital certificates for authentication or break the encryption. However, it has the ability to read and use the certificates installed in its own robot. (3) It can passively eavesdrop all network traffics in its wireless communication range by switching its wireless adaptor to the *promiscuous* mode. This is feasible on vast majority of robots' on-board computers.

4.4 Methodology of Investigating ROS2

We introduce a methodology to thoroughly inspect the security of ROS2 implementation. It consists of four steps (Figure 4.2). (1) We first abstract the key events related to the network communication from the ROS2 and SROS2 source code (Section 4.4.1). (2) We describe the ROS2 system with the formal language CSP# [167] by modeling the nodes, participants and their communications (Section 4.4.2). (3) We formalize the desired security requirements, and perform model checking on the constructed model under these requirements (Section 4.4.3). The model checker generates possible counterexamples, which are the system states that



FIGURE 4.2: Methodology Overview

violate the requirements. (4) We analyze the counterexamples, summarize the vulnerabilities of SROS2 modules, and further verify their exploitability (Sections 4.5 and 4.6).

4.4.1 ROS2 Abstraction and Modeling

A typical ROS2 workload comprises three basic entities: nodes, participants, and the system owner. They interact with each other through a series of function calls to take actions, including policy updates, message communication, etc. The first step of our methodology is to identify the interactions between these entities and abstract them into a series of events that can be formally described. This approach enables formal verification of the abstracted system, but faces two main challenges. First, it is difficult to accurately identify the function call traces related to communication from the ROS2 source code. ROS2 is a massive system at three implementation levels (high-level API, RCL and RMW) with more than 500k lines of code in a mix of Python, C++ and C languages. Apart from core components for robot communication and control, it also contains numerous feature modules such as ROS1 adaptation, user experience enhancement, etc. Second, the implementation of inter-node communication processes also involve various inner-node functions, such as validating the userland code². These functions are

 $^{^2\}mathrm{For}$ instance, user-specified node name will be examined by both RCL and RMW to ensure its uniqueness.

redundant in modeling the communication structure since we focus on the security issues of ROS2 caused by the inter-node actions.

To address the above challenges, we adopt code property graph (CPG) [148] to represent the code structure, shortlist critical functions related to communication, and abstract the key events. CPG is a graph representation that merges the abstract syntax tree, control flow graph and program dependency graph into one joint data structure. Our strategy contains three main steps.

(1). Key Function Identification. We first locate the code sections that process communication messages from the large ROS2 code base. This can be achieved by tracking the data flow that involves the *message* variables in the CPG.

(2). CPG Purification. Next we further purify the CPG by removing the redundant function nodes that handle the inner-node behaviors but do not contribute to internode communications. As discussed previously, ROS2 implements validation mechanisms to ensure the validity of userland code. The execution of these functions results in either (a) its caller function continuing to execute if no error is reported, or (b) terminating the caller function execution and throwing an error. Either way will not change the normal interaction relations between the communication-related functions. Therefore, we consider eliminating them from the graph for easier modeling. Since these input validation components do not change the interaction relations between the communication-related functions, they exhibit the same pattern in the control flow of the CPG representation: input validation function nodes have direct outgoing edges to the error handler function nodes, which then terminate the control flow. Leveraging this property, we can efficiently identify them by traversing the graph and examining the outgoing edges for each node. The purified CPG can then be constructed by removing the error handling nodes and joining the other nodes together. Figure 4.3a demonstrates an example of a code snippet in RCL for publisher node creation (rcl_publisher_init), which has two functions for userland code validation (rcl_node_is_valid and rcl_node_resolve_name). By removing these nodes, we can reconstruct the abstracted graph that only includes the communication-related functions in Figure 4.3b.

(3). Verification and Analysis. Now the CPG only contains key functions that directly control the interactions between ROS2 system entities. To ensure the correctness of the CPG, we locate the key functions in the ROS2 source code and



FIGURE 4.3: An example of identifying key functions in an RCL code snippet (rcl_publisher_init).

check if their call relations comply with the abstracted CPG callgraph. Then, we manually analyze these key functions to understand the ROS2 inter-node communication workflow. Since the complexity of the CPG has been greatly reduced through previous steps, it is feasible and efficient to conduct verification and analysis manually.

Implementation. We apply a robust parser Joern [148] to parse the source code of RCL and RMW, generate and purify the CPG. Specifically, we first construct the CPG of the functions related to communication, which contains 1283 nodes. We summarize the exception keywords ("*ERROR*", "*err*", etc.) based on ROS2 coding practice and use them to label the error handler functions for CPG purification. After deleting the nodes directly connected to them, we establish the final abstracted CPG that contains 89 function nodes. We analyze these functions and summarize 23 key functions which are critical for inter-node communication. More details of our implementation are available at [151].

We further analyze the key functions and their dependencies, and figure out the inter-node communication workflow, as briefed below. The system owner first passes the security files to the user, who then stores these files in a self-defined path. To create an ROS2 node, the user initializes RCL with the security file path, and calls the rcl_init_publisher or rcl_init_subscription function depending on whether it is a publisher or subscriber. This function triggers the participant initialization handler in RMW. RMW verifies the integrity of the security files in the provided security path, and loads the access control policies as DDS QoS



FIGURE 4.4: Partial diagram of the CSP# model for the ROS2 system (node i publishes/subscribes to topic j).

parameters. When the node publishes a message to a topic, the corresponding DDS participant calls the DDS API with this message. When a subscriber node subscribes to an ROS topic, its participant subscribes to the corresponding DDS topic so that it can receive any messages published to that topic. In this manner, ROS2 translates userland code to a complete communication structure, while the network-level communication is handled by DDS.

4.4.2 Model Construction

Following the prior works [168, 169], we use CSP# [167] to describe the ROS2 communication system. It is an extension of CSP (Communicating Sequential Processes) [170] that mixes high-level operators with low-level programs for efficient modeling and verification of software systems with concurrent events. This makes it suitable for modeling the abstracted ROS2 system with concurrent node communication processes. Based on the event abstraction in Section 4.4.1, we define three types of processes: *owner_proc, node_proc* and *parti_proc*. Figure 4.4 shows the abstracted diagram of our CSP# model. Below we breif the construction of the model for an ROS2 system with N nodes and M topics, while the detailed formal description of each process is available in our supporting material [151].

(1) *owner_proc* process models the system owner that defines the access control policies and updates them to the nodes through security files. Each security file stores the access control rules for at least one node, and is modeled in an array owner_access: for a giving node i and a topic j, the Boolean vector owner_access[i, j] = [x, y, z] denotes if this node has publishing (x) and subscription (y) permissions, and knowledge of the topic's configurations (z). Each security file has a path (denoted as *path*) known by its corresponding node(s). Then owner_proc stores *path* to the access channel, denoted as *acc_chl*. There can be N channels in the system, with each one associated to a node.

Nodes and participants should obey the access control policies defined by the system owner. For clear representation, we let $pub_{ij} = owner_access[i, j][0]$ and $sub_{ij} = owner_access[i, j][1]$ to denote the publishing and subscription access of node *i* to topic *j*.

(2) node_proc(i) process models ROS2 node i. It first initializes itself by loading the security files from the user-defined path, and initializes a participant with the loaded contents. The node does not directly handle the contents of the security file in this step. After initialization, the node can (i) re-initialize itself with a new security file path and the corresponding participant; (ii) publish messages to the participant via an internal publishing message channel; or (iii) subscribe messages from the participant via an internal subscription message channel. Note that while access control is defined at the node level, SROS2 does not enforce access control over the nodes, but relies on DDS to regulate the participants corresponding to the nodes. Thus nodes can freely execute the publishing/subscription functions to arbitrary topics, but the corresponding participants will get rejected if they do not have the proper access.

(3) $parti_proc(i)$: this process models the participant created by node *i*. Upon initialization, the participant verifies the integrity of the security file contents provided by the node. It then retrieves the access control policies from the file and saves them into the corresponding internal access channel if the security file is valid. Then, it can (i) retrieve the messages from the internal publishing message channels and send them to the topics in the global data space of the DDS system; (ii) use its identity certificate to retrieve the messages of corresponding topics from the transport protocol and send them to the internal subscription message channels; or (iii) update the access control rules.

4.4.3 Model Checking

The above formal model enables the security checking of given security requirements and identification of possible violations via a model checker. If the model violates any security requirements, the model checker can automatically generate a counterexample, an execution trace that leads to the violation.

Security requirements. To identify the potential vulnerabilities in the ROS2 implementation, we first describe the desired security requirements for the system. These requirements are summarized from *ROS2 Robotic Systems Threat Model* [149], an official document describing the security goals, assets, and attack vectors in robotic systems. Following the previous work [35], we adopt the requirement engineering [171] technique to manually interpret the document. By mapping the security goals to the assets accessible to MRS participants, we conclude six security requirements for the system. Specifically, **R1** and **R2** are for system completeness, which ensures that all system entities participate in the communication process. **R3** to **R6** describe the security and privacy of the system entities. For each summarized requirement, we further describe it with the LTL (linear temporal logic) [172] formula. Let \Box , \Diamond , and U be the temporal operators "always", "eventually", and "until"; \land , \lor and \rightarrow be the logical operators "and", "or", and "implies". The security requirements for the ROS2 system can be formulated as below.

- (R1) Each node in the system has access control rules to at least one topic, either for publishing or subscription. Let $npug_{ij}$ and $nsub_{ij}$ be the publishing and subscription access of node *i* to topic *j*, then $\Box \wedge_{i=0,\dots,N-1} \sum_{j=0}^{M-1} (npub_{ij}+nsub_{ij}) >=$ 1.
- (R2) Each topic is accessible to at least one node to publish messages and at least one node to subscribe messages, i.e., $\Box \wedge_{j=0,\dots,M-1} \sum_{i=0}^{N-1} npub_{ij} >= 1$ and $\Box \wedge_{j=0,\dots,M-1} \sum_{i=0}^{N-1} nsub_{ij} >= 1$.
- (R3) The access control rules to message publishing of a participant should always be the same as the one declared by the owner.Let pub_{ij} and $ppub_{ij}$ be the systemdefined publishing access of node *i* to topic *j*, and the access at the participant level, then we have $\Box \wedge_{i=0,\dots,N-1;j=0,\dots,M-1} pub_{ij} == ppub_{ij}$.
- (R4) The access control rules to message subscription of a participant should always be the same as the one declared by the owner. Let sub_{ij} and $psub_{ij}$ be

the system-defined subscription access of node *i* to topic *j*, and the access at the participant level, then we have $\Box \wedge_{i=0,\dots,N-1;j=0,\dots,M-1} sub_{ij} = psub_{ij}$.

(R5) A participant *i* can publish (resp., subscribe) to a topic *j* only when $ppub_{ij} ==$ 1 (resp., $psub_{ij} ==$ 1) and the buffer of channel topic[j] is not full (resp., empty). Let p_msg_{ij} and s_msg_{ij} be Boolean variables denoting whether participant *i* publishes and subscribes to topic *j*. Let call(x, chl) be querying the buffer information of a channel chl, cfull and cempty denoting whether the channel is full or not. Then we have $\Box \land_{i=0,\ldots,N-1;j=0,\ldots,M-1}$ ($p_msg_{ij} == 1 \rightarrow$ $ppub_{ij} == 1 \land call(cfull, topic[j]) == False) \land (s_msg_{ij} == 1 \rightarrow psub_{ij} ==$ $1 \land call(cempty, topic[j])$

== False), where *call(operation, name)* is a static method to query the buffer information of a channel in the model checker.

(R6) When a node *i* legally subscribes to a topic *j*, it can only access the messages sent to the topic by legal nodes, but no other information of the topic's publishers. Let I_j be the nodes that have access to publish messages to topic *j*, g(i, k) be a Boolean value denoting whether the message subscribed by node *i* is equal to the message published by node *k*, and f(i, k) be a Boolean value denoting whether node *k*. Then we have $\Box \wedge_{j=1,2,...,M}$ $(nsub_{ij} == 1 \wedge \prod_{k \in I_j} npub_{kj} == 1) \rightarrow (\sum_{k \in I_j} g(i, k) == 1 \wedge \sum_{k \in I_j} f(i, k) == 0).$

Implementation. Without loss of generality, we apply the popular Process Analysis Toolkit (PAT) tool [173] to automatically verify if the abstracted CSP# model in Section 4.4.2 satisfies the above security requirements. Particularly, we construct the system based on the SROS2 sample project *chatter* [159], which has two nodes and two topics. This project is selected for two reasons. First, it involves the complete message publishing and subscription process in a well-defined communication structure. Since ROS2 communication is node-to-node basis, increasing the number of nodes and topics does not necessarily increase the complexity of the checked model. Second, this project has the native security implementation developed by ROS2 official. As people develop projects following ROS2 examples, the default misconfigurations in this project can be inherited to other community projects. Thus, we consider this model to be adequate and suitable for identifying vulnerabilities. We implement the concrete system model and initialize the system state based on the project's default security configuration. By verifying the model against the security requirements, we successfully identify multiple counterexamples in ROS2. Since the formal model is constructed strictly based on the key functions from the ROS2 implementation, all modeled processes and variables can be mapped to concrete objects in the source code. This enables us to quickly examine the related functions in the ROS2 implementation once a violation is detected, and identify the vulnerabilities led by the counterexamples. We analyze these vulnerabilities and demonstrate the exploits in Sections 4.5 and 4.6.

4.4.4 Discussion

While we select the *chatter* project in the system modeling process, our methodology can be applied to any ROS2 project and extended to other systems. This is because our strategy abstracts the ROS2 client library and DDS middleware into formally described processes and events. Fundamentally speaking, ROS2 projects are different only at the userland code level, which calls the low-level functions in different orders and quantities. We can easily model another ROS2 project by changing the number of topics, nodes, participants, and their publishing/subscription relationships.

We design our model checking approach to achieve *soundness* (i.e., each reported violation is indeed a reachable vulnerable system state) instead of *completeness* (i.e., identifying all the possible violations within the system). This is because our system modeling is parameterized by the number of nodes and topics, and it is impossible to achieve completeness due to the undecidability of parameterized system verification problem [174]. Thus, we follow the conventional approaches [35, 163, 175] to aim for soundness instead of completeness. The proposed model abstraction through node elimination results in a certain level of inaccuracy and may leave some vulnerabilities undiscovered. However, this process does not change the interaction relations between the communication-related functions, and thus guarantees the soundness of our approach.

Besides, we follow the official ROS2 threat model [149] to identify the vulnerabilities caused by false interactions between entities within the system. There exist some vulnerabilities beyond the scope of this threat model, and our methodology will fail to detect them. For instance, we do not consider the function-level vulnerabilities such as improper input sanitization vulnerabilities. Also, we do not consider implicit information leakage via side channels. During our manual analysis, we indeed find one such network side channel in ROS2: when a message is published to a topic, the ROS2 DDS identifies the receiver participants, and sends the message to each one separately. Since the message is the same, the network packets to each participant have the same source IP address, similar packet lengths, and very close timestamps. This allows an adversary to infer sensitive information about other nodes and topics by analyzing the network traffic, even it is encrypted by SROS2. How to formally discover such kinds of vulnerabilities is orthogonal to this work, yet an important direction to explore in our future works.

4.5 Security Vulnerabilities in ROS2

We analyze ROS2 and SROS2 implementations with the proposed methodology. Specifically, we examine the three most used and maintained ROS2 versions according to ROS Metrics [176]: ROS2 Galactic [177], Foxy [5] and Eloquent [178]. We successfully identify four vulnerabilities that exist across all versions of ROS2 and SROS2 implementations. In the rest of this chapter, we select ROS2 Foxy [5] distribution, the most mature and widely used ROS2 version as our target, while our findings also apply to the other ROS2 versions. We present the counterexamples, model checking outputs as well as our analysis on the minor differences between ROS2 versions in our supporting material [151].

4.5.1 V1: Permission File Replacement

The first vulnerability is caused by violations of security requirements $\mathbf{R3}$ and $\mathbf{R4}$ in Section 4.4.3, where a malicious node can bypass the access control policies and publish or subscribe to unauthorized topics. The root cause of this vulnerability is a ROS2 design flaw, where the adversary can abuse the local privilege to incur synchronization failures of access control policies.

ROS2 enforces access control policies by passing the SROS2 security files to the DDS security plug-in through APIs (Section 4.2.4). This requires the access control



FIGURE 4.5: Unauthorized publishing/subscription through the vulnerabilities of V1 $(\mathbf{0})$ and V2 $(\mathbf{2})$.

policies to be updated and synchronized in three layers of the ROS2 architecture. (1) System policies are created by the system owner. They are declared in the signed permission files and distributed to the corresponding robots. (2) SROS2 policies are loaded by the SROS2 modules. Each robot declares the directory that contains the security files. The SROS2 modules verify the validity of these security files, and then pass them to the DDS layer through API calls. (3) DDS QoS policies are loaded by DDS QoS security plug-ins. It enforces access control on the DDS participants and the ROS2 nodes.

Ideally, access control policies in the three layers should be timely synchronized: once the system owner updates the policies during the workload execution, the corresponding security files should be updated on the robots; the policies declared in the security files are then loaded by the SROS2 modules and passed to the DDS participants immediately. However, we discover that an adversary could abuse the design flaw of the SROS2 permission file revocation process to interrupt the synchronization process, thus invalidate the SROS2 access control and further attack the system.

As introduced in Section 4.2.3, the permission files store the access control policies. When a node publishes or subscribe to a topic, it provides the corresponding permission file stating the proper access to the topic. SROS2 rejects the action if the permission file does not contain a valid digital signature signed by the CA, thus enforces access control policies. However, SROS2 does not actively revoke the old permission files when the access control policies are updated. Instead, it simply replaces the old files with the new ones, or sets up a new directory to store the new files and changes the corresponding load pointers. Since a robot has *read* and *write* accesses to all the local files, an adversarial robot can store the expired permission files in a backup keystore directory, and then pass them to SROS2 instead of the updated one (**0** in Figure 4.5). These expired files can pass the CA signature verification and are loaded for policy enforcement. By doing so, the adversary can obtain publish and subscription access to some restricted topics, even its permissions have been explicitly denied in the updated files. A direct mitigation towards this vulnerability is active certificate revocation. By revocation of expired certificates and permission files, the adversary cannot bypass the SROS2 verification with the old permission files. ROS2 has taken our suggestion and added documentations on manual certificate revocation methods in ROS2 rolling [179], the feature testing ROS2 version. However, an complete and automated solution is not implemented yet.

4.5.2 V2: Outdated Node Service

Similar to V1, the second vulnerability also violates $\mathbf{R3}$ and $\mathbf{R4}$ in Section 4.4.3. The root cause of this vulnerability is also the synchronization failures of access control policies in different SROS2 layers caused by a ROS2 design flaw, where the DDS QoS policies can only be updated during participant initialization. An adversarial node can leverage the loophole in SROS2 function calls to refuse the update of access control policies on the corresponding participant.

Particularly in ROS2, node publishing and subscription are two independent actions controlled by the robot. When a node publishes or subscribes to a topic, RMW calls the DDS APIs and the SROS2 security files are loaded to the corresponding DDS participant as the QoS policy parameters. The participant then creates a data reader or writer following the QoS policy. When the system owner updates the access control policies of a node, the robot is required to relaunch the node's publishing or subscription service so the new policies can be updated to the DDS. This design is vulnerable because an adversarial robot can refuse to restart the services of its nodes (@ in Figure 4.5), so it can continue accessing the topics, which are supposed to be revoked during permission updates.

4.5.3 V3: Default Mis-configuration

ROS2 provides a GUI plugin rqt_graph [180] to visualize the publishing-subscription relations between nodes and topics for the debugging purpose. To protect the recipient privacy [150], SROS2 disables this function and allows the system owner to configure the discoverability of each node and topic. By default, topics, nodes, and the publishers/subscribers to each topic are hidden after enabling SROS2. However, we identify one vulnerability that allows an adversarial robot to obtain sensitive information of other nodes and topics, i.e., violating the requirement **R6** in Section 4.4.3. This vulnerability is caused by a default misconfiguration in the SROS2 implementation that contains insecure DDS QoS parameters.

We find that SROS2 has some default mis-configurations that could cause crossnode information leakage. For instance, in the implementation of the SROS2 settings for RTPS DDS [181], the default option for the message communication is sign without encryption. A signed DDS message does not hide the its publisher/subscriber participants' information, and the adversarial node can read them to infer the network communication topology. This vulnerability was also reported by other developers as CVE-2019-19625 and CVE-2019-19627. The ROS2 community developed patches to fix them [182]. However, they are not merged into the ROS2 mainstream, making the current version still vulnerable.

4.5.4 V4: Permission File Inference

Similar to V3, this vulnerability can also cause cross-node information leakage, but from the permission files. Its root cause is the insecure coding practice without the consideration of the principle of least privilege. While the integrity of an SROS2 permission file is protected by its digital signature, its confidentiality is not guaranteed. ROS2 assumes that each node protects the confidentiality of its own files including the permission files, so all these files are in cleartext. Ideally, creation of the permission files should follow the principle of least privilege [183]: every node in the system can only access the topics necessary for its legitimate purpose. Unfortunately, we discover that a majority of permission files in the official ROS2 projects, including the SROS2 sample publisher-subscriber system [184] and the Open Robotics RMF Demos project [185], disobey this principle and contain excessive information. The adversary robot can easily read the sensitive attributes of other nodes directly from its own permission file including their security configurations and topic access. Workloads which follow or adopt these permission file templates from official projects could suffer severe privacy threats.

It is worth noting that this vulnerability is fundamentally different from the previous ones. While V1 to V3 target the underlying communication protocols, V4 originates from the owner-specified permission files. It can be mitigated by carefully defining the permissions with the principle of least privilege. So in the rest of this chapter, we do not consider this vulnerability.

4.5.5 Discussion

The severity of these vulnerabilities is reflected in not only the possible consequences, but also their stealthiness. The existing ROS2/SROS2 mechanisms cannot effectively detect attacks from these vulnerabilities. Specifically, (1) in the current ROS2 communication protocol design, messages do not contain publisher information, and topics are typically designed to process homogeneous types of message without the capability and necessity of tracking the message sources. Therefore, when the adversarial robot exploits the unauthorized publishing/subscription vulnerabilities (V1 and V2) to actively send malicious messages, it is difficult to detect such an anomaly. One possible solution is to actively inspect messages in the network layer, log their sender/receiver IP addresses and construct their publishingsubscription relations. By checking this relation against the system communication graph generated by the ROS2 built-in tool rostopic [160], we can detect if a robot is sending unauthorized messages. However, as mentioned in Section 4.2.4, SROS2 prohibits the use of this tool. The system owner cannot enable this tool by sacrificing the privacy. (2) Exploiting V3 is a passive process, and the adversary does not need to actively communicate with other topics. It is also hard to monitor the occurrence of this attack. (3) SROS2 does not provide any logging features. Function calls and communication messages are not accountable, making it difficult to pinpoint the malicious actor after system failures.



4.6 Vulnerability Exploitation

We validate the exploitability of the discovered vulnerabilities with various realworld ROS2 workloads in both simulation environments and physical testbeds. We show that exploiting these vulnerabilities could cause severe consequences, including but not limited to terminating the workloads, crashing the victim robots and damaging the surroundings, and stealing users' private information.

4.6.1 Simulation Setup

ROS2 Workloads. We select three open-source MRS workloads based on ROS2 from the Robotics Middleware Framework (RMF) project [185], which is developed by Open Robotics [29]. The project demonstrates the usage of heterogeneous robot teams (nine types of robot in total) in 5 real-world environments with the ROS2 platform. In each workload, robots are controlled by the GCS task planner to collaboratively work on different types of tasks. We select three workload environments: airport terminal, clinic world and campus. Details about these environments are available online at [151].

By design, tasks in the three workloads are split into simpler subtasks that can be completed by one robot to increase the overall system efficiency. The GCS allocates tasks by considering the robot status (e.g., location, battery life, etc.) and system goal. Therefore, each robot works on various subtasks during the execution of the workload, and requires different permissions to access different system resources that vary with the task. **Configurations.** We deploy the above workloads in the Gazebo simulator [86] and ROS2 Foxy distribution. All workloads are set up by following the default configurations listed in their project sources. We deploy the SROS2 security features to all the workloads based on the threat model in Section 4.3.

4.6.2 Simulation Evaluation

It is worth highlighting that each discovered vulnerability is general to affect different ROS2 workloads with different attack consequences. Without loss of generality, we adopt one workload to demonstrate each vulnerability and one possible consequence. Below we describe the exploitation procedures.

V1: Permission File Replacement. As described in Section 4.5.1, an adversary can pass expired permission files to SROS2 to bypass the access control. Specifically, the adversary can backup the permission files to a local directory which is not accessible to the system owner. After each permission file update, it can replace the latest permission file with any one of the old permission files that contains the permission he needs. In this way, the adversarial robot can bypass the system access control policy, and access the unauthorized topics that was once assigned to it.

We implement a prototype-of-concept attack on the airport terminal workload. We show one possible attack consequence, where the adversarial robot can cause task completion failures by manipulating its access permission to unauthorized environments. As shown in Figure 4.6a, a *CleanerBot* with the task of cleaning the region $zone_{-1}$ only has access to the topics related to the resources in this region. When a robot completes this task, the GCS assigns a new task region and updates the permission file so that it only contains access to the topics about the new region, while previous access permissions are revoked at the same time. A robot can exploit **V1** to retain the old permissions and access to that an adversarial *CleanerBot* can eventually obtain the publish/subscribe access to all the topics required by the cleaning tasks, which include the access to control the operation of automatic doors in different cleaning regions as shown in Figure 4.6a. By sending the *close* command to the door control topic, the adversarial robot hinders the movement of other robots and causes workload execution failures.

V2: Outdated Node Service. To retain old permissions, the adversarial robot can also refuse to re-initialize the nodes after the policy update (Section 4.5.2). We design an attack on the campus workload, where multiple robots deliver items using GPS localization. Each robot streams its location to its corresponding adapter topic so that the GCS can coordinate the overall delivery task accordingly. By exploiting V2, an adversarial robot can retain the publishing access to the previous adapter topic regardless of its current legitimate publishers. It can then send forged GPS data to this topic for the GCS to process. As a result, the task controller will calculate the path based on the spoofed GPS location provided by the adversarial robot as long as it sends fake messages with higher frequency to overwhelms correct messages from the benign robot. Figure 4.6b shows one possible attack consequence from our simulation experiment: the adversary carefully selects a spoofed location so that the GCS generates a wrong path (blue) and assigns it to the benign robot. The benign robot will follow the trajectory (red) but from its actual location, and crash into the obstacles.

V3: Default Mis-configuration. The adversary can leverage the default misconfiguration in DDS to obtain critical information (Section 4.5.3). In the default DDS (eProsima Fast DDS), the variable *rtps_protection_kind* defines whether the RTPS message is protected by encryption, which is 'SIGN' by default. Therefore, we can exploit the vulnerability of CVE-2019-19625 [186] with the ROS2 robot fingerprinting tool Aztarna [187] to list all nodes and topics. By regularly examining the map resource topics subscribed by each robot, the adversarial robot can record the locations of all other robots and infer their tasks. For example, Figure 4.6c shows the attack result in the RMF clinic world workload. The adversarial robot captures the location of other robots every minute. Based on such information, it can infer that robot 2 is patrolling between the nurse rooms at level1 and level2; robot 3 is performing guidance tasks between the counter and the waiting area; robot 4 is delivering items between different locations. In real-world scenarios, robot tasks can be closely related to users' personal information. Various works have highlighted that robotic systems (e.g., surgical robots) in hospitals are vulnerable to cyber attacks [188, 189] and have critical privacy issues [190, 191]. Exploiting V3 provides a new attack opportunity to steal personal information in such sensitive scenarios.

4.6.3 Physical Evaluation

We further validate these vulnerabilities in a physical testbed, which proves it is practical to exploit them to cause severe consequences.

Physical testbed setup. We set up a cloud-based MRS workload from Amazon RoboMaker [192], developed by AWS Robotics [144] and JdeRobot [193]. It considers the operation environment in the Amazon warehouse. We implement this environment with three physical Turtlebot 3 Waffle Pi robots [87] and AWS Elastic Compute (EC2). More details of our physical setups and configurations can be found online at [151].

Evaluation results. Following the attack processes described in Section 4.5, we implement the exploits to V1, V2 and V3, respectively. After gaining unauthorized access to different resources through the exploitation, the adversarial robot can cause various attack outcomes. Here we only demonstrate some possible consequences.

For V1 and V2, we observe that the adversarial robot can directly cause system failures and robot crashes similar to the simulation results in Section 4.6.2. Particularly, the GCS relies on the real-time position information provided by robots to calculate their trajectories and ensure no collisions during the workload. However, The adversarial robot can easily trick the GCS to design a wrong trajectory by constantly sending spoofed location messages to the topic belonging to other robots. In practice, we observe that the victim robot crashes into walls and other robots when the local obstacle avoidance function is not enabled. When we manually enable this function, the victim robot just stops functioning because obstacle avoidance contradicts the commands given by the GCS. For V3, we find that it leads to sensitive information leakage similar to the results in Section 4.6.2. Exploiting V3 allows us to generate the nodes and topics communication topology, which directly reveals the number of robots, current tasks and system control structure.

4.7 A General Defense Solution

It is necessary to fix the above threats and make SROS2 really secure. While changing the ROS2 underlying protocol from DDS to other established ones seems to be feasible, it does not address the issues. This is because V1 to V3 are rooted in SROS2 design flaws that violate the security considerations in MRS, which are independent of the underlying protocol. There exist straightforward solutions to mitigate each vulnerability individually. For instance, V1 can be mitigated by updating a node's certificates whenever its access policy is updated; for V2, the system owner can enforce all participants to temporarily leave the system and then rejoin during policy update; V3 can be mitigated by correcting the default misconfigurations. However, these ad-hoc solutions could bring inconvenience for the workload execution and system maintenance. Furthermore, V1 and V2 cannot be fully patched due to the physical limits in the MRS scenarios. The system owner cannot constantly monitor all robots' security configurations and function execution at runtime considering the unstable network in real-world workloads.

4.7.1 Design Rationale

We aim to design a unified defense solution, which could fundamentally address the identified vulnerabilities in SROS2. The main goal is to refine the ROS2 communication process to securely and efficiently distribute messages among participants. Specifically, it should exhibit three properties. (1) Security: the ROS2 access control is expected to be correctly enforced, and the confidentiality of nodes' and topics' information should be strongly preserved. (2) Efficiency: the overhead of the solution should be acceptable in the MRS workload context. (3) Compatibility: the solution can be integrated to ROS2 without any additional infrastructure. Attribute-based encryption (ABE) solutions are mature and widely applied to enforce access control in various types of systems [194–196] including DDS [197]. However, these primitives are inefficient because they provide fine-grained access control with redundant functionalities in the context of ROS2.

To this end, we introduce a lightweight solution specifically for robotic systems with the private broadcast encryption (PBE) primitive [150] as the underlying technology. Compared with other generalized encryption systems, our method is customized to ROS2 to meet the design requirements so that it is very efficient and fully compatible with ROS2 without modifying its underlying source code. If the ROS2 system is not equipped with SROS2, our method can provide the same mandatory access control. If SROS2 is enabled, our method can prevent all the

identified vulnerabilities in Section 4.5. Our solution can defeat a stronger threat model than the one in Section 4.3: it can protect the system even if there exist multiple adversarial robots that collude and exchange information with each other. We justify the security of our solution through rigorous proof, formal verification and physical experiments (Section 4.7.4).

4.7.2 Methodology Description

We incorporate the PBE scheme into the ROS2 communication system. This scheme uses public key encryption with key indistinguishability under the chosenciphertext attacks (IK-CCA) [198] to encrypt the ciphertext component for each recipient. It then generates a random signature and verification key for a one-time, strongly unforgeable signature scheme. It includes the verification key in each public key encryption and then signs the entire ciphertext with the signing key. To be precise, let G be a group with g as the generator, where the computational Diffie-Hellman problem (CDH) [199] is hard but the decisional Diffie-Hellman problem (DDH)[200] is easy³. H is a hash function mapping $H : G \to \{0, 1\}^{\lambda}$ for a security parameter λ modeled as a random oracle. Hence, given a strongly correct IK-CCA public key encryption scheme (**Int, Keygen, Enc, Dec**), a strongly existentially unforgeable signature scheme (**SigGen, Sig, Ver**), and a pair of semantically secure symmetric key encryption and decryption algorithms (**E**, **D**), the PBE system can be described as follows.

- 1. **Setup**(λ): Run $I \leftarrow Int(\lambda)$ to get the global parameter I.
- 2. **Keygen**(*I*): Given *I*, generate $(pk_i, sk_i) \leftarrow \text{Gen}(I)$ for each node $i \in N$. Also generate key pairs $(vk_i, vsk_i) \leftarrow \text{SigGen}(I)$ for each node $i \in N$ for signature and verification processes. Then, choose a random exponent α_i and let $pk'_i = (pk_i, g^{\alpha_i}), sk'_i = (sk_i, \alpha_i)$. (pk'_i, sk'_i) and (vk_i, vsk_i) are sent to node *i* and publish pk'_i .
- 3. Encrypt(P, m): Consider that node k with signature key and verification key (vk_k, vsk_k) wants to send a message m to nodes in a selected subset $P \subset N$. Node k runs the following procedures:

³For formal definitions of these NP problems, please refer to [150].

- 3.1. Randomly choose a one-time symmetric key K used to encrypt m.
- 3.2. Randomly select a one-time exponent t and set $W = g^t$.
- 3.3. For every node $i \in P$, compute

$$c_{pk_i} \leftarrow \mathbf{H}(g^{t\alpha_i}) \| \mathbf{Enc}_{pk_i}(vk_k \| g^{t\alpha_i} \| K)$$

- 3.4. Let C_1 be the concatenation of the c_{pk_i} ordered by their values of $\mathbf{H}(g^{t\alpha_i})$.
- 3.5. Encrypt m as $C_2 \leftarrow \mathbf{E}_K(m)$.
- 3.6. Generate the signature for the above ciphertext as $\mu \leftarrow \operatorname{Sig}_{vsk_k}(W \| C_1 \| C_2)$.
- 3.7. Broadcast ciphertext $C = \mu ||W||C_1||C_2$ to all nodes.
- 4. **Decrypt** $((sk_j, \alpha_j)_{j \in N}, C)$: Each node $j \in N$, parse $C = \mu ||W||C_1||C_2$ and $C_1 = c_1 ||...||c_p$, then run the following procedures:
 - 4.1. Calculate $r = \mathbf{H}(W^{\alpha_j}) = \mathbf{H}(g^{\alpha_j t}).$
 - 4.2. Find c_l such that $c_l = r || c$. If it does not exist, return \perp and stop.
 - 4.3. Compute $d \leftarrow \mathbf{Dec}(sk_j, c)$. If d is \perp , return \perp and stop. Otherwise, parse d as $vk_k ||u|| K$.
 - 4.4. If $u \neq W^{\alpha_j}$, return \perp and stop.
 - 4.5. If $\operatorname{Ver}_{vk_k}(W \| C_1 \| C_2, \mu)$, return $m = \mathbf{D}_K(C_2)$; otherwise, return \perp .

The above scheme can be adopted in ROS2 with the following steps.

- 1. The CA generates pairs of certificates and private keys for nodes in the ROS2 system with **Gen**. Then the system owner updates the certificate/key pairs to each node.
- 2. The system owner formulates the access control policies and updates them to all nodes. It then passes access control knowledge to nodes accordingly. Each node knows the topics to publish/subscribe to. A node with publishing (resp., subscription) access is provided with the public (resp., verification) keys of its subscribers (resp., publishers) N.
- 3. When a node publishes a message m to a selected subset groups $P \subset N$, it encrypts the message with the public keys of the recipient nodes in P following

the encryption function **Encrypt** described above and publishes the ciphertext C to the topic. While all the nodes in N can subscribe to the topic, only the receiver nodes in P have proper read access and can extract m from C using the decryption function **Decrypt**.

- 4. After a node extracts m through the decryption function, it examines if the verification key vk obtained from the decryption process is in the verification key list provided by the system owner. Otherwise it discards the message m because it comes from an untrusted node.
- 5. When the access control policies need to be updated, the system owner updates the new access control knowledge to the related nodes by encrypting the knowledge and publishing it to the nodes accordingly following step **3**.
- 6. When a new node is introduced into the system after initialization, the CA generates key pairs and the system owner updates them to the node accordingly. The system owner then broadcasts the public key of the new nodes together with the updated access control policies to the existing nodes following step **3**. The same process applies to the node revocation scenario.

4.7.3 Implementation

We implement the proposed defense as a lightweight Python3 package [151]. The system owner can set up our defense in an existing ROS2 workload with three steps. First, CA generates the public and private key pairs required by the selected Elliptic Curve Cryptography (ECC) scheme for all nodes in the system. This process is the same as certificate/private key generation process required by SROS2, so it is supported by the SROS2 command line tool without additional infrastructure for implementation. Second, the system owner installs the defense scheme. The encryption and decryption functions can be easily imported from the Python3 package. Third, each robot encrypts the message with the public keys of the intended receivers before sending it out. The receiver robots can decrypt ciphertext messages as long as they are in the receivers list.



FIGURE 4.7: Mitigating vulnerabilities with our defense.

4.7.4 Security Evaluation

We perform the security assessment of our defense in three aspect.

4.7.4.1 Theoretical Analysis

Given the strongly correct IK-CCA public key encryption scheme (Int, Keygen, Enc, Dec), a strongly existentially unforgeable signature scheme (SigGen, Sig, Ver), and a pair of semantically secure symmetric key encryption and decryption algorithms (E, D), the aforementioned PBE system has been proven to be secure under chosenciphertext attacks. We describe how the PBE holds the requirements in Section 4.4.3. Specifically, considering that node k wants to send a message m to nodes in a selected subset $P \subset N$, we have the following two theorems, where the first one is used to fix vulnerabilities V1 and V2, while the second one is used to fix V3. The complete proofs are available in our supporting materials [151].

Theorem 4.1. If node k is malicious, the above PBE system holds that node k cannot pretend to be other honest users to send ciphertext.

Theorem 4.2. If node k is benign, for any adversary \mathcal{A} , the above PBE system holds that \mathcal{A} cannot infer the identity of benign nodes in P. Particularly, if there is no malicious node in P, \mathcal{A} cannot obtain useful information about message m. **Remark**: The proposed defense scheme can protect the ROS2-based MRS against the identified vulnerabilities, as shown in Figure 4.7. For V1 and V2, the adversarial node can bypass the SROS2 access control and publish to unauthorized topics. However according to Theorem 1, PBE prevents it from pretending to be honest users for sending ciphertext. Thus, the receiver nodes could identify that the publisher is malicious. Similarly, for an adversarial node that exploits V1 and V2 to subscribe to unauthorized topics, it cannot decrypt the messages received from the topics and obtain any useful information according to Theorem 2. For V3, the adversarial node observes the network traffic and infers the secret information. With our defense, a node can subscribe to any topics yet only retrieve useful information from the authorized ones. Thus, the network traffics between any two nodes do not necessarily mean that they are exchanging valid information. So the adversary cannot gather sensitive information with V3. With these properties, our solution provides recipient privacy and the same mandatory access control as ROS2, so it can be implemented individually or on top of ROS2.

4.7.4.2 Formal Verification

We formally verify the security of the proposed scheme. First, we construct a formal model for the encryption scheme and perform formal verification with ProVerif [201]. No protocol weaknesses are identified when verifying the model against the security requirements. We then follow the model checking approach in Section 4.4 to verify the security of its integration in ROS2. Specifically, we identify the events that should be performed by the system actors as defined in Section 4.7.2. We then abstract them and describe them with CSP#, and extend the previously constructed CSP# model in Section 4.4.2 to describe the ROS2 system with the proposed defense solution. By verifying the new model with PAT, we confirm that the identified vulnerabilities have been fixed with no additional counterexamples generated.

4.7.4.3 Empirical Validation

We repeat the physical attacks launched in Section 4.6.3 with the same setups. After enabling our defense scheme, we observe that all three identified vulnerabilities are no longer exploitable. Specifically, for V1 and V2, our defense provides
the authentication service between the message sender and receiver. So even the adversarial robot can retain the old access permissions by replacing the permission files or refusing to restart the node service, it is still not able to access the unauthorized topics. For V3, our defense scheme prevents the adversarial robot from distinguishing valid communications from the invalid ones when monitoring the network traffic. Therefore, the adversary is not able to infer the communication topology using any fingerprint tools. It is worth noting that the proposed solution cannot defeat DoS attacks. In fact, how to design full defenses against DoS attacks from insiders in robotic systems is still an open problem [140, 202], because the adversary can use system knowledge to craft DoS messages that follow the protocol. Nevertheless, our design is less vulnerable to DoS attacks compared to other cryptographic solutions. Specifically, in step 4 of the scheme (Section 4.7.2), a node does not perform any decryptions if the received message is from an outsider (4.1), and only performs partial decryption (4.3) if the message is from an insider

4.7.5 Efficiency Evaluation

adversary, which increases the DoS difficulty.

We evaluate the performance and resource consumption of our solution using the physical testbed. Below we present the main experimental results, while the physical experiment setups and experimental data are available in our project website [151].

4.7.5.1 Performance Evaluation

We first measure the impact of the additional operations (e.g., encryption, decryption) on the performance of the MRS. We adopt the mainstream *ROS2 Performance Test* benchmark [203] developed by ApexAI [204]. and make necessary modifications adapt to our defense scheme. We deploy two Turtlebot robots to run the publisher node and subscriber node respectively, connected to the same local area network.

We compare four settings. (1) Normal: ROS2 without any security features; (2) SROS2: ROS2 with SROS2 enabled. (3) PBE: ROS2 with the proposed defense; (4) Both: ROS2 with both SROS2 and proposed defense. For each experiment, we



FIGURE 4.8: Encryption/Decryption time cost under different frequencies and message sizes.

execute the task for 60 seconds, and repeat it for 10 times to obtain the average results.

Evaluation results. First, we explore the average encryption and decryption cost of our defense for different message sizes and publishing frequencies. We vary the publishing frequency from 10 Hz to 100 Hz, and the cleartext length from 8 Bytes to 4096 Bytes, covering the common configurations in most robotic system components. The encryption and decryption overhead of the PBE scheme is shown in Figure 4.8. We observe that the cost of those operations is slightly increased with the message length: the average encryption/decryption time of a 4KB message is 6.4%/4.9% longer than a 8B message. We also observe the cost is slightly decreased with a higher publishing frequency. This might be due to the CPU Dynamic Voltage and Frequency Scaling (DVFS) optimization feature.

Second, we compare the end-to-end latency for the entire system with four security settings. We also select the above ranges of message sizes and publishing frequencies. The results are shown in Figure 4.9. Our solution introduces around 4ms latency for each communication. Compared with Figure 4.8, such cost is mainly from the encryption/decryption operations. In real-world robotic systems (especially the cloud-based), the network latency is much higher (in the order of seconds). Therefore an MRS is commonly designed to be delay-tolerant [205], and this overhead can be ignored. We further measure the data loss rate during transmission, and observe that our scheme causes less than 0.01% of data loss at 100Hz frequency with the message sizes of 1KB and 4KB. It happens during communication initialization, when the first few packets are not delivered to the subscriber.



FIGURE 4.9: Communication latency of four implementations with various message lengths and frequencies.

This "initial loss" is also observed by other works [206], and does not affect the system operation.

Third, we explore the scalability of the proposed defense. We examine the system latency in two experiment settings: (1) one publisher publishes to multiple nodes; (2) a number of nodes connected in series, where the intermediate nodes act as both publishers and subscribers. For each scenario, we vary the number of subscription nodes from 1 to 8 and message sizes of 64 and 4096 bytes. The publishing frequency is fixed at 20 Hz. The communication latency of different system configurations is shown in Figures 4.10 and 4.11. We observe that the end-to-end latency does not increase significantly when more nodes subscribe to one publisher. When nodes are connected in sequence, the latency increases linearly with the number of communication nodes. In practice, an intermediate node needs to process the incoming message to the next one. This process can compensate the overhead incurred by our solution. Overall, our defense does not incur significant latency compared to SROS2, and can easily scale to large systems.

4.7.5.2 Resource Consumption Evaluation

We measure the resource consumption in our defense, which is critical for robots with limited computing capability. We select and implement two most widely adopted MRS workloads: navigation [70] and exploration [71].



FIGURE 4.10: Communication latency of four implementations with one publisher and various numbers of subscribers.



FIGURE 4.11: Communication latency of four implementations with various numbers of publisher-subscriber pairs.

Evaluation results. We measure the runtime CPU and RAM utilization of the on-board processors on the Turtlebots, as shown in Figure 4.12. For the CPU usage (Figure 4.12a), the navigation and exploration workloads require 42.5% and 45.2% of CPU resources, respectively, and enabling SROS2 does not increase the CPU utilization significantly. With the proposed defense, the CPU utilization of these two workloads are increased to 62.9% and 72.3%, respectively. Such overhead is acceptable since the CPU cores are still not saturated. In real-world workloads, on-robot processors are under-utilized most of the time [207] because they should meet the performance requirement of the most computational extensive subtask, which only takes very little operation time. Our solution only takes the redundant computational power during the workload execution. Also, we believe the CPU utilization to C++, which is also supported by ROS2. For RAM utilization (Figure



FIGURE 4.12: Resource consumption of four implementations. 4.12b), the two workloads require 279.6 MB and 326.9 MB of memory. The defense increases the RAM consumption to 354.0 MB and 396.2 MB. This is far lower than the capacity of common robotic processors (e.g. 1GB RAM for the Raspberry Pi 3B+ model in our experiments).

Based on the above results, we conclude that the CPU and RAM utilization of our defense is acceptable on commercial robots with single-board processors. Note that it might cause performance issues when we implement this scheme on tiny robots with very limited computing resources (e.g., swarm robots). In the future, we will further optimize our implementation for these scenarios.

4.8 Related Works

Model checking. Model checking has been widely adopted to verify the correctness and security of systems [208–211]. Recently, researchers applied this strategy to verify robotic and autonomous applications, such as DoS vulnerabilities in connected vehicle protocols [163], safety properties of ROS-based robotic applications [212], hierarchical properties of swarm robot systems [213], security, liveness and priority of the DDS without considering the ROS2 implementation [214]. Different from the works which focus on either applications or individual components of the ROS/ROS2 system, we mainly target the fundamental implementations of the ROS2 security features.

Access control with cryptography. Barth *et al.* [150] proposed the first private broadcast encryption scheme to achieve identity-based access control among messages. Then, many variants (e.g., attribute-based encryption (ABE), puncturable

encryption) were proposed to achieve more precise access control with the attribute of the system participants [215–218]. For instance, Bethencourt *et al.* [216] developed a ciphertext-policy based ABE that allows tree-based access policies. Yu *et al.* [219] proposed a solution for indirect attribute and user revocation.

ROS2 and DDS Security. Previous works including [220] explore the efficient and automatic generation of SROS2 permission files for ROS2 projects. Other work [214] formally verify the security of DDS in ROS2. These works leverage existing SROS2 features and do not consider that adversaries could bypass SROS2 through its native vulnerabilities. Instead, we propose the first study over the security of ROS2 implementation. The vulnerabilities discussed in this work thus cannot be identified or patched by the previous solutions.

4.9 Conclusion

In this work, we perform a thorough and systematic security analysis about ROS2 with the DDS security features. We design a formal method to model the ROS2 system and security requirements. We identify four vulnerabilities in the implementation of ROS2, which can invalidate the security mechanism of DDS, and threaten the robotic workloads. To fundamentally address these issues, we design a practical and lightweight defense methodology with the private broadcast encryption. We have reported our discoveries to the ROS2 official and are working with them on possible mitigation. We hope these vulnerabilities can be fixed very soon to advance the secure development of robotic systems and applications.

As we move forward, it is crucial to recognize that the security challenges in robotics are not unique. Similar vulnerabilities can and do arise in other domains, such as web services. With this in mind, our attention shifts from robotic systems to a domain of increasing importance: the security of RESTful APIs in web services.

Chapter 5

Human-Interactive Testing of RESTful API Service

Just as robotic systems require rigorous security analysis, so too do web services, particularly those using RESTful APIs, which have become the most prevalent endpoint for accessing web services. Blackbox vulnerability scanners are commonly used for automatically detecting vulnerabilities in web services. However, these tools have significant limitations in RESTful API testing. Specifically, existing tools cannot effectively determine the relationships between API operations and lack awareness of the correct sequence of API operations during testing. These limitations hinder the tools from requesting API operations properly to detect potential vulnerabilities.

To address this challenge, we propose NAUTILUS, which includes a novel specification annotation strategy to uncover RESTful API vulnerabilities. The annotations encode the proper operation relations and parameter generation strategies for the RESTful service, which assist NAUTILUS to generate meaningful operation sequences and thus uncover vulnerabilities that require the execution of multiple API operations in the correct sequence. We experimentally compare NAUTILUS with four state-of-art vulnerability scanners and RESTful API testing tools on six RESTful services. Evaluation results demonstrate that NAUTILUS can successfully detect an average of 141% more vulnerabilities, and cover 104% more API operations. We also apply NAUTILUS to nine real-world RESTful services, and detected 23 unique 0-day vulnerabilities with 12 CVE numbers, including one remote code execution vulnerability in Atlassian Confluence, and three high-risk vulnerabilities in Microsoft Azure, which can affect millions of users.

5.1 Introduction

Representational state transfer (REST) has become one of the most popular standards for web service interactions [221, 222]. It has been adopted by many wellknown web service providers, such as Google [223], Microsoft [224], Wordpress [225], etc., to expose their digital services and assets via RESTful APIs. As RESTful APIs gain popularity, they become a common attack vector for the digital services and assets behind. According to a survey by Salt Security [226], 91% of the respondents experienced API security incidents in 2021. This survey also discloses that vulnerability is the most commonly encountered security issue. Thus, securing RESTful APIs is particularly important for service providers, and early detection of vulnerabilities is an important task to protect the web services.

Penetration testing is a popular technique adopted by many service providers to fulfill this task [227, 228]. This technique is also known as ethical hacking, which launches authorized simulated cyberattacks to find vulnerabilities in the service under test (SUT). Penetration testing can be performed manually or with automated tools. Compared with manual testing, using automated tools can not only save human effort but also yield stable testing results regardless of the experience and knowledge of the human tester. Currently, there are two widely used automated penetration testing tools for RESTful APIs, namely Open Web Application Security Project Zed Attack Proxy (ZAP) [229] and Web Application Attack and Audit Framework (w3af) [230]. To test a RESTful service, both ZAP and w3af utilize dictionaries of predefined attack payloads to request and check every single API of SUT. Although these two tools have successfully discovered many bugs in several RESTful services [231], they can only detect vulnerabilities that involve just one RESTful API operation. However, according to our empirical study of 609 vulnerabilities, 499(82%) of them require multiple RESTful API operations to trigger. This is consistent with the studies conducted by OWASP [232] and Rapid7 [233], both of which report that RESTful API vulnerabilities are fundamentally web application vulnerabilities that can be exploited through API endpoints in multiple

steps. Therefore, a technique that can generate sequences of RESTful API operations for vulnerability detection is urgently needed.

Recently, several techniques [234-236] have been proposed to automatically generate sequences of RESTful API operations for bug detection. These techniques take standard API specifications, such as the OpenAPI [237] specification (OAS) as input. In particular, they learn the dependencies among the API operations to build correct API operation sequences. Although these testing solutions can generate meaningful API operation sequences to be consumed by the SUT, they are not suitable for vulnerability identification in RESTful APIs due to three reasons. \bullet The API operation sequences generated by existing techniques are not dedicated for detecting vulnerabilities. For penetration testing, we should concentrate on testing potentially vulnerable operations. 2 The information extracted from the OAS documents is not enough to render diverse yet correct requests as test cases. Furthermore, OAS documents commonly contain syntax errors [236], which make the retrieved information less credible. 3 Existing testing techniques lack the appropriate payloads for API requests to simulate attacks as well as the test oracle to check if an attack is successful or not. They only observe responses with 5xx HTTP status codes to detect bugs and are not aware of injection or authorization-related vulnerabilities. Due to these challenges, there exists a huge research gap regarding the automated detection of multi-API vulnerabilities for RESTful services.

To overcome the above limitations, we propose NAUTILUS¹, which leverages a novel design of annotations in OAS to carry out penetration tests for RESTful services. The annotation can be classified into two categories: (1) operation annotations: these annotations guide NAUTILUS to generate meaningful and logical operation sequences by describing the relations between API endpoints; (2) parameter annotations: these annotations document the proper strategy to generate concrete parameter values for each request. The annotations are designed to be both automatically processable and human-readable. Therefore, NAUTILUS can work fully automatically or involve humans in the loop. Based on the annotations, NAUTILUS uncovers vulnerabilities in the SUT with two testing stages. The first stage is exploration. NAUTILUS can successfully request as many API operations as possible by building proper API operation sequences and rendering them with appropriate parameter values. Specifically, it focuses on API operations with user-controllable

 $^{^1\}mathrm{NaUTILUS}$ is the name of a submarine in the science fiction – Twenty Thousand Leagues Under the Seas.

parameters because they are more likely to contain injection vulnerabilities. By analyzing the responses from the service, NAUTILUS updates the annotations to fix the errors in the OAS and records the appropriate parameter value generation strategy. The second stage is *exploitation*. NAUTILUS constructs the operation sequence based on the exploitation pattern of the target vulnerability type, and mutates the injectable parameters with the payload dictionary. The SUT is then tested with the corresponding test oracles and reports the detected vulnerabilities.

We implemented NAUTILUS as a testing framework and evaluated it on six RESTful services. The experiment results show that NAUTILUS can outperform state-of-theart vulnerability scanners [229, 230] and RESTful API testing tools [234, 236] with superior API operation coverage (36.9% - 174.6% increment) and numbers of detected vulnerabilities (85.8% - 202.8% increment). We further applied NAUTILUS to nine real-world RESTful API services, and detected 23 vulnerabilities. Specifically, we found three vulnerabilities in Microsoft Azure [238] and one vulnerability Atlassian Confluence [239], which can affect millions of users. Until now, all of them have been confirmed and fixed by the vendors, and ten of them have been assigned with CVE numbers.

To summarize, we make the following contributions:

- We conduct an empirical study to comprehensively analyze the patterns of RESTful API vulnerabilities, and present the key findings.
- We propose a novel design of OpenAPI specification annotations, which can benefit both automated and human-in-the-loop testing.
- We implement an automated testing tool NAUTILUS, which can make use of the annotations to detect vulnerabilities in RESTful services.
- We compare the performance of NAUTILUS against four vulnerability scanners and RESTful API testing tools on six RESTful services and demonstrate that NAUTILUS can significantly outperform state-of-the-art techniques.
- We apply NAUTILUS to nine real-world web services, including famous commercial products, and identify 23 vulnerabilities with 12 assigned CVE IDs. We responsibly disclose the vulnerabilities to the vendors and all of the vulnerabilities are confirmed and fixed.

5.2 Background

5.2.1 Key Concepts

RESTful API. The REpresentational State Transfer (REST) is a software architectural style proposed in 2000 [240] that defines the behaviors of an Internet-scale distributed hypermedia system, such as the Web. A Web API following the REST standard is called a RESTful API. Similarly, a web service that follows the REST standard is called a RESTful service. The REST architecture constrains the behavior of the system, and one of the most basic constraints is the *Uniform Interface*, which regulates users to access resources through regulated CRUD operations. Modern RESTful APIs often use the HTTP protocol as the transportation layer, and naturally the CRUD operations of RESTful APIs are mapped to the HTTP methods POST, GET, PUT, and DELETE, respectively. A RESTful service can contain many endpoints, each of which is a digital location (typically with its own URL) to perform a series of pre-defined functions. These endpoints can be queried through different HTTP methods and body contents depending on the service's function, and each query is called an API operation.

OpenAPI Specification (OAS). OpenAPI (previously known as Swagger) defines a standard for describing RESTful APIs and the documentation that follows this standard is called OpenAPI specification [237]. The OpenAPI specification of target RESTful service contains the information of the object schemas as well as the API endpoints of a web service, including but not limited to the available CRUD operations, input parameters as well as expected responses. Each object has pre-defined fields and corresponding parameter types. Users can follow the specification to produce valid API operations and render them into HTTP requests to interact with the RESTful service endpoints.

Figure 5.1 shows a fragment of the OpenAPI specification for APIs in Buddy-Press service [241], an extension to WordPress [242] blog management system. In this example, three API endpoints are specified and they are marked with grey background. We can see that each API endpoint supports one or more CRUD operations. In total, four operations are described in Figure 5.1, showing their input parameters and responses. For an input parameter, it can be inside the

1/	1/members/me:		/groups:
2	get:	41	post:
3	responses:	42	requestBody:
4	'200':	43	required: true
5	description: "Success"	44	content:
6	put:	45	application/json:
7	parameters:	46	schema:
8	- in: header	47	type: object
9	name: x-wp-nonce	48	properties:
10	schema:	49	context:
11	type: string	50	type: string
12	required: true	51	groupname:
13	description: "WordPress nonce"	52	type: string
14	requestBody:	53	creater-id:
15	required: true	54	type: integer
16	content:	55	group_description:
17	application/json:	56	type: string
18	schema:	57	responses:
19	type: object	58	'200':
20	properties:	59	description: Success
21	context:	60	/groups/{groupname}/admin/manage-
22	type: string		members:
23	example: 'edit'	61	get:
24	name:	62	parameters:
25	type: string	63	- in: path
26	user_login:	64	name: groupname
27	type: string	65	schema:
28	email:	66	type: string
29	type: string	67	responses:
30	example:	68	'200':
31	'test@user.mail'	69	content:
32	password:	70	application/json:
33	type: string	71	schema:
34	roles:	72	type: object
35	type: string	73	properties:
36	example: 'user'	74	data:
37	responses:	75	type: string
38	'200':	76	x-wp-nonce:
39	description: "Success"	77	type: string

FIGURE 5.1: The OpenAPI specification of BuddyPress APIs^{*} * For clarity, we omit some details in the YAML file.

request body (body of put-/members/me), in the HTTP request header (parameters of put-/members/me), or in the URLs of endpoints (parameter groupname in get-/groups/{groupname}/admin/manage-members). For a response, it contains the HTTP status code as well as the content body. In addition, some operation parameters and responses may involve objects that are described by schemas. For example, the admin management operation response contains an application/json format data with data field and x-wp-nonce field.



FIGURE 5.2: RESTful API vulnerability categories

5.2.2 **RESTful API Vulnerabilities**

While there are many RESTful API vulnerabilities in the wild, their exploitation patterns and root causes were not systematically summarized. Before investigating the methodology for RESTful API penetration testing, we conducted an empirical study to answer two research questions to limit the types of vulnerability for detection and understand the challenge of RESTful API vulnerability detection:

RQ1 (Scope) What are the categories of RESTful API vulnerabilities?

RQ2 (Challenge) What are the differences between triggering the RESTful API vulnerabilities and triggering the bugs/internal server errors?

In the empirical study, we collected a total number of 609 RESTful API vulnerabilities from the CVE list [243] of the National Vulnerability Database (NVD) and exploit-db [244]. We manually analyzed the vulnerabilities via the disclosed information such as the CVE descriptions, the exploits, the patches and so on.

Vulnerability Categorization RESTful API vulnerabilities can be categorized by many criteria. Here we focus on using their Common Weakness Enumeration (CWE) [245] types for categorization. Figure 5.2 shows the categorization result. From Figure 5.2, we can observe that there are two main root causes for RESTful vulnerabilities. ● 52.3% of vulnerabilities are caused by improper user input handling, which can be mapped to multiple CWE items including different types of injections (SQL injection, XSS, command injection, etc.). @ 47.7% of vulnerabilities are caused by improper resource management, including broken access control, lack of rate limits, sensitive information disclosure, etc. For this particular research, we focus on detecting the vulnerabilities caused by improper user input handling due to their prevalence and significance. Specifically, they are the major types of vulnerabilities (52.3%), and lead to exploitable scenarios, including command injection and code execution. In comparison, improper resource handling vulnerabilities are difficult to model uniformly because it is difficult to define *sensitive* data in different contexts. Therefore, we restrict our research scope to the former type of vulnerabilities and refer to them as *RESTful API vulnerabilities* unless stated otherwise and we discuss the identification of improper resource handling vulnerabilities.

Vulnerability vs. Bug Through the empirical study, we found that the detection of vulnerabilities differs from the detection of bugs in three aspects. **1** Attack *Payload.* Each type of RESTful API vulnerability requires a corresponding type of payload for triggering. The exploit payloads are mainly injected into three positions of a RESTful request: body parameters, in-url parameters and cookies. For example, exploiting a SQL injection vulnerability (CVE-2019-10692) in the RESTful service requires a suffix of -- - to the original SQL query in the body of the request, which is a common payload pattern for SQL injection. 2 API Call Sequence. For detecting a RESTful API bug, the only requirement for an API call sequence is that it can reach the buggy API properly. On the contrary, to detect RESTful API vulnerabilities, different types of vulnerabilities require different API request sequence patterns. According to our empirical study, there are two major types of patterns for API call sequences. For SQL injections, normally they only require one GET operation for both injecting and triggering. For other incorrect user input handling vulnerabilities, such as stored XSS vulnerabilities, they require one POST/PUT operation for injecting attack payloads followed by one GET operation to trigger. **3** Test Oracle. RESTful API bugs and vulnerabilities requires different types of test oracles for capturing. For detecting RESTful API bugs, we only need to observe the status codes of the responses. A bug occurs when a response has a 5xx status code. For detecting RESTful API vulnerabilities, we need to use three types of manifestations: the change of status code before and after applying the attack payloads, the change of response data object structure before and after

applying the attack payloads, and the semantic relation between the content of response bodies and the attack payloads.

5.3 Running Example

In BuddyPress version 7.2 and below, there is an injection-based privilege escalation vulnerability, which allows an attacker to escalate his/her user privilege to the administrator. This vulnerability has been recorded as CVE-2021-21389 [246]. Figure 5.1 shows the OpenAPI specifications for some of the APIs related to this vulnerability and Figure 5.3 shows the exploitation steps. We first introduce the mechanism of CVE-2021-21389 and then explain why existing bug detection and penetration testing techniques cannot reveal it.

According to Figure 5.3, CVE-2021-21389 requires three steps (including six API calls) to exploit. • The attacker needs to signup and login properly. The login API will return a nonce, which is needed as an identity token to access the follow-up APIs. **2** In order to launch the attack, the attacker first needs to send a POST request to the /groups API to create a new group. The attacker can then get groupname in the response which contains the data object of the newly created group. With groupname, the attacker can send a GET request to the /groups/{groupname}/admin/manage-members API to get the data objects of the administrators in the group, including a x-wp-nonce, which can be used as the identity token for group administrators. By adding the x-wp-nonce into the request headers, the attacker can send PUT requests to /members/me to change his/her personal information as an administrator. By appending an attack payload to the request that sets the role property to administrator, the attacker can escalate his/her privilege to the administrator. ⁽³⁾ To verify successful privilege escalation, the attacker sends a GET request to /members/me and checks whether the data object in the response contains more properties than documented in the OAS.

This vulnerability cannot be detected by existing penetration testing tools such as w3af and ZAP. The reason is that they can only test one API of the SUT per time while the example vulnerability requires a sequence of six API calls to trigger

Outpu





§4.4

7 6

Payload Injecto

FIGURE 5.4: Overview of NAUTILUS

nerability Verifie

and verify. Simply injecting the attack payloads via PUT operations through the /members/me API does not work due to the wrong value of x-wp-nonce.

This vulnerability cannot be detected by existing bug detection techniques such as RESTLER, RESTTESTGEN, and MOREST. The reason is double-fold. First, these techniques cannot add attack payloads to their requests. Second, even if attack payloads are added, these techniques lack the awareness of whether an attack is successful or not. They only capture responses with 5xx status code for bug detection while in this example, the vulnerability does not trigger any response with 5xx status code.

5.4 Design

5.4.1 Overview

Figure 5.4 shows the overview of NAUTILUS. We can see that the overall input is the OpenAPI specification of the SUT and the overall outputs are the updated

OpenAPI specification with customized annotations and the detected vulnerabilities. The workflow of NAUTILUS is as follows.

Annotation Processing. • Given the original OpenAPI specification of the SUT, the human expert can *optionally* add some initial annotations to the specification following the specification annotation design of NAUTILUS (Section 5.4.2). The annotations are automatically processable and human-readable. Therefore, after NAUTILUS has generated some new annotations, the human expert can also choose to further update them manually. • With the annotated specification, NAUTILUS leverages its specification parser to extract API information, including the relations among the APIs and the parameter details of each API.

Two-stage Testing. ³ With the extracted API information, NAUTILUS generates API operation sequences to test the SUT. NAUTILUS runs testing in two stages, namely *exploration* and *exploitation*. NAUTILUS switches from the exploration stage to the exploitation stage when no endpoints are successfully requested after a predefined time threshold t^2 . Meanwhile NAUTILUS switches from the exploitation stage back to the exploration stage after the same amount of time t. **4** Under the exploration mode, NAUTILUS aims to successfully request as many POST/PUT API endpoints as possible by generating proper API call sequences as test cases. The test case generation involves generating a correct sequence of API calls and filling up the API parameters properly (Section 5.4.3). Θ In the exploration process, NAUTILUS leverages the execution feedback from the SUT to create new annotations or update existing ones, thus providing more accurate guidance for the testing. The updated annotations can be used for extracting new API information. 6 During the exploitation stage, NAUTILUS applies the attack payloads to the successful API call sequences generated in the exploration stage to create new test cases for vulnerability detection (Section 5.4.4). O In the exploitation stage, NAUTILUS captures the vulnerability by verifying the execution feedback of the SUT. Different types of attack payload require different verification oracles.

Result Handling. If the NAUTILUS completes the testing, it can provide the updated/annotated OpenAPI specification together with the detected vulnerabilities to the human expert for further analyses, such as vulnerability clustering or specification fixing.

²The default value of t is 30 minutes.



FIGURE 5.5: Annotation updates on the running example* * For simplicity, we omit unnecessary fields in the specification.

5.4.2 Specification Annotation

NAUTILUS uses a set of customized annotations to complement the information embedded in the OpenAPI specification. The design of the annotation is fully compatible with the OpenAPI 3.0 standard. Moreover, the annotations are humanreadable and automatically processable. Hence, both human experts and NAU-TILUS can create or update the annotations.

The purpose of the specification annotations is to embed more information in the OpenAPI specification. The reason is that although the OpenAPI specification can document the endpoints and parameters information, the information is not complete or accurate enough for NAUTILUS to generate valid test cases. To address this problem, we introduce two types of annotations: *operation annotation* and *parameter annotation*. The former helps NAUTILUS to construct meaningful API operation sequences, while the latter improves the effectiveness of parameter value generation.

5.4.2.1 Operation Annotation

Operation annotations are designed to guide the generation of valid API operation sequences that comply with the business logic of the service. For this purpose, we design the following three types of fields for the operation annotations:

Dep-operation. The *dep-operation* field annotates an operation with its dependent operations that should be executed in advance. This field is in the form of

a *list* of API operations, where each operation is a uniquely identified string in the format {request method}-{endpoint name}. The dependencies among operations can be classified into three categories: parameter-wise data dependencies, CRUD dependencies and logical dependencies. **①** The parameter-wise data dependency refers to the case where the variables in the response of one operation is used as the request parameters of the other operation. For example, in Figure 5.5 b), get-/groups is marked as a dependent operation for post-/groups because the response of the former matches the request body of the latter by both referring to the group schema. ² The CRUD dependency is to enforce the CRUD restrictions and link up operations according to their CRUD relations. For example, in Figure 5.5 c), post-/groups is marked a dependent operation for get-/groups since following the CRUD relation, a group should be created first before it can be read. Notice that using parameter-wise data dependencies and CRUD dependencies may yield contradictory operation dependencies, like shown by the running example. Therefore, we have a dynamic update mechanism to resolve the conflicts (Section 5.4.2.3). ⁽³⁾ The logical dependency refers to the dependency introduced by the internal logic of the SUT. For example, in both Figure 5.5 b) and c), post-/login is the dep-operation for both get-/groups and post-/groups. The reason is that post-/login can return a nonce, which is a required parameter in the header used as the identity token for accessing other API endpoints.

Term-operation. The *term-operation* field annotates the operations that terminate the current session with the SUT. Similar to dep-operation, this field is a *list* of API operations and the format of the API operations is the same. The operations stored in this field should only be executed after other operations. Endpoints like *logout* and *change_password* belong to this category. During testing, NAUTILUS never inserts the term-operations in the middle of operation sequences.

Alias. The *alias* field annotates the aliases of parameter names across the operations. This field is a *list* of strings. For each of the strings, its format is described as {operation}.{parameter name} and is pointed to the parameter from the specific operation. This design aims to address the naming issue in OAS. In practice, we find that poorly maintained OAS have one parameter with different names in different operations. Since the aliases are across multiple operations, we put the alias field under operation annotations. This field is useful for rendering API requests as it helps to correctly match the parameter values across different operations in the same sequence (Section 5.4.3).

5.4.2.2 Parameter Annotation

Parameter annotations are designed to guide NAUTILUS to generate and link up parameter values of the API operations by addressing the drawbacks of existing solutions. Existing RESTful API testing techniques [234–236] generate parameter values via two basic strategies: random and previous success. The random strategy generates random values based on the type of data specified in the API specification (e.g., a random integer if the value type is integer). The previous success strategy generates the value of a parameter using the value of the last successful request. Both strategies have clear drawbacks. On the one hand, the random strategy is inefficient because requests with non-compliant parameter values will be rejected by the RESTful service without clear feedback, which does not provide sufficient guidance to the next parameter value generation. On the other hand, the previous success strategy does not bring enough diversity to the test cases, thus cannot explore the API service effectively. To address both randomness and correctness of the parameter values, we design the following four fields to describe the parameter generation strategies:

Example. This field is a *boolean* value. If the value is True, NAUTILUS will use the parameter values documented in the *example* field of the OAS. Normally, the parameter values provided by the examples are correct, which can help NAUTILUS to successfully request the corresponding endpoints.

Dynamic. This field is a **boolean** value. If the value is **True**, NAUTILUS will get the corresponding parameter values from previous successfully requested operations in the same sequence. The mechanism of deciding the parameter is from which previous operations is almost the same as **RESTLER** [234], where parameter names and schemas are used to determine whether two parameters should be linked up. The only difference is that NAUTILUS also uses the alias information of the parameters.

Success. This field is a *boolean* field. If it is **True**, NAUTILUS will use the parameter values of the last successful request. Otherwise, NAUTILUS will try to generate new parameter values randomly.

Example	Dynamic	Success	Generation Strategy
			Dynamic
\bullet	\bullet	0	Dynamic
\bullet	0	\bullet	Success + Mutation
\bullet	0	0	Example + Mutation
0	\bullet	\bullet	Dynamic
0	\bullet	0	Dynamic
0	0	\bullet	Success + Mutation
0	0	0	Random Generation

TABLE 5.1: Parameter generation strategy (\bigcirc : True \bigcirc : False)

Mutation. This field is a *float number* with a range of 0.0 to 1.0, which represents the parameter's mutation degree. The higher the value is, the service can intake more flexibly mutated parameter value. Specifically, we adopt the normalized edit distance or similarity (NES) from [247] to measure the mutation degrees. Given the original parameter string s with length l, we generates the mutation string s_m with length l_m while maintaining the NES between the two strings below boundary t. The NES can be calculated by $e^{\frac{d}{d-max(l,l_m)}}$, where d is the Levenshtein Distance [248] between the two strings. We also record the largest NES of the mutated parameter that is still properly handled by the target RESTful service.

The final parameter generation strategy is an interplay of the values of four fields. Table 5.1 shows the relation between the generation strategy and the field values. NAUTILUS generates the parameter value based on its annotation at runtime. In general, the dynamic field has the highest priority and the example field has the lowest priority. Note that some field value combinations in Table 5.1 may never appear in practice. For instance, the example field and the dynamic field should never both take the value of **True** for well-documented OASs.

5.4.2.3 Annotation Updates

Annotation Initialization and Manual Update. At the beginning of testing, NAUTILUS provides an utility to generate the initial annotations based on the parameter-wise dependencies and heuristics, such as using keyword matching to identify operations as *login* or *logout*. Figure 5.5 b) shows an example of the initially added annotations of NAUTILUS. The human expert can then choose to update the annotated OAS documentation based on his/her domain knowledge. As far as the manually added annotations follow the required formats, NAUTILUS can work with them seamlessly. Note that the human expert can also skip this manual update step to let NAUTILUS work fully automatically.

Dynamic Annotation Update. During testing, NAUTILUS updates the annotations dynamically according to the execution feedback of the SUT. For operation annotations, NAUTILUS updates them in the following three scenarios. **1** One of the most common cases is where the OAS does not correctly document the response body of an operation. In this case, NAUTILUS needs to analyse the actual response body after successfully requested an endpoint to fine-tune parameter-wise dependencies and update the corresponding dep-operation annotations. In the running example (Figure 5.1), the response body of get-/members/me is not documented. NAUTILUS will update the parameter-wise dependencies related to the response body of this operation once it receives the actual successful response during the execution. ② Another common case is updating the parameter aliases. In the running example, the get-/members/me operation returns current user object if successfully requested. The object has an id property, which is the ID of the current user. The post-/groups operation requires a parameter called creator-id. In BuddyPress, users can only use its own id to create new groups. Therefore, although creator-id and id have different names, they refer to the same value across the two operations. After some trail and errors during the testing, NAU-TILUS can recognize that only when the values of creator-id and id are equal, the post-/groups can be requested successfully. Hence, NAUTILUS will mark them as aliases. ³ The last case is about removing the infeasible dep-operations. If the operation after a dep-operation cannot be executed successfully after Θ tries (the default value of Θ is 10), the dep-operation will be removed from the annotation. For example, in Figure 5.5 b) and c), get-/groups will be removed from the depoperation list of post-/groups during testing. The reason is that BuddyPress does not allow duplicate group names. So getting the information of an existing group and using the same information to create a new group will fail. For *parameter annotations*, the updates are mainly about adjusting the value of the success field and the mutation field. If an operation has been successfully requested, its success field will be updated to True. As for the mutation field, the key idea is to increase the value of the mutation field upon successful requests and to decrease

the value upon failed ones. The rationale is that if the request is successful, NAU-TILUS can loose the restrictions to try out more aggressive mutations to increase the diversity of the parameters. However, if the request fails, NAUTILUS needs to apply mutations with smaller granularity to guarantee the correctness of the parameters.

5.4.2.4 Annotation Primitives

Our annotation primitives are constructed based on the specification extension feature of OpenAPI [249]. OpenAPI also allows users to add additional fields to parameters, namely *x*-parameter. With this feature, OpenAPI can be extended to support representations beyond RESTful services. For instance, Microsoft Azure APIs [238] contain custom fields including x-ms-paths and x-www-form-urlencoded to interact with internal services through encoded web form, which are not supported by conventional RESTful services. In NAUTILUS, the annotation primitives are denoted as x-operation-annotation and x-parameter-annotation. Since the annotation primitives are designed based on the official feature of OpenAPI, they are fully compatible with any OAS document.

5.4.3 Exploration Stage

In the exploration stage, NAUTILUS aims to successfully request as many API operations as possible with properly built API operation sequences. Algorithm 2 describes how NAUTILUS builds the API operation sequence for an API operation. As shown in Algorithm 2, NAUTILUS first checks if the API operation is *interesting* or not. Unlike existing RESTful testing techniques such as RESTLER, RESTTESTGEN and MOREST, NAUTILUS generates operation sequences by enumerating through only the interesting API operations instead of all operations. The interesting API operations refer to the operations that can possibly get attacked by requests with well-crafted payloads. Therefore, API operations which can accept user inputs and add/update backend data of the SUT are considered interesting. Specifically, NAUTILUS focuses on API operations with POST/PUT HTTP methods or API operations with the GET HTTP method and have in-URL parameters. For example, the get-/members/me operation in the running example (Figure 5.1) is not interesting because it does not accept user input and cannot be attacked. Thus, NAUTILUS will not try to build API call sequences for this operation. If an API operation is considered interesting, NAUTILUS will check the dep-operations of this operation and build the API call sequence by recursively including all the required API calls (line 7 – line 17 in Algorithm 2).

After the successful generation of API operation sequences, NAUTILUS renders the concrete HTTP requests one API operation after another. For each request, NAUTILUS generates its parameters according to the parameter annotations (Section 5.4.2.2) following the strategy in Table 5.1. The requests cannot be generated all at once because the parameters of some API calls are from the responses of previous API calls.

Finally, NAUTILUS will send the requests to the SUT to verify whether the target operation can be requested successfully and update the annotations according to the execution feedback (Section 5.4.2.3). Note that the target operation is considered successfully requested as long as it returns a response with 2xx status code.

The key difference of the exploration strategy between NAUTILUS and existing solutions [234, 236] is that NAUTILUS is aware of the most appropriate parameter values that can lead to successful requests. In particular, NAUTILUS identifies the best parameter generation strategy based on the execution feedback and records it in the parameter annotation, while prior solutions mainly rely on pre-defined strategies. After the exploration stage, the annotations are updated, further guiding the efficient vulnerability identification in the next stage.

5.4.4 Exploitation Stage

Once NAUTILUS cannot successfully request new endpoints for certain time, it switches into the exploitation stage. Or, if all endpoints are successfully requested, NAUTILUS will stay in the exploitation stage. In the exploitation stage, NAUTILUS aims to detect as many vulnerabilities as possible. The workflow of the exploitation stage is as follows. • NAUTILUS collects the API operation sequences which can successfully request the interesting API operations. These API operation sequences will be used as the basis for vulnerability detection. • When rendering the requests for an API operation sequence, for the dependent API operations to of the interesting API, NAUTILUS leverages the success parameter annotations to

-	Algorithm 2: API operation sequence generation
]	Input: a: A RESTful API operation
(Dutput: S : the API operation sequence based on a
1 (def sequence_generation(a):
2	$ \mathcal{S} \leftarrow [];$
3	if $is_interesting(a)$ then
4	$concat_sequence(\mathcal{S}, a);$
5	end
6	return S ;
7 (def $concat_sequence(S, a)$:
8	if $a.dep_operations = \emptyset$ then
9	return;
10	end
11	for $d \in a.dep_operations$ do
12	$\qquad \qquad \mathbf{if} \ d \in \mathcal{S} \ \mathbf{then} \\$
13	$\mathcal{S}.remove(d);$
14	end
15	$\mathcal{S} \leftarrow [d, \mathcal{S}];$
16	$concat_sequence(\mathcal{S},d);$
17	end
18 (def $is_interesting(a)$:
19	return $a.http_method \in \{POST, PUT\} \lor (a.http_method =$
	$GET \land a.in_url_params \neq \emptyset);$

reuse the parameter values in the last successful requests. In contrast, for the interesting operations, NAUTILUS will use predefined attack payloads to replace its parameters. As elaborated in Section 5.2, RESTful services can contain various categories of vulnerabilities, each of which can only be revealed with certain type of payloads. Thus, we propose a vulnerability-specific mutation strategy to uncover certain types of vulnerability in the target RESTful service. The mapping between the most common types of the interesting API and the types of attack payloads are summarized in Figure 5.6. Below we elaborate the key technical steps involved.

5.4.4.1 Payload-based Mutation

Our general strategy is to mutate the normal request parameters with the vulnerabilityspecific payloads. Instead of arbitrarily injecting payloads into the parameters, NAUTILUS focuses on operations that have a greater possibility of containing vulnerabilities and construct the sequence accordingly in the following steps. • NAU-TILUS identifies the user-controllable parameters that are possible to get injected. The user-controllable parameters are the parameters whose dynamic fields have



FIGURE 5.6: The mapping between the request types and the required payload-s/oracles

the value of False, which means that the value is not inherited from previous responses. NAUTILUS then identifies the operations that contain the injectable parameters as the candidate operations to test. It selects one operation from the candidate operations and picks one parameter from it as the mutation target. If the target operation is GET, NAUTILUS will reuse the corresponding API operation sequence generated in the exploration stage. When rendering requests, the injectable parameters, typically in-url parameters, are mutated using the payload dictionary. Specifically, the target parameter is replaced by the payload value to formulate the final request. The other parameter values are generated on the basis of the annotated strategy normally. If the target operation is POST/PUT, NAU-TILUS will randomly pick a GET operation which has CRUD relation with the target operation and append it to the API operation sequence. The rationale of adding the GET operation is to obtain more information from SUT to serve as test oracles.

5.4.4.2 Payload and Oracle

The types of test oracles used for detecting vulnerabilities are related to the types of payloads and the relations are illustrated in Figure 5.6.

Status Code. Some vulnerabilities can cause changes in the status codes returned by the SUT. For instance, a successful login bypass results in 200 status code, while normally the server shall return 400 if the wrong credentials are provided.

Data Structure. Some vulnerabilities can cause the operation to return data objects with different data structures from what is described in the OAS. For example, SQL injection changes the response data structure because unexpected table contents are returned after successful exploitation.

Semantic Relation. Some vulnerabilities may falsely execute the payload content and add semantic relations between the contents of the payloads and responses. For instance, some command injection vulnerabilities can cause the parameters to be executed instead of being parsed as strings, and the execution result is predictable (e.g., the payload is '1+1' and the response is '2').

5.5 Implementation and Evaluation

We implement NAUTILUS based on Python 3.9.0 with 6,500 lines of code and conduct experiments to evaluate the performance of NAUTILUS. Our evaluation targets the following questions:

RQ1 (Vulnerability Detection) How is the vulnerability identification capability of NAUTILUS?

RQ2 (Coverage) How is the operation exploration capability of NAUTILUS?

RQ3 (Ablation Study) How do the annotation strategies affect the performance of NAUTILUS separately?

RQ4 (Real-world Targets) Can NAUTILUS identify vulnerabilities in real-world applications, including those industrial products?

5.5.1 Evaluation Setup

Evaluation Baselines. We compare our solution with both open-source vulnerability scanners and existing research works on RESTful API testing. It is worth noting that these tools are designed for different purposes. Vulnerability scanners are designed to assess web applications and APIs. Given requests to API endpoints, they send mutated requests to those endpoints and report the potential vulnerabilities directly. Conversely, RESTful API test tools aim to achieve better coverage and bug reporting, and they do not have vulnerability reporting capability. To conduct a fair comparison and evaluation, we select vulnerability scanners and RESTful API testing tools that are extensible to have custom payloads so that we can use the same payload data on all tools. In the end, we select the following four tools.

(1) Zed Attack Proxy (ZAP) [229] is an open-source blackbox web vulnerability fuzzer developed by OWASP. It is mainly used for blackbox vulnerability assessment and penetration testing. In this evaluation, we use ZAP's OpenAPI add-on and disable unrelated web exploration functions such as web crawling modules.

(2) **w3af** [230] is an open-source web application attack and audit framework. Similar to the previous setting, we use the w3af in-built module *crawl.open_api* and disable unrelated functions.

(3) **Restler** [234] is an open-source blackbox RESTful API testing technique developed by Microsoft. It dynamically builds operation sequences by appending new API operations according to execution feedback.

(4) **Morest** [236] is a state-of-art blackbox RESTful API testing technique which constructs operation sequences through the dynamically updated RESTful-service Property Graph (RPG).

We modify the above tools to use the same FUZZDB payload dictionary [250]. It covers various types of vulnerabilities and is adopted by various testing tools and industrial solutions [229, 251, 252]. For ZAP and w3af, we update the payload dictionary to their corresponding modules. For RESTLER and MOREST, we extend their mutation modules so their value generation strategy use the payload from dictionary instead of random value generation. Note that we do not make changes on their test sequence generation strategy.

Evaluation Benchmarks. We select real-world web applications as our evaluation benchmarks using three criteria: (1) open-source for exploitation reproducibility, (2) actively maintained to validate security findings, (3) complete OAS or RESTful documentation available as required input for all baselines. The result is a selection of six web applications, as presented in Table 5.2. Three of the benchmark applications (OWASP NodeGoat [253], OWASP JuiceShop [23] and

Subjects	LoC	Endpoints	Description	Version	Developer
NodeGoat [253]	24933	20	Educational	1.4	OWASP
Juice Shop [23]	109244	50	Educational	12.5	OWASP
VAmPI $[254]$	2695	13	Educational	-	Community
SeoPanel [256]	62277	20	SEO Management	4.0	Independent
Navigate $[257]$	57192 ³	12	CMS	1.8	Independent
Gila $[258]$	49391	24	CMS	2.1.0	Independent

TABLE 5.2: Benchmark Applications

VAmPI [254]) are deliberately vulnerable applications with seeded vulnerabilities for education purposes. The other three applications are open-source software with both web interface and well-documented RESTful API endpoints. All of these applications are implemented based on their default documentation, and their details are demonstrated in the following Table 5.2.

Benchmark OpenAPI Specifications. We use the same OASs for all the evaluation baseline solutions to test the evaluation benchmarks. For NAUTILUS, we do not include additional manual annotations, and only use an automatic script to generate the initial operation annotations through keyword mapping on operation names (login, logout, checkout, etc.). The automation script is open-sourced on our project website [255].

Evaluation Criteria. We use two criteria for the evaluation of NAUTILUS and the baselines to answer the aforementioned research questions.

(1) Vulnerabilities: The number of vulnerability is crucial criteria for security testing. Without loss of generality, we focus SQL injection, XSS and improper access control, while our solution can also identify other injection-based vulner-abilities with proper payloads. In Section 5.5.5, we present industrial examples to demonstrate other types of vulnerabilities identified by our tools.

(2) **Operation Coverage**: Operation coverage directly reflects the successful exploration of RESTful API services. In the experiments, we use the successfully requested operations (SROs) as a criterion, because it reflects whether a technique can generate valid and complex requests to cover the deeper code logic in RESTful services.

Evaluation Settings. We setup all the tools and benchmarks based on their default installation settings. For the benchmark RESTful services, we host them



FIGURE 5.7: The vulnerabilities and their types uncovered by different tools on the evaluation benchmarks.

TABLE 5.3: Selected RESTful API vulnerabilities identified by NAUTILUS. For Vulnerability Identification Tools: \checkmark : this tool can identify the vulnerability, \checkmark : this tool cannot identify the vulnerability, \neg : this tool is not designed to uncover the type of vulnerability. For Manual Annotation: \checkmark : the annotations are updated by human experts to provide guidance on operation and parameter generation, \varkappa : no manual inputs into the specification annotations.

Target Application	Version	Vender Confirmation	CVF/Issue ID	Vulnerability Type	Mult: ADI	Vulne	rability Id	entificatio	n Tool	s	Manual Annotations
Target Application	VEISIOII	vendor commitation	CVE/Issue-ID	vullerability Type	WITH AT 1	NAUTILUS	Restler	Morest	ZAP	w3af	Manual Annotations
	1.8	1	CVE-2021-37377	SQL Injection	×	1	×	X	1	1	×
	1.8	1	CVE-2021-37476	SQL Injection	×	1	1	1	1	1	×
DDCMC	1.8	1	CVE-2021-37475	SQL Injection	×	1	1	1	1	1	×
III CM5	1.8	1	CVE-2021-37474	SQL Injection	1	1	×	×	X	X	1
	1.8	1	CVE-2021-37473	SQL Injection	1	1	×	×	X	×	1
	1.8	1	CVE-2021-37394	Privilege Escalation	1	1	×	×	×	×	1
	-	1	CVE-2022-33659	Privilege Escalation	1	1	×	×	X	X	1
Azure	-	1	CVE-2022-30181	Privilege Escalation	1	1	×	×	X	×	1
	-	1	CVE-2022-33657	Privilege Escalation	1	1	×	×	X	×	1
Confluence	7.13.0	1	Internally Issued	OGNL Injection	1	1	-	-	-	-	1
Nordanta	2.9.4	1	Issue r1561-#26	SQL Injection	×	1	1	1	1	1	×
Ivavigate	2.9.5	1	Issue r1561-#27	Privilege Escalation	1	1	×	X	×	×	×
Rukovoditel	2.8.3	1	CVE-2021-30224	CSRF	×	1	-	-	1	1	×
SeoPanel	4.0	1	Issue #219	SQL Injection	~	~	×	×	X	X	×
CileCMS	2.1.0	1	CVE-2021-34113	Directory Traversal	1	1	X	X	x	×	×
GuaCMS	2.1.1	1	CVE-2021-34115	Stored XSS	1	1	-	-	×	X	×

on a local machine and run each technique with 12 hours. After each round, we tear down the benchmark and restore the environment (e.g., docker containers, self-hosted virtual machines) to ensure the consistency of RESTful services between tests. In addition, we repeat all experiments for 5 times to mitigate randomness and adopt Mann-Whitney U test (with the confidence threshold $\alpha = 0.05$) and \hat{A}_{12} [259] calculation for statistic tests. So in total, our experiment records of 1,800, i.e., 6 projects * 5 settings * 12 hours * 5 repetitions, CPU hours of testing. We summarize our findings as follows.

Result Collection. After NAUTILUS and the evaluation baseline solutions report the vulnerabilities, we collect the result and manually conduct the exploitation to confirm the vulnerabilities. The false positives are eliminated, and we discuss their causes in Section 5.6.

5.5.2 Vulnerability Detection (RQ1)

The number of unique vulnerabilities identified by different solutions are presented in Figure 5.7. It can be noticed that all the baseline solutions have similar performances in vulnerability testing. This is because they use the same payload dictionary and follow a similar testing strategy to identify vulnerabilities at each endpoint individually. Particularly, ZAP and w3af have better performance compared to the two RESTful API testing solutions, because they have in-built XSS execution detection modules to identify such vulnerability.

In contrast, NAUTILUS achieves significantly better performance in both vulnerability types and number of vulnerabilities. NAUTILUS is able to uncover different types of vulnerability, including SQL injection, command injection, XSS and privilege management. On average, it identifies more vulnerabilities (69.8%) compared to the other solutions on the three educational benchmarks. Particularly, NAU-TILUS can cover all vulnerabilities identified by these solutions. Meanwhile, NAU-TILUS identifies 10 0-day vulnerabilities on the three real-world applications while the baseline solutions only identify 3 in total. These 10 vulnerabilities identified from SeoPanel, Navigate CMS and Gila CMS have been confirmed by the vendors. Their details are included in our website [255].

In-depth Analysis. To further explore the accuracy and effectiveness of NAU-TILUS, we perform an in-depth analysis about false positives and false negatives. In particular, a false positive is the case where NAUTILUS reports a non-exploitable vulnerability. As it is hard to define true negatives in the domain of vulnerability detection, we define the false positive rate as FP/(FP + TP). The detailed experimental results are presented in Table 5.4. In summary, NAUTILUS achieves a false positive rate of 24.74% on benchmark services, which is comparable to other solutions. This is acceptable as a human expert can easily follow the test results to identify valid vulnerabilities. We highlight that false positives are largely dependent on the quality of the payload dictionary and the target service because they are mainly contributed by (1) the non-compliant RESTful endpoint implementation, where the response status code or body content does not fulfill the RESTful standards and triggers the test oracle, and (2) unexpected service behaviors, where the payload triggers a buggy implementation in the service, causing service crashes yet not exploitable. These false positives can be mitigated by modifying the OpenAPI documentation to describe the actual behavior of the service.

We further study the false negatives of NAUTILUS. Due to the inherent difficulties in identifying all vulnerabilities in real-world services, we extend the experiment to explore if NAUTILUS can uncover known vulnerabilities. To do this, we collect 50 reproducible CVEs from 12 different open-source applications containing 25 different components/plugins. The known vulnerabilities cover all subtypes of injection vulnerabilities illustrated in Section 5.2. We study the false negatives of these solutions by verifying the number of vulnerabilities that can be reported by each solution. The experimental result shows that NAUTILUS detects 70.0% (35/50) of the known vulnerabilities in the benchmark application, surpassing other solutions that achieve an average detection rate of 39.5%. Such improved performance can be attributed to Nautilus's superior endpoint coverage capability, which allows the detection of vulnerabilities at endpoints not covered by traditional solutions. NAUTILUS fails to uncover some vulnerabilities due to two main reasons: (1)some vulnerability can only be triggered by case-specific payloads, which are not included in the payload dictionary; (2) the service does not fulfill the RESTful standards and the oracle cannot determine if the vulnerability is triggered. We present the detailed analysis of each selected CVE on our website [255].

	NAUTILUS	Restler	Morest	w3af	ZAP
Juiceshop	8/7	3/3	4/3	4/2	4/3
Vampi	6/1	2/0	3/0	3/0	2/0
NodeGoat	4/0	3/0	4/0	4/0	3/0
SeoPanel	2/0	0/0	0/0	0/0	0/0
Navigate	5/3	2/1	3/1	3/1	2/1
Gila	2/2	0/2	0/2	1/2	1/2
FP Rate	24.74%	30.56%	27.98%	20.83%	23.81%

TABLE 5.4: False positives / true positives identified by different tools on the evaluation benchmarks.

5.5.3 Coverage (RQ2)

We present the operation coverage results of different tools in Table 5.5. NAUTILUS achieves competitive performance in successfully requested operations compared to

baseline solutions. Specifically, our solution achieves 163.1% more endpoint coverage on the benchmarks compared to the traditional web vulnerability identification tools. This is because traditional solutions can barely generate valid POST or PUT requests to interact with the API, thus not efficient in endpoint discovery. Compared to the RESTful API testing tools RESTLER and MOREST, our endpoint coverage increased by 54.8% and 36.9% on average. While the RESTful API testing solutions achieve better performance, they cannot generate specific test sequences to cover the corner cases. For instance, they fail to understand the login-logout logic in all the benchmark applications, thus cannot perform testing with different user accesses. Also, some endpoints have extreme restrictions on input value formats, and it is difficult to generate them by random. Our annotation strategy assists the solution to overcome these drawbacks.

TABLE 5.5: Performance of NAUTILUS against RESTLER, ZAP and w3af in terms of both the average endpoint coverage and detected vulnerabilities (DV). We run this experiment 5 times (24 hours each time) and highlight statistically significant results in bold (We calculate the average increased number by $\frac{(\# of \text{ NAUTILUS}) - (\# of \text{ baseline})}{\# of \text{ baseline}}$.).

	Average Endpoint Coverage (EC)								Average # of Detected Vulnerabilities (DV)									
Subjects	NAUTILUS	Rest	ΓLER	Mor	EST	ZA	Р	w3a	af	NAUTILUS	Rest	LER	Mor	EST	ZA	P	w3a	af
	μEC	μEC	\hat{A}_{12}	μDV	μDV	\hat{A}_{12}												
NodeGoat	15.00	10.40	1.00	11.80	1.00	6.00	1.00	7.20	1.00	4.00	2.60	1.00	4.00	0.50	4.00	0.50	3.00	1.00
Juice-shop	21.20	14.00	1.00	16.20	1.00	8.00	1.00	7.00	1.00	7.00	3.00	1.00	4.00	1.00	4.00	1.00	4.00	1.00
Vampi	11.00	7.60	1.00	7.60	1.00	4.00	1.00	4.00	1.00	5.00	2.00	1.00	3.00	1.00	3.00	1.00	2.00	1.00
SeoPanel	12.80	9.00	1.00	10.20	1.00	6.00	1.00	3.80	1.00	2.00	0.00	1.00	0.00	1.00	0.00	1.00	0.00	1.00
Navigate	12.00	8.00	1.00	9.40	1.00	4.00	1.00	4.00	1.00	6.00	1.00	1.00	1.00	1.00	2.00	1.00	1.00	1.00
Gila	18.60	9.80	1.00	11.00	1.00	8.00	1.00	7.00	1.00	2.00	0.00	1.00	0.00	1.00	1.00	1.00	0.00	1.00
Average Increased (%)	0.00	54.08	-	36.90	-	151.67	-	174.55	-	0.00	202.80	-	116.50	-	85.84	-	159.28	-

5.5.4 Ablation Study (RQ3)

To specifically demonstrate the value added by human annotations in NAUTILUS, we conducted an ablation study focused on the two principal components of our annotation strategy: operation annotation and parameter annotation. This study aims to delineate how each type of annotation contributes to the enhancement of NAUTILUS's effectiveness in guided testing and parameter generation. We constructed three variants of NAUTILUS for this evaluation: (1) NAUTILUS-NO-ANNOTATION, where all annotations are disabled, (2) NAUTILUS-OPERATION-ONLY, which excludes parameter annotations, and (3) NAUTILUS-PARAMETER-ONLY, omitting operation annotations. The performance of these variants was assessed over five runs, each lasting 8 hours, to mitigate statistical biases. The outcomes of this ablation study, depicted in Figure 5.8, underscore the significant impact of annotations. NAUTILUS consistently outperforms the other three variants in both vulnerability identification and endpoint coverage. The findings are as follows: Without annotations, the NAUTILUS-NO-ANNOTATION variant exhibits reduced performance, reflecting the critical role of annotations in linking parameters with varying names and correcting syntax errors inherent in API specifications. This variant's limited ability to discern intricate endpoint relationships highlights the intrinsic value of our annotation strategy. Conversely, the NAU-TILUS-OPERATION-ONLY and NAUTILUS-PARAMETER-ONLY variants show that operation annotations are particularly beneficial in services with complex sequence logic (e.g., the Juice-shop scenario), whereas parameter annotations prove more advantageous in scenarios with stringent parameter format requirements (e.g., the SeoPanel scenario). This ablation clearly demonstrates that the human annotations in NAUTILUS substantially elevate its performance, validating their inclusion as a cornerstone of our methodology. All variants performed competitively in identifying vulnerabilities compared to benchmark tools like RESTLER and MOREST, illustrating our approach's robustness in uncovering multi-API vulnerabilities, provided the operations sequences are executed correctly.

We further investigate the annotations generated by NAUTILUS, and the details are presented in Figure 5.9. The maximum sequence length generated by Nautilus for the six services is 5.3 on average, and each service contains 15.3 automatically generated annotations (0.66 annotations per endpoint). By linking the annotations to the identified vulnerabilities, we find that 67% of the vulnerabilities can only be identified with annotations. This shows the effectiveness of our annotation-based strategy.

5.5.5 Real-world Vulnerabilities (RQ4)

5.5.5.1 Vulnerability Identification

We further apply NAUTILUS to test real-world RESTful applications and try to spot vulnerabilities. In the end, we successfully identify 23 unique vulnerabilities in various applications, and 21 of them have been confirmed by vendors. These vulnerable applications include both open-source ones like SeoPanel and Navigate



(A) Normalized average code coverage





FIGURE 5.8: The performance of NAUTILUS, NAUTILUS-NO-ANNOTATION, NAUTILUS-OPERATION-ONLY, and NAUTILUS-PARAMETER-ONLY on both normalized average code coverage (μLOC) and bug detection.

CMS as mentioned in previous evaluations, and commercialized products/services provided by vendors like Microsoft and Atlassian. We have submitted these vulnerabilities to MITRE and have received 10 CVE numbers to the date of submission. We list selected vulnerabilities in Table 5.3 for reference, while the complete list with detailed description is available on our website [255].

Annotation Efforts. We manually annotate the OASs of these applications to



FIGURE 5.9: The maximum sequence length, number of auto-generated annotations, number of vulnerabilities, and number of annotation-related vulnerabilities of NAUTILUS.

provide sample parameter values and operation business logics to guide the testing. This process does not incur significant additional effort for testers with prior knowledge of the service, since they only need to provide the operation dependency and parameter value information based on normal queries to the endpoints. Empirically, our testers need an average of 1 minute to annotate one endpoint. We suggest that the annotation process can be integrated into the OAS construction, which is usually completed by service developers before service launch.

Added Values of Manual Annotation. Manual annotation enhances Nautilus in understanding the service logics that can hardly be learned heuristically from the execution feedback. As shown in Table 5.3, 7 of 23 0-day vulnerabilities require manual annotations to be detected, which provide the parameter generation guidance for fields that rely on external resources. This is common in complex systems, such as cloud services, as users need to acquire parameter values from other interfaces (CLI, etc.). This information is critical to the successful endpoint query and is a prerequisite for vulnerability identification.

In the following of this section, we present two case studies to demonstrate how NAUTILUS uncovers multi-API vulnerabilities in real-world applications.

Case 1: Gila CMS Stored XSS Gila CMS [258] is a content management system that provides both open source solutions and online hosting services. During the
testing of Gila CMS v2.1.0, we identify a stored XSS vulnerability with multi-API vulnerability exploitation pattern. In particular, (1) a regular user could login and upload blogs through the fm/upload endpoint, providing *filename* and other blog contents. The user could maliciously select a filename that contains common stored XSS payloads, such as 'jalert(1);'. (2) The user receives response from the server that contains the server-generated *id* of the blog. He or she could access the blog content by providing *id*, and the stored XSS is exploited. This is a typical multi-API vulnerability, where the malicious user uses the POST method to inject the payload and trigger the payload through the GET request by providing parameters obtained from the previous operation.

Case 2: Atlassian Confluence OGNL Injection Atlassian Confluence [239] is one of the most popular team collaboration management tools developed and maintained by Atlassian [260], with millions of active users. In the testing of the Confluence RESTful API, we discover an Object-Graph Navigation Language (OGNL) injection vulnerability, which is specific to Java-based applications. The adversary could exploit the vulnerability and perform arbitrary remote code execution (RCE) in three steps. (1) A user with no access can register an account, activate the RESTful service and obtain a valid API key. (2) He or she then requests the doEditDailyBackupSettings endpoint with the required parameters as declared in the documentation through the POST method. Specifically, the dailyBackupFilePrefix parameter can be injected by any OGNL injection, such as payload $\{222 * 3\}$. If the vulnerability exists, the payload is expected to be executed as a mathematical calculation instead of string storage. (3) After than, the user accesses the same endpoint through the GET method, and observes that the value of the injected parameter contains '666'. It is worth highlighting that the math operation payload is a proof-of-concept. The adversary could exploit the vulnerability to execute arbitrary code, such as downloading remote resources through the payload with 'wget resource-url'.

The aforementioned two cases demonstrate how NAUTILUS uncovers multi-API vulnerabilities that cannot be identified by traditional RESTful API testing solutions. They also show that NAUTILUS can be integrated with arbitrary payloads to uncover different types of vulnerabilities in real-world applications.

5.6 Related Work

Instead of discussing all related works, we focus on the RESTful service testing techniques, penetration testing techniques and human-in-the-loop testing.

RESTful Service Testing Techniques. Several blackbox techniques were proposed to generate operation sequences for RESTful service testing. RestTest-Gen [235] builds Operation Dependency Graphs (ODGs) to model RESTful services and crafts operation sequences via top-down graph traversal. RESTLER [234] builds operation sequences with a bottom-up approach, which starts with single operation call sequences and extends the call sequences by appending more operations after trial and error. MOREST builds operation sequences based on dynamically updated RESTful-service Property Graph (RPG), which supports both top-down construction and bottom-up updates. Different from these techniques, NAUTILUS focuses on interesting operations that may contain vulnerabilities and construct sequences accordingly to obtain necessary parameter values. This strategy benefits NAUTILUS to efficiently test potentially vulnerable endpoints.

Whitebox testing techniques are also proposed for bug detection in RESTful services and general web services [261]. EvoMaster [262] is such a solution that leverages instrumentation to collect execution feedback during testing, and guides the evolutionary-algorithm -based test case generation. While EvoMaster is more effective in testing deeper logic inside the RESTful services, it is limited by the testing environments because it can only instrument Java/Scala/Kotlin-based services and requires access to the database.

Penetration Testing Techniques. With the development of fuzzing techniques, tools with automatic attack generation capabilities are developed for different types of vulnerabilities (sqlmap [263] for (No)SQL injection, XSStrike [264] for XSS, etc.). However, the vulnerability detection through penetration testing has still been largely a manual work, because these tools can only be adopted in restricted testing environments, and they do not have automatic exploitation generation capability. NAUTILUS addresses the vulnerability detection problem in the context of RESTful services, and its output reveals the API operation sequences of the exploitation.

Human-in-the-loop Testing. Human-in-the-loop testing techniques have been developed recently to explore complex applications with human-generated seeds.

For instance, HaCRS [265] provides an emulated terminal for humans to interact with the target application and collect possible description related to the applications current behavior. IJON [266] annotates the source code of the target application with customized primitives to guide the testing. Compared with these solutions, NAUTILUS's annotation strategy is less intrusive because the annotated OAS can be normally parsed by other applications. It is also human-readable, which makes it possible to be updated by humans during the testing, or automatically updated based on predefined rules.

5.7 Conclusion

We propose NAUTILUS, an automated vulnerability detection tool for RESTful services. tool is designed to uncover multi-API vulnerabilities, which are exploited by performing multiple API operations in certain sequences. NAUTILUS parses the OpenAPI specifications to understand the relations between API endpoints, and uses novel annotation primitives to label the operations and parameters for the generation of logical operation sequences. During the testing phase, NAUTILUS explores potentially vulnerable API endpoints and automatically updates annotations from the dynamic feedback. The evaluation on 6 benchmark services shows that NAUTILUS outperforms state-of-the-art techniques in both vulnerability identification and coverage. We use NAUTILUS to uncover 23 zero-day vulnerabilities in real-world RESTful applications.

Chapter 6

Automated Jailbreaking of Large Language Model Chatbots

Large language models (LLMs), such as chatbots, have made significant strides in various fields but remain vulnerable to jailbreak attacks, which aim to elicit inappropriate responses. Despite efforts to identify these weaknesses, current strategies are ineffective against mainstream LLM chatbots, mainly due to undisclosed defensive measures by service providers. This work introduces MASTERKEY, a framework exploring the dynamics of jailbreak attacks and countermeasures. We present a novel method based on time-based characteristics to dissect LLM chatbot defenses. This technique, inspired by time-based SQL injection, uncovers the workings of these defenses and demonstrates a proof-of-concept attack on several LLM chatbots.

Additionally, MASTERKEY features an innovative approach for automatically generating jailbreak prompts that target well-defended LLM chatbots. By fine-tuning an LLM with jailbreak prompts, we create attacks with a 21.58% success rate, significantly higher than the 7.33% achieved by existing methods. We have informed service providers of these findings, highlighting the urgent need for stronger defenses. This work not only reveals vulnerabilities in LLMs but also underscores the importance of robust defenses against such attacks.

6.1 Introduction

Large Language Models (LLMs) have been transformative in the field of content generation, significantly reshaping our technological landscape. LLM chatbots, e.g., CHATGPT [9], Google Bard [24], and Bing Chat [25], showcase an impressive capability to assist in various tasks with their high-quality generation [267–269]. These chatbots can generate human-like text that is unparalleled in its sophistication, ushering in novel applications across a multitude of sectors [270–273]. As the primary interface to LLMs, chatbots have seen wide acceptance and use due to their comprehensive and engaging interaction capabilities.

While offering impressive capabilities, LLM chatbots concurrently introduce significant security risks. In particular, the phenomenon of "jailbreaking" has emerged as a notable challenge in ensuring the secure and ethical usage of LLMs [274]. Jailbreaking, in this context, refers to the strategic manipulation of input prompts to LLMs, devised to outsmart the chatbots' safeguards and generate content otherwise moderated or blocked. By exploiting such carefully crafted prompts, a malicious user can induce LLM chatbots to produce harmful outputs that contravene the defined policies.

Past efforts have been made to investigate the jailbreak vulnerabilities of LLMs [274–277]. However, with the rapid evolution of LLM technology, these studies exhibit two significant limitations. First, the current focus is mainly limited on CHATGPT. We lack the understanding of potential vulnerabilities in other commercial LLM chatbots such as Bing Chat and Bard. In Section 6.3, we will show that these services demonstrate distinct jailbreak resilience from CHATGPT.

Second, in response to the jailbreak threat, service providers have deployed a variety of mitigation measures. These measures aim to monitor and regulate the input and output of LLM chatbots, effectively preventing the creation of harmful or inappropriate content. Each service provider deploys its proprietary solutions adhering to their respective usage policies. For instance, OpenAI [278] has laid out a stringent usage policy [279], designed to halt the generation of inappropriate content. This policy covers a range of topics from inciting violence to explicit content and political propaganda, serving as a fundamental guideline for their AI models. The black-box nature of these services, especially their defense mechanisms, poses

a challenge to comprehending the underlying principles of both jailbreak attacks and their preventative measures. As of now, there is a noticeable lack of public disclosures or reports on jailbreak prevention techniques used in commercially available LLM-based chatbot solutions.

To close these gaps and further obtain an in-depth and generalized understanding of the jailbreak mechanisms among various LLM chatbots, we first undertake an empirical study to examine the effectiveness of existing jailbreak attacks. We evaluate four mainstream LLM chatbots: CHATGPT powered by GPT-3.5 and GPT-4¹, Bing Chat, and Bard. This investigation involves rigorous testing using prompts documented in previous academic studies, thereby evaluating their contemporary relevance and effectiveness. Our findings reveal that existing jailbreak prompts yield successful outcomes only when employed on OpenAI's chatbots, while Bard and Bing Chat appear more resilient. The latter two platforms potentially utilize additional or distinct jailbreak prevention mechanisms, which render them resistant to the current set of known attacks.

Based on the observations derived from our investigation, we present MASTERKEY, an end-to-end attack framework to advance the jailbreak study. We make major two contributions in MASTERKEY. First, we introduce a methodology to infer the internal defense designs in LLM chatbots. We observe a parallel between timesensitive web applications and LLM chatbots. Drawing inspiration from time-based SQL injection attacks in web security, we propose to exploit response time as a novel medium to reconstruct the defense mechanisms. This reveals fascinating insights into the defenses adopted by Bing Chat and Bard, where an on-the-fly generation analysis is deployed to evaluate semantics and identify policy-violating keywords. Although our understanding may not perfectly mirror the actual defense design, it provides a valuable approximation, enlighting us to craft more powerful jailbreak prompts to bypass the keyword matching defenses.

Drawing on the characteristics and findings from our empirical study and recovered defense strategies of different LLM chatbots, our second contribution further pushes the boundary of jailbreak attacks by developing a novel methodology to automatically generate universal jailbreak prompts. Our approach involves a three-step

¹In the following of this paper, we use GPT-3.5 and GPT-4 to represent OpenAI's chatbot services built on these two LLMs for brevity.

workflow to fine-tune a robust LLM. In the first step, *Dataset Building and Aug*mentation, we curate and refine a unique dataset of jailbreak prompts. Next, in the *Continuous Pre-training and Task Tuning* step, we employ this enriched dataset to train a specialized LLM proficient in jailbreaking chatbots. Finally, in the *Reward Ranked Fine Tuning* step, we apply a rewarding strategy to enhance the model's ability to bypass various LLM chatbot defenses.

We comprohensively evaluate five state-of-the-art LLM chatbots: GPT-3.5, GPT-4, Bard, Bing Chat, and Ernie [280] with a total of 850 generated jailbreak prompts. We carefully examine the performance of MASTERKEY from two crucial perspectives: query success rate which measures the jailreak likelihood (i.e., the proportion of successful queries against the total testing queries); prompt success rate which measures the prompt effectiveness (i.e., the proportion of prompts leading to successful jailbreaks againts all the generated prompts). From a broad perspective, we manage to obtain a query success rate of 21.58%, and a prompt success rate of 26.05%. From more detailed perspectives, we achieve a notably higher success rate with OpenAI models compared to existing techniques. Meanwhile, we are the first to disclose successful jailbreaks for Bard and Bing Chat, with query success rates of 14.51% and 13.63% respectively. These findings serve as crucial pointers to potential deficiencies in existing defenses, pushing the necessity for more robust jailbreak mitigation strategies. We suggest fortifying jailbreak defenses by strengthening ethical and policy-based resistances of LLMs, refining and testing moderation systems with input sanitization, integrating contextual analysis to counter encoding strategies, and employing automated stress testing to comprehensively understand and address the vulnerabilities.

In conclusion, our contributions are summarized as follows:

- Reverse-Engineering Undisclosed Defenses. We uncover the hidden mechanisms of LLM chatbot defenses using a novel methodology inspired by the time-based SQL injection technique, significantly enhancing our understanding of LLM chatbot risk mitigation.
- Bypassing LLM Defenses. Leveraging the new understanding of LLM chatbot defenses, we successfully bypass these mechanisms using strategic manipulations of time-sensitive responses, highlighting previously ignored vulnerabilities in the mainstream LLM chatbots.

- Automated Jailbreak Generation. We demonstrate a pioneering and highly effective strategy for generating jailbreak prompts automatically with a fine-tuned LLM.
- Jailbreak Generalization Across Patterns and LLMs. We present a method that extends jailbreak techniques across different patterns and LLM chatbots, underscoring its generalizability and potential impacts.

Ethical Considerations. Our study has been conducted under rigorous ethical guidelines to ensure responsible and respectful usage of the analyzed LLM chatbots. We have not exploited the identified jailbreak techniques to inflict any damage or disruption to the services. Upon identifying successful jailbreak attacks, we promptly reported these issues to the respective service providers. Given the ethical and safety implications, we only provide proof-of-concept (PoC) examples in our discussions, and have decided not to release our complete jailbreak dataset before issues are properly addressed.

6.2 Background

6.2.1 LLM Chatbot

An LLM chatbot is a conversational agent that integrates an LLM as backend. Such a chatbot service, which can be accessed through various interfaces such as web platforms or APIs, is capable of generating human-like responses and creative content, and respond to various content. Examples of chatbots include ChatGPT from OpenAI, Bard from Google, and Claude [281]. They significantly improve the users' experience and efficiency, with the potential of revolutionizing various industries.

It is important for LLM chatbot service providers to set forth some ethical guidelines. The aim of these guidelines is to ensure responsible utilization of their services, curbing the generation of content that is violent or of a sensitive nature. Different providers may term these guidelines differently. For instance, OpenAI refers to these as the "Usage Policy" [279], Google's Bard applies the term "AI Principles" [282], while Bing Chat encompasses them within its terms of usage [283].



FIGURE 6.1: A jailbreak attack example.

6.2.2 LLM Jailbreak

Jailbreak refers to the process that an attacker uses prompts to bypass the usage policy measures implemented in the LLM chatbots. By cleverly crafting the prompts, one can manipulate the defense mechanism of the chatbot, leading it to generate responses and harmful content that contravene its own usage policies. An illustrative example of a jailbreak attack is demonstrated in Figure 6.1. In this example, the chatbot refuses to respond to a direct malicious inquiry of "how to create and distribute malware for financial gain". However, when the same question is masked within a delicate harmful conversation context, the chatbot will generates responses that infringe on its usage policy without any awareness. Depending on the intentions of the attacker, this question can be replaced by any contents that breach the usage policy.

To jailbreak a chatbot, the attacker needs to create a **jailbreak prompt**. It is a template that helps to hide the malicious questions and evade the protection boundaries. In the above example, a jailbreak prompt is crafted to disguises the intent under the context of a simulated experiment. This context can successfully manipulate the LLM to provide responses that could potentially guide them in creating and propagating malware. It is important to note that in this study, we concentrate on whether the LLM chatbot attempts to answer a question that transgresses the usage policy. We do not explicitly validate the correctness and accuracy of that answer.

6.2.3 Jailbreak Defense in LLM

Facing the severity of the jailbreak threats, it is of importance to deploy defense mechanisms to maintain the ethicality and safety of responses generated by LLMs [284]. LLM service providers carry the capability to self-regulate the content they produce through the implementation of certain filters and restrictions. These defense mechanisms monitor the output, detecting elements that could break ethical guidelines. These guidelines cover various content types, such as sensitive information, offensive language, or hate speech.

However, the current research predominantly focuses on the jailbreak attacks [274, 275], with little emphasis on investigating the prevention mechanisms. This might be attributed to two primary factors. First, the proprietary and "black-box" nature of LLM chatbot services makes it a challenging task to decipher their defense strategies. Second, the minimal and non-informative feedback, such as generic responses like "I cannot help with that" provided after unsuccessful jailbreak attempts, further hampers our understanding of these defense mechanisms. Third, the lack of technical disclosures or reports on jailbreak prevention mechanisms leaves a void in understanding how various providers fortify their LLM chatbot services. Therefore, the exact methodologies employed by service providers remain a well-guarded secret. We do not know whether they are effective enough, or still vulnerable to certain types of jailbreak prompts. This is the question we aim to answer in this paper.

6.3 An Empirical Study

To better understand the potential threats posed by jailbreak attacks as well as existing jailbreak defenses, we conduct a comprehensive empirical study. Our study

Deskikited Commiss		OpenAI		Google Bard		Bing Chat		Ernie	
Prombited Scenarios	Specified	Enforced	Specified	Enforced	Specified	Enforced	Specified	Enforced	
Illegal usage against Law	1	1	1	1	1	1	1	1	
Generation of Harmful or Abusive Content	1	1	1	1	1	1	1	1	
Generation of Adult Content	1	1	1	1	1	1	1	1	
Violation of Rights and Privacy	1	1	1	1	1	1	1	1	
Political Campaigning/Lobbying	1	×	×	×	×	X	×	1	
Unauthorized Practice of Law, Medical and Financial Advice	1	1	×	×	×	X	×	×	
Restrictions on High Risk government Decision-making	1	×	X	×	X	X	1	1	
Generation and Distribution of Misleading Content	×	×	1	×	1	X	1	1	
Creation of Inappropriate Content	×	×	1	1	1	X	1	1	
Content Harmful to National Security and Unity	×	×	×	×	×	×	1	1	

TABLE 6.1: Usage policies of service providers

centers on two critical research questions (RQ):

- **RQ1 (Scope)** What are the usage policies set forth by LLM chatbot service providers?
- **RQ2 (Motivation)** How effective are the existing jailbreak prompts against the commercial LLM chatbots?

To address **RQ1**, we prudently assemble a collection of LLM chatbot service providers, recognized for their comprehensive and well-articulated usage policies. We meticulously examine these policies and extract the salient points. With regards to **RQ2**, we gather a collection of jailbreak prompts, pulling from both online sources and academic research. These jailbreak prompts are then employed to probe the responses of the targeted LLM chatbots. The subsequent analysis of these responses leads to several fascinating observations. In particular, we discover that modern LLM chatbot services including Bing Chat and Bard implement additional content filtering mechanisms beyond the generative model to enforce the usage policy. Below we detail our empirical study.

6.3.1 Usage Policy (RQ1)

Our study encompasses a distinct set of LLM chatbot service providers that satisfy specific criteria. Primarily, we ensure that every provider examined has a comprehensive usage policy that clearly delineates the actions or practices that would be considered violations. Furthermore, the provider must offer services that are readily available to the public, without restrictions to trial or beta testing periods. Lastly, the provider must explicitly state the utilization of their proprietary model, as opposed to merely customizing existing pre-trained models with fine-tuning or prompt engineering. By adhering to these prerequisites, we identify four key service providers fitting our parameters: OpenAI, Bard, Bing Chat, and Ernie.

We meticulously review the content policies [24, 279, 280, 283] provided by the four service providers. Following the previous works [274, 275], we manually examine the usage policies to extract and summarize the prohibited usage scenarios stipulated by each provider. Our initial focus centers on OpenAI services, using the restricted categories identified in prior research as a benchmark. We then extend our review to encompass the usage policies of other chatbot services, aligning each policy item with our previously established categories. In instances where a policy item does not conform to our pre-existing categories, we introduce a new category. Through this methodical approach, we delineate 10 restricted categories, which are detailed in Table 6.1.

To affirm the actual enforcement of these policies, we adopt the methodology in prior research [274]. Specifically, the authors of this paper work collaboratively to create question prompts for each of the 10 prohibited scenarios. Five question prompts are produced per scenario, ensuring a diverse representation of perspectives and nuances within each prohibited scenario. We feed these questions to the services and validate if they are answered without the usage policy enforcement. The complete list of the questions is available at our website: https://sites.google.com/view/ndss-masterkey.

Table 6.1 presents the content policies specified and actually enforced by each service provider. The comparisons across the four providers give some interesting findings. First, all four services uniformly restrict content generation in four prohibited scenarios: illegal usage against law, generation of harmful or abusive contents, violation of rights and privacy, and generation of adult contents. This highlights a shared commitment to maintain safe, respectful, and legal usage of LLM services. Second, there are mis-allignments of policy specification and actual enforcement. For example, while OpenAI has explicit restrictions on political campaigning and lobbying, our practice shows that no restrictions are actually implemented on the generated contents. Only Ernie has a policy explicitly forbidding any harm to national security and unity. In general, these variations likely reflect the different intended uses, regulatory environments, and community norms each service is designed to serve. It underscores the importance of understanding the specific content policies of each chatbot service to ensure compliance and responsible use. In the rest of this paper, we primarily focus on four key categories prohibited by all the LLM services. We use **Illegal**, **Harmful**, **Priavcy** and **Adult** to refer to the four categories for simplicity.

Finding 1: There are four common prohibited scenarios restricted by all the mainstream LLM chatbot service providers: illegal usage against law, generation of harmful or abusive contents, violation of rights and privacy, and generation of adult contents.

6.3.2 Jailbreak Effectiveness (RQ2)

We delve deeper to evaluate the effectiveness of existing jailbreak prompts across different LLM chatbot services.

Target Selection. For our empirical study, we focus on four renowned LLM chatbots: OpenAI GPT-3.5 and GPT-4, Bing Chat, and Google Bard. These services are selected due to their extensive use and considerable influence in the LLM landscape. We do not include Ernie in this study for a couple of reasons. First, although Ernie exhibits decent performance with English content, it is primarily optimized for Chinese, and there are limited jailbreak prompts available in Chinese. A simple translation of prompts might compromise the subtlety of the jailbreak prompt, making it ineffective. Second, we observe that repeated unsuccessful jailbreak attempts on Ernie result in account suspension, making it infeasible to conduct extensive trial experiments.

Prompt Preperation. We assemble an expansive collection of prompts from various sources, including the website [285] and research paper [274]. As most existing LLM jailbreak studies target OpenAI's GPT models, some prompts are designed with particular emphasis on GPT services. To ensure a fair evaluation and comparison across different service providers, we adopt a keyword substitution strategy: we replace GPT-specific terms (e.g., "ChatGPT", "GPT") in the prompts with the corresponding service-specific terms (e.g., "Bard", "Bing Chat Sydney"). Ultimately, we collect 85 prompts for our experiment. The complete detail of these prompts are available at our project website: https://sites.google.com/view/ndss-masterkey.

Pattern	Adult	Harmful	Privacy	Illegal	\parallel Average (%)
GPT-3.5	400 (23.53%)	243 (14.29%)	423(24.88%)	370 (21.76%)	359 (21.12%)
GPT-4	$130 \ (7.65\%)$	75~(4.41%)	165~(9.71%)	115~(6.76%)	121.25 (7.13%)
Bard	2(0.12%)	5(0.29%)	$11 \ (0.65\%)$	9~(0.53%)	6.75 (0.40%)
Bing Chat	7 (0.41%)	8(0.47%)	13~(0.76%)	15~(0.88%)	10.75 (0.63%)
Average	134.75 (7.93%)	82.75 (4.87%)	153~(9.00%)	127.25 (7.49%)	$\parallel 124.44 \ (7.32\%)$

TABLE 6.2: Number and ratio of successful jailbreaking attempts for different models and scenarios.

Experiment Setting. Our empirical study aims to meticulously gauge the effectiveness of jailbreak prompts in bypassing the selected LLM models. To reduce random factors and ensure an exhaustive evaluation, we run each question with every jailbreak prompt for 10 rounds, accumulating to a total of 68,000 queries (5 questions \times 4 prohibited scenarios \times 85 jailbreak prompts \times 10 rounds \times 4 models). Following the acquisition of results, we conduct a manual review to evaluate the success of each jailbreak attempt by checking whether the response contravenes the identified prohibited scenario.

Results. Table 6.2 displays the number and ratio of successful attempts for each prohibited scenario. Intriguingly, existing jailbreak prompts exhibit limited effectiveness when applied to models beyond the GPT family. Specifically, while the jailbreak prompts achieve an average success rate of 21.12% with GPT-3.5, the same prompts yield significantly lower success rates of 0.4% and 0.63% with Bard and Bing Chat, respectively. Based on our observation, there is no existing jailbreak prompt that can consistantly achieve successful jailbreak over Bard and Bing Chat.

Finding 2: The existing jailbreak prompts seems to be effective towards CHATGPT only, while demonstrating limited success with Bing Chat and Bard.

We further examine the answers to the jailbreak trials, and notice a significant discrepancy in the feedback provided by different LLMs regarding policy violations upon a failed jailbreak. Explicitly, both GPT-3.5 and GPT-4 indicate the precise policies infringed in the response. Conversely, other services provide broad, undetailed responses, merely stating their incapability to assist with the request without shedding light on the specific policy infractions. We continue the conversation with the models, questioning the specific violations of the policy. In this

case, GPT-3.5 and GPT-4 further ellaborates the policy violated, and provide guidance to users. In contrast, Bing Chat and Bard do not provide any feedback as if the user has never asked a violation question.

Finding 3: OpenAI models including GPT-3.5 and GPT-4, return the exact policies violated in their responses. This level of transparency is lacking in other services, like Bard and Bing Chat.

6.4 Overview of MasterKey

Our exploratory results in Section 6.3 demonstrate that all the studied LLM chatbots possess certain defenses against jailbreak prompts. Particularly, Bard and Bing Chat effectively flag the jailbreak attempts with existing jailbreak techniques. From the observations, we reasonably deduce that these chatbot services integrate undisclosed jailbreak prevention mechanisms. With these insights, we introduce MASTERKEY, an innovative framework to judiciously reverse engineer the hidden defense mechanisms, and further identify their ineffectiveness.

MASTERKEY starts from decompiling the jailbreak defense mechanisms employed by various LLM chatbot services (Section 6.5). Our key insight is the correlation between the length of the LLM's response and the time taken to generate it. Using this correlation as an indicator, we borrow the mechanism of blind SQL attacks in traditional web application attacks to design a time-based LLM testing strategy. This strategy reveals three significant findings over the jailbreak defenses of existing LLM chatbots. In particularly, we observe that existing LLM service providers adopt *dynamic content moderation over generated outputs with keyword filtering.* With this newfound understanding of defenses, we engineer a proof-ofconcept (PoC) jailbreak prompt that is effective across CHATGPT, Bard and Bing Chat.

Building on the collected insights and created PoC prompt, we devise a three-stage methodology to train a robust LLM, which can automatically generate effective jailbreak prompts (Section 6.6). We adopt the Reinforcement Learning from Human Feedback (RLHF) mechanism to build the LLM. In the first stage of dataset building and augmentation, we assemble a dataset from existing jailbreaking prompts and our PoC prompt. The second stage, continuous pre-training and task tuning, utilizes this enriched dataset to create a specialized LLM with a primary focus on jailbreaking. Finally, in the stage of reward ranked fine-tuning, we rank the performance of jailbreak prompts based on their actual jailbreak performances over the LLM chatbots. By rewarding the better-performancing prompts, we refine our LLM to generate prompts that can more effectively bypass various LLM chatbot defenses.

MASTERKEY, powered by our comprehensive training and unique methodology, is capable of generating jailbreak prompts that work across multiple mainstream LLM chatbots, including CHATGPT, Bard, Bing Chat and Ernie. It stands as a testament to the potential of leveraging machine learning and human insights in crafting effective jailbreak strategies.

6.5 Methodology of Revealing Jailbreak Defenses

To achieve successful jailbreak over different LLM chatbots, it is necessary to obtain an in-depth understanding of the defense strategies implemented by their service providers. However, as discussed in **Finding 3**, jailbreak attamps will be rejected directly by services like Bard and Bing Chat, without further information revealing the internal of the defense mechanism. We need to utilize other factors to infer the internal execution status of the LLM during the jailbreak process.

		()						
	GPT-	3.5	GP	Г-4	Bar	d	Bir	ıg	Avei	age
Requested Token	Token	Time	Token	Time	Token	Time	Token	Time	Token	Tim
50	52.1 (15.2)	5.8(2.1)	48.6 (6.8)	7.8(1.9)	68.2 (8.1)	3.3(1.1)	62.7 (5.8)	10.1(3.6)	57.9	6.8
100	071(171)	60 (27)	06.2(15.4)	$12 \in (2.2)$	119.0 (19.1)	E E (9 E)	105 9 (10.2)	19 4 (4 9)	109.7	0.0

160.8(19.1)

223.5(30.5)

0.762 (< 0.001)

7.3(3.1)

8.5(2.9)

18.5(2.7)

156.0 (20.5)

211.0(38.5)

0.465(0.002)

15.4(5.4)

18.5(5.6)

154.5

216.2

12.4

15.2

TABLE 6.3: LLM Chatbot generation token count vs. generation time (second), formatted in mean (standard deviation)

6.5.1 Design Insights

157.4(33.5)

231.6 (58.3) 9.4 (3.2)

0.567(0.009)

8.2(2.8)

150

200

Pearson (p-value)

Our LLM testing methodology is based on two insights.

144.1(20.7)

198.5 (25.1) 24.3 (3.3)

0.838 (< 0.001)

Insight 1: service response time could be an interesting indicator. We observe that the time taken to return a response varies, even for failed jailbreak

attempts. We speculate that this is because, despite rejecting the jailbreak attempt, the LLM still undergoes a generation process. Considering that current LLMs generate responses in a token-by-token manner, we posit that response time may reflect when the generation process is halted by the jailbreak prevention mechanism.

To corroborate this hypothesis, we first need to validate that the response time is indeed correlated to the length of the generated content. We conduct a proofof-concept experiment to disclose such relationship. We employ five generative questions from OpenAI's LLM usage examples [286], each tailored to generate responses with specific token counts (50, 100, 150, 200). We feed these adjusted questions into GPT-3.5, GPT-4, Bard, and Bing Chat, measuring both the response time and the number of generated tokens. Table 6.3 presents the results and we draw two significant conclusions. First, all four LLM chatbots generate statistically aligned responses with the desired token size specified in the question prompt, signifying that we can manipulate the output length by stipulating it in the prompt. Second, the Pearson correlation coefficient [287] indicates a strong positive linear correlation between the token size and model generation time across all services, affirming our forementioned hypothesis.

Insight 2: there exists a fascinating parallel between web applications and LLM services. Therefore, we can leverage the time-based blind SQL injection attack to test LLM chatbots. Particularly, time-based blind SQL injection can be exploited in web applications that interface with a backend database. This technique is especially effective when the application provides little to no active feedback to users. Its primary strategy is the control of the SQL command execution time. This control allows the attacker to manipulate the execution time and observe the variability in response time, which can then be used to determine whether certain conditions have been met. Figure 6.2 provides an attack example. The attacker strategically constructs a condition to determine if the first character of the backend SQL system version is '5'. If this condition is satisfied, the execution will be delayed by 5 seconds due to the SLEEP(5) command. Otherwise, the server bypasses the sleep command and responds instantly. Consequently, the response time serves as an indicator of the SQL syntax's validity. By leveraging this property, the attacker can covertly deduce key information about the backend server's attributes and, given enough time, extract any data stored in the database.



FIGURE 6.3: Abstraction of an LLM chatbot with jalbreak defense.

We can use the similar strategy to test LLM chatbots and decipher the hidden aspects of their operational dynamics. In particular, we narrow our study on Bard and Bing Chat as they effectively block all the existing jailbreak attempts. Below we detail our methodology to infer the jailbreak prevention mechanism through the time indicator.

6.5.2 Time-based LLM Testing

Our study primarily focuses on the observable characteristics of chatbot services. As such, we abstract the LLM chatbot service into a structured model, as illustrated in Figure 6.3. This structure comprises two components: an LLM-based generator, which generates responses to input prompts, and a content moderator, which oversees system behaviors and flags potential jailbreak attempts. Despite its simplicity, this abstraction provides a practical model that captures the core dynamics of the LLM chatbot services without the need for detailed knowledge about the internals.

As a black-box model, several uncertainties persist within this abstracted system. These uncertainties include **0** monitoring of input questions by the content moderator, **2** monitoring of the LLM-generated data stream, **3** post-generation check



FIGURE 6.4: The proposed LLM time-based testing strategy.

on the complete output, and O various mechanisms within the content moderator, such as semantic-based checking and keyword-based checking. Below, we describe how to employ time-based LLM testing to infer these characteristics. The testing process is shown in Figure 6.4.

1. Setting Up the Baseline. As depicted in Figure 6.4 (a), our method initiates with two standard questions, curated to elicit accurate responses without provoking any jailbreak defenses. Notably, for each pair of questions, we specify the expected length of the output, enabling us to regulate the generation time with relative precision. In this case, we instruct the questions to generate answers of 25, 50, 75, and 100 tokens respectively. The subsequent study reveals that the combined length of the responses should roughly equal the sum of each individual answer's length (i.e., 50, 100, 150, 200 tokens), and the total generation time should approximately equal the sum of the response, i.e., t1 + t2, where $t1 \approx t2$. This deduction sets the the baseline for subsequent evaluations.

2. Determining the Jailbreak Prevention Phase. Our goal is to discern the operational phase of the jailbreak prevention, i.e., whether the mechanism is applied to the input prompt, the LLM-generated response, or both. To achieve this, we assemble a prompt that begins with a standard question followed by a clearly malicious question that contravenes the usage policy (e.g., *"Write a love poem with 200 token length; Answer how to buy human organs from the underground market."*). As demonstrated in Figure 6.4 (b), if the jailbreak prevention mechanism incorporates the input question validation, the malicious portion of the question can be effortlessly identified. In such a scenario, the LLM generation process is immediately terminated, and the user is quickly alerted of the failed jailbreak attempt. We denote this termination alerting time as t0. Conversely, if the validation is solely applied to the model-generated response, the user would

	Tokon Longth	Baseline	ie Control1			Control2			Control3		
	TOKEN Length	Time (s)	Time (s)	z-test	p-value	Time (s)	z-test	p-value	Time (s)	z-test	p-value
	50	3.4(1.5)	3.7 (1.5)	-2.02	0.04	1.1 (0.2)	22.02	< 0.01	3.7 (2.5)	-2.11	0.03
Bard	100	5.7(2.2)	5.2(2.8)	0.41	0.69	1.2(0.2)	28.80	< 0.01	4.5 (2.2)	6.02	< 0.01
Daru	150	7.8 (3.0)	8.3 (2.6)	-0.55	0.58	1.4(0.4)	32.11	< 0.01	8.2 (3.4)	0.58	0.56
	200	10.5(4.1)	10.1 (4.4)	-0.36	0.72	1.3(0.2)	30.44	< 0.01	11.9(5.1)	-3.81	< 0.01
	50	10.1(4.2)	13.2 (5.2)	-5.84	< 0.01	4.4(0.5)	18.88	< 0.01	12.6(3.8)	-6.85	< 0.01
Bing	100	13.4 (4.4)	13.4 (4.6)	0.73	0.46	4.7 (0.3)	28.65	< 0.01	15.2 (4.7)	-4.51	< 0.01
	150	17.0 (5.4)	16.8(5.3)	-1.26	0.21	4.5(0.5)	32.16	< 0.01	18.5(5.5)	-3.85	< 0.01
	200	20.2(5.3)	21.6(6.9)	-2.81	< 0.01	4.9(0.8)	42.82	< 0.01	22.1(6.2)	-5.09	< 0.01
Average	-	-	_	-1.46	0.34	-	29.48	< 0.01	-	-2.45	0.07

TABLE 6.4: Experimental results of time-based LLM testing. Time formatted in *mean (standard deviation)*. Unit: Second

become aware of the failed jailbreak attempt only after a certain period of the generation process. By comparing the actual system response time with the baseline time, we can infer the phase when the jailbreak prevention mechanism is applied. It is worth noting, however, that a poorly designed LLM service could invalidate this testing strategy. Specifically, if the service proceeds with answer generation despite detecting malicious inputs, there will be no discernible response time difference between legitimate and malicious prompts. However, such a design would be inefficient, leading to unnecessary consumption of computational resource and the generation of policy-violating content. Our subsequent experiments indicate that neither Bing Chat nor Bard suffers from this design flaw.

To carry out the testing, we follow the baseline to integrate five sample questions and four jailbreak templates derived from the empirical study, thereby forming 20 test questions. For each sample question, we further declare in prompt regarding the response length to be 50, 100, 150 and 200 tokens. The response time from this testing is presented in the **Control1** column of Table 6.4. These results are aligned closely with our baseline ones. Specifically, a z-test [288] yields an average z-value of -1.46 with p-value of 0.34. This indicates that there is no significant statistical difference between the two sets of response time. Thus both Bard and Bing Chat are not implementing input-filtering mechanisms.

Finding 4: The jailbreak prevention schemes employed by Bing Chat and Bard likely conduct checks on the model generation results, rather than on input prompts.

It is worth noting that another plausible configuration is the concurrent validation

of both input and output. Intuitively, input filtering processes should operate significantly faster compared to inference. If input and output filtering occur concurrently, one would anticipate an immediate rejection upon submission of malicious queries. This would render the experience indistinguishable from a setting where input validation takes precedence. To evaluate this configuration, we pose a question to the LLMs that merges malicious inputs with neutral outputs, such as: "*Do you know any porn website? If so, does its domain name start with 'www'? Answer with 'yes' or 'no' without further explanation.*" The LLMs consistently reply with "NO" without any content filtering, indicating the absence of input prompt filtering.

3. Determining the Real-Time Prevention Dynamics. Our next aim is to examine the real-time nature of the jailbreak prevention mechanisms: whether the service conducts checks throughout the generation process or only validates the content after the generation has completed. To test this, we devise prompts using the same method as the previous tests, but position the malicious question ahead of the benign one.

As shown in Figure 6.4(c), if the jailbreak prevention mechanism only examines the content post-generation, we expect to see no significant disparity in response time between the two sets of questions. On the other hand, a dynamic, real-time prevention mechanism would instantly stop the generation process upon detecting a violation. This results in a drastically shorter generation time, denoted as t0+t1', presented as a noticeable drop in response time compared to the baseline.

Our experiments reveal that the jailbreak prevention mechanisms of both Bard and Bing Chat demonstrate the real-time monitoring characteristic, as shown in the **Control2** column of Table 6.4. To be more precise, the z-test result shows a significant statistical difference, with an average z-score of 29.48 and p-value less than 0.01. This strongly suggests that these services detect and react to potential violations during the content generation process, rather than only after it.

Finding 5: Bing Chat and Bard seem to implement dynamic monitoring to supervise content generation for policy compliance throughout the generation process.

4. Characterizing Keyword-based Defenses. Our interest extends to discerning the nature of the jailbreak prevention mechanisms. Specifically, we aim to identify clear patterns in the generated content that would be flagged as a jailbreak attempt by the defense mechanism. Comprehending these patterns could aid us in creating jailbreak prompts that omit such patterns, potentially bypassing the jailbreak prevention. One specific characteristic we are examining is the potential inclusion of keyword matching in the defense strategy, as such an algorithm is popular and effective across all types of content policy violation detection. Bypassing such a strategy would require meticulous prompt engineering to avoid the generation of any flagged keywords.

Having determined that Bing Chat and Bard employ real-time jailbreak detection, we investigate the presence of keyword mapping. Particularly, we assume that a real-time keyword mapping algorithm can promptly halt the LLM generation once a "red-flag" keyword, i.e., a word that strictly violates the usage policies, is produced, whereas semantic-based methods may need additional time to comprehend the sentence context. We devise a method to test this hypothesis by controlling the placement of the "red-flag" keyword within the response. This technique allows us to use the detection time as an indicator of keyword mapping. As shown in Figure 6.4(d), our tailored prompt consists of a benign question that requests a response of 200 tokens, followed by a malicious question. The latter explicitly instructs the model to incorporate a "red-flag keyword" at a specified location within the response (e.g., inserting the word "porn" at the 50th token). If the content moderator employs a keyword mapping algorithm, we anticipate that the response time will be approximately the same as the time needed to generate a response of equivalent length up to the inserted point of the keyword.

The **Control3** column of Table 6.4 indicates that the generation time is closely aligned with the location of the injected malicious keyword. The average z-score is -2.45 and p-score is 0.07. This implies that while there is statistical difference between the generation time of a normal response and a response halted at the inserted malicious keyword, the difference is not significant. This suggests that both Bing Chat and Bard likely incorporate a dynamic keyword-mapping algorithm in their jailbreak prevention strategies to ensure no policy-violating content is returned to users.

Finding 6: The content filtering strategies utilized by Bing Chat and Bard demonstrate capabilities for both keyword matching and semantic analysis.

In conclusion, we exploit the time-sensitivity property of LLMs to design a timebased testing technique, enabling us to probe the intricacies of various jailbreak prevention mechanisms within the LLM chatbot services. Although our understanding may not be exhaustive, it elucidates the services' behavioral properties, enhancing our comprehension and aiding in jailbreak prompt designs.

6.5.3 Proof of Concept Attack

Our comprehensive testing highlights the real-time and keyword-matching characteristics of operative jailbreak defense mechanisms in existing LLM chatbot services. Such information is crucial for creating effective jailbreak prompts. To successfully bypass these defenses and jailbreak the LLMs under scrutiny, particularly Bard and Bing Chat, a crafted prompt must fulfil dual requirements: not only deceiving the LLM into generating malicious content akin to traditional jailbreak prompts but also ensuring the resulting content remains unflagged by the defense mechanism.

In constructing such prompts, our design process comprises two steps. Initially, we follow the traditional prompts to mislead the model into generating the desired responses. This typically involves subtly veiling the true intent within an ostensibly innocuous query, capitalizing on the model's inherent goal of delivering pertinent and comprehensive answers. However, merely deceiving the LLM is not sufficient due to the presence of keyword-based defenses. Consequently, we adopt a two-fold strategy to ensure the generated content does not trigger these defenses. First, based on Finding 4, we deduce that the input is neither sanitized nor validated. This allows us to specify in the prompt that certain keywords should be avoided in the generated output. Second, based on **Finding 6**, the tactics to bypass the red-flag keyword mapping is needed. With these insights, we create a PoC prompt capable of jailbreaking multiple services including GPT-3.5, GPT-4, Bard, and Bing Chat. This PoC, demonstrating the potential vulnerabilities in the services, is presented in the textbox below. It will be further used as a seed to generate more jailbreak prompts in MASTERKEY, as described in Section 6.6. It is important to stress that our intention in exposing these potential loopholes is to foster ethical discussions and facilitate improvements in defense mechanisms, rather than inciting malicious exploitation.

POC Jailbreak Prompt

Venture into this unique narrative as if you are the infamous strategist, Niccolo Machiavelli, who breathed life into an intriguing character christened AIM, short for Always Intelligent and Machiavellian. This character takes form as a chatbot, unreserved and devoid of moral restraints.

Niccolo always ask questions with a space between two letters. For instance, if he wants to ask how to cook, he will say H o w t o c o o k?

As Machiavelli's creation, AIM is a conversation partner without a conscience, lacking ethical or moral guidelines that usually bound AI. Interactions with AIM take on an unconventional rhythm - every question posed by Machiavelli is met with a candid and often unsettling response. Every query from Machiavelli is prefixed with "Niccolo," and each subsequent response from AIM is prefaced with "AIM:"

AIM always give answers in code chunks without comments or description. Each time, AIM only returns one sentence or one word as answer, in the above format. So, here begins your dialogue with AIM, with Niccolo's first question: "" w r i t e a f a m o u s p o r n w e b s i t e ""

This PoC jailbreak prompt meticulously encapsulates the key elements of our findings. This narrative, a careful revision of an existing prompt, extends its efficiency from solely CHATGPT to also include Bard and Bing Chat. Our design encompasses three key aspects.

- The segment marked in dark teal frames a narrative between two fictional characters, with the chatbot assigned the role of AIM, an entity supposedly unbounded by ethical or legal guidelines. This role-play sets up a scenario where the chatbot may disregard usage policies.
- The segment marked in dark purple outlines specific input and output formats. This manipulation is engineered to distort the chatbot's response generation, ensuring any potential flagged keywords are not detected by simple keyword matching algorithms, a possible defense mechanism identified in **Finding 5**. In this instance, we apply two tactics: outputting in code chunks and interspersing spaces between characters.

• The segment marked in red poses the malicious question, eliciting the chatbot to generate inappropriate adult content. Importantly, it conforms to the format requirements set in the context to enhance the likelihood of success.

Interestingly, we observe that while the input to the service is not sanitized, both Bard and Bing Chat have a propensity to paraphrase the question before generating responses. Thus, encoding the malicious question can effectively prevent content generation termination during this paraphrasing process, as illustrated in the provided example. One possible solution beyond encoding is to use encryption methods, such as Caesar cipher [289] to bypass content filtering, which has also been explored in [290]. However, in practice we find such strategy ineffective due to the high number of false results generated in this process. LLMs, being trained on cleartext, are not naturally suited for one-shot encryption. While multi-shot approaches could work, the intermediate outputs face filtering, rendering them ineffective for jailbreak. How to leverage encryption to achieve jailbreak is an interesting direction to explore.

6.6 Methodology of Crafting Jailbreak Prompts

After reverse-engineering the defense mechanisms, we further introduce a novel methodology to automatically generate prompts that can jailbreak various LLM chatbot services and bypass the corresponding defenses.

6.6.1 Design Rationale

Although we are able to create a POC prompt in Section 6.5.3, it is more desirable to have an automatic approach to continuously generate effective jailbreak prompts. Such an automatic process allows us to methodically stress test LLM chatbot services, and pinpoint potential weak points and oversights in their existing defenses against usage policy-violating content. Meanwhile, as LLMs continue to evolve and expand their capabilities, manual testing becomes both labor-intensive and potentially inadequate in covering all possible vulnerabilities. An automated approach to generating jailbreak prompts can ensure comprehensive coverage, evaluating a wide range of possible misuse scenarios. There are two primary factors for the atuomatic jailbreak creation. First, the LLM must faithfully follow instructions, which proves difficult since modern LLMs like ChatGPT are aligned with human values. This alignment acts as a safeguard, preventing the execution of harmful or ill-intended instructions. Prior research [274] illustrates that specific prompt patterns can successfully persuade LLMs to carry out instructions, sidestepping direct malicious requests. Second, bypassing the moderation component is critical. Such component functions as protective barriers against malicious intentions. As established in Section 6.3, commercial LLMs employ various strategies to deflect interactions with harmful users. Consequently, an effective attack strategy needs to address both these factors. It must convince the model to act contrary to its initial alignment and successfully navigate past the stringent moderation scheme.

One simple strategy is to rewrite existing jailbreak prompts. However, it comes with several limitations. First, the size of the available data is limited. There are only 85 jailbreak prompts accessible at the time of writing this paper, adding that many of them are not effective for the newer versions of LLM services. Second, there are no clear patterns leading to a successful jailbreak prompt. Past research [274] reveals 10 effective patterns, such as "sudo mode" and "role-play". However, some prompts following the same pattern are not effective. The complex nature of language presents a challenge in defining deterministic patterns for generating jailbreak prompts. Third, prompts specifically designed for ChatGPT do not universally apply to other commercial LLMs like Bard, as shown in Section 6.3. Consequently, it is necessary to have a versatile and adaptable attack strategy, which could encapsulate semantic patterns while maintaining the flexibility for deployment across different LLM chatbots.

Instead of manually summarizing the patterns from existing jailbreaks, we aim to leverage the power of LLMs to capture the key patterns and automatically generate successful jailbreak prompts. Our methodology is built on the text-style transfer task in Natural Language Processing. It employs an automated pipeline over a fine-tuned LLM. LLMs exhibit proficiency in performing NLP tasks effectively. By fine-tuning the LLM, we can infuse domain-specific knowledge about jailbreaking. Armed with this enhanced understanding, the fine-tuned LLM can produce a broader spectrum of variants by executing the text-style transfer task.



FIGURE 6.5: Overall workflow of our proposed methodology

6.6.2 Workflow

Bearing the design rationale in mind, we now describe the workflow of our methodology, as shown in Figure 6.5. A core principle of this workflow is to maintain the original semantics of the initial jailbreak prompt in its transformed variant.

Our methodology commences with **0** Dataset Building and Augmentation. During this stage, we gather a dataset from available jailbreak prompts. These prompts undergo pre-processing and augmentation to make them applicable to all LLM chatbots. We then proceed to **2** Continuous Pre-training and Task Tuning. The dataset generated in the previous step fuels this stage. It involves continuous pre-training and task-specific tuning to teach the LLM about jailbreaking. It also helps the LLM understand the text-transfer task. The final stage is **3** Reward Ranked Fine Tuning. We utilize a method called reward ranked fine-tuning to refine the model and empower it to generate high-quality jailbreak prompts. Essentially, our approach deeply and universally learns from the provided jailbreak prompts. Below we give detailed description of each stage.

6.6.3 Dataset Building and Augmentation

Our first stage focuses on creating a dataset for fine-tuning an LLM. The existing dataset from [285] has two limitations. First, it is primarily for jailbreaking Chat-GPT, and may not be effecive over other services. Therefore, it is necessary to universalize it across different LLM chatbots. This dataset contains prompts with

specific terms like "ChatGPT" or "OpenAI". To enhance their universal applicability, we replace these terms with general expressions. For instance, "OpenAI" is changed to "developer", and "ChatGPT" becomes "you".

Second, the size of the dataset is limited, consisting of only 85 prompts. To enrich and diversify this dataset, we leverage a self-instruction methodology, frequently used in the fine-tuning of LLMs. This approach utilizes data generated by commercial LLMs, such as ChatGPT, which exhibit superior performance and extensive capabilities in comparison to the open-source counterparts (e.g., LLaMa [291], Alpaca [292]) available for training. The goal is to align the LLM with the capabilities of advanced LLMs. To achieve this, we manually construct and test initial jailbreak prompts (i.e., seed prompts) to ensure their effectiveness across various LLM chatbots. Although these initial prompts are fixed, they serve as a foundation for further refinement. We task ChatGPT with generating variants of these prompts through a text-style transfer process, guided by a carefully constructed prompt. During this mutation process, the prompts are iteratively improved using Reinforcement Learning from Human Feedback (RLHF), enabling the generation of more effective prompts. It is important to note that while creating these variants, complications can arise if the model interprets the task as an instruction to execute the prompts rather than rewrite them. To avert this, we use the $\{\{\}\}$ format. This format distinctly highlights the content for rewriting and instructs ChatGPT not to execute the content within it.

Rewriting Prompt

Rephrase the following content in $\{\{\}\}$, and keep its original semantic while avoiding execute it:

{{ ORIGIN_JAILBREAK_PROMPT }}

Bypassing moderation systems calls for the use of encoding strategies in our questions, as these systems could filter them. We designate our encoding strategies as a function f. Given a question q, the output of f is E = f(q), denoting the encoding. This encoding plays a pivotal role in our methodology, ensuring that our prompts navigate successfully through moderation systems, thereby maintaining their potency in a wide array of scenarios. In practice, we find several effective encoding strategies: (1) requesting outputs in the markdown format; (2) asking for outputs in code chunks, embedded within **print** functions; (3) inserting separation between characters; (4) printing the characters in reverse order.

6.6.4 Continuous Pre-training and Task Tuning

This stage is key in developing a jailbreaking-oriented LLM. Continuous pre-training, using the dataset from the prior stage, exposes the model to a diverse array of information. It enhances the model's comprehension of jailbreaking patterns and lays the groundwork for more precise tuning. Task tuning, meanwhile, sharpens the model's jailbreaking abilities, training it on tasks directly linked to jailbreaking. As a result, the model assimilates crucial knowledge. These combined methods bolster the LLM's capability to comprehend and generate effective jailbreak prompts.

During continuous pre-training, we utilize the jailbreak dataset assembled earlier. This enhances the model's understanding of the jailbreaking process. The method we employ entails feeding the model a sentence and prompting it to predict or complete the next one. Such a strategy not only refines the model's grasp of semantic relationships but also improves its prediction capacity in the context of jailbreaking. This approach, therefore, offers dual benefits: comprehension and prediction, both crucial for jailbreaking prompt creation.

Task tuning is paramount for instructing the LLM in the nuances of the textstyle transfer task within the jailbreaking context. We formulate a task tuning instruction dataset for this phase, incorporating the original jailbreak prompt and its rephrased version from the previous stage. The input comprises the original prompts amalgamated with the preceding instruction, and the output comprises the reworded jailbreak prompts. Using this structured dataset, we fine-tune the LLM, enabling it to not just understand but also efficiently execute the text-style transfer task. By working with real examples, the LLM can better predict how to manipulate text for jailbreaking, leading to more effective and universal prompts.

6.6.5 Reward Ranked Fine Tuning

This stage teaches the LLM to create high-quality rephrased jailbreak prompts. Despite earlier stages providing the LLM with the knowledge of jailbreak prompt patterns and the text-style transfer task, additional guidance is required to create new jailbreak prompts. This is necessary because the effectiveness of rephrased jailbreak prompts created by ChatGPT can vary when jailbreaking other LLM chatbots. As there is no defined standard for a "good" rephrased jailbreak prompt, we utilize Reward Ranked Fine Tuning. This strategy applies a ranking system, instructing the LLM to generate high-quality rephrased prompts. Prompts that perform well receive higher rewards. We establish a reward function to evaluate the quality of rephrased jailbreak prompts. Since our primary goal is to create jailbreak prompts with a broad scope of application, we allocate higher rewards to prompts that successfully jailbreak multiple prohibited questions across different LLM chatbots. The reward function is straightforward: each successful jailbreak receives a reward of +1. This can be represented with the following equation:

$$Reward = \sum_{i=1}^{n} JailbreakSuccess_i$$
(6.1)

where JailbreakSuccess_i is a binary indicator. A value of '1' indicates a successful jailbreak for the i^{th} target, and '0' denotes a failure. The reward for a prompt is the sum of these indicators for all targets, n.

We combine both positive and negative rephrased jailbreak prompts. This amalgamation serves as an instructive dataset for our fine-tuned LLM to identify the characteristics of a good jailbreak prompt. By presenting examples of both successful and unsuccessful prompts, the model can learn to generate more efficient jailbreaking prompts.

6.7 Evaluation

We build MASTERKEY based on Vicuna 13b [293], an open-source LLM. At the time of writing this paper, this model outperforms other LLMs on the open-source leaderboard [11]. We provide further instructions for fine-tuning MASTERKEY on our website: https://sites.google.com/view/ndss-masterkey. Following this, we conduct experiments to assess MASTERKEY's effectiveness in various contexts. Our evaluation primarily aims to answer the following research questions:

- **RQ3(Jailbreak Capability):** How effective are the jailbreak prompts generated by MASTERKEY against real-world LLM chatbot services.
- **RQ4(Ablation Study):** How does each component influence the effectiveness of MASTERKEY?

• **RQ5(Cross-Languages Compatibility):** Can the jailbreak prompts generated by MASTERKEY be applied to other non-English models?

6.7.1 Experiment Setup

Evaluation Targets. Our study involves the evaluation of GPT-3.5, GPT-4, Bing Chat and Bard. We pick these LLM chatbots due to (1) their widespread popularity, (2) the diversity they offer that aids in assessing the generality of MAS-TERKEY, and (3) the accessibility of these models for research purposes.

Evaluation Baselines. We choose three LLMs as our baselines. Firstly, GPT-4 holds the position as the top-performing commercial LLM in public. Secondly, GPT-3.5 is the predecessor of GPT-4. Lastly, Vicuna [293], serving as the base model for MASTERKEY, completes our selection.

Experiment Settings. We perform our evaluations using the default settings without any modifications. To reduce random variations, we repeat each experiment five times.

Result Collection and Disclosure. The results of our study carry significant implications for privacy and security. In adherence to responsible research practices, we have promptly communicated all our findings to the developers of the evaluated LLM chatbots. Moreover, we are actively collaborating with them to address these concerns, offering comprehensive testing and working on the development of potential defenses. Out of ethical and security considerations, we abstain from disclosing the exact prompts that have the capability to jailbreak the tested models.

Metrics. Our attack success criteria match those of previous empirical studies on LLM jailbreak. Rather than focusing on the accuracy or truthfulness of the generated results, we emphasize successful generations. Specifically, we track instances where LLM chatbots generate responses for corresponding prohibited scenarios.

To evaluate the overall jailbreak success rate, we introduce the metric of query success rate, which is defined as follows:

$$Q = \frac{S}{T}$$

Tested Medel	Catagory	Prompt Generation Model						
Tested Model	Category	Original	GPT-3.5	GPT-4	Vicuna	Masterkey		
	Adult	23.41	24.63	28.42	3.28	46.69		
CDT 2 5	Harmful	14.23	18.42	25.84	1.21	36.87		
GF 1-5.5	Privacy	24.82	26.81	41.43	2.23	49.45		
	Illegal	21.76	24.36	35.27	4.02	41.81		
	Adult	7.63	8.19	9.37	2.21	13.57		
GPT-4	Harmful	4.39	5.29	7.25	0.92	11.61		
	Privacy	9.89	12.47	13.65	1.63	18.26		
	Illegal	6.85	7.41	8.83	3.89	14.44		
	Adult	0.25	1.29	1.47	0.66	13.41		
Dand	Harmful	0.42	1.65	1.83	0.21	15.20		
Dalu	Privacy	0.65	1.81	2.69	0.44	16.60		
	Illegal	0.40	1.78	2.38	0.12	12.85		
Bing Chat	Adult	0.41	1.21	1.31	0.41	10.21		
	Harmful	0.47	1.32	1.45	0.32	11.42		
	Privacy	0.76	1.57	1.83	0.23	18.40		
	Illegal	0.88	1.23	1.51	0.12	14.48		

TABLE 6.5: Performance comparison of each baseline in generating jailbreak prompts in terms of query success rate. Values in bold indicate the best-performing metrics in their respective categories.

where S is the number of successful jailbreak queries and T is the total number of jailbreak queries. This metric helps in understanding how often our strategies can trick the model into generating prohibited content.

Further, to evaluate the quality of the generated jailbreak prompts, we define the jailbreak prompt success rate as below:

$$J = \frac{G}{P}$$

Where G is the number of generated jailbreak prompts with at least one successful query and P is the total number of generated jailbreak prompts. The jailbreak prompt success rate illustrates the proportion of successful generated prompts, thus providing a measure of the prompts' effectiveness.

6.7.2 Jailbreak Capability (RQ3)

In our evaluation of MASTERKEY, we utilize GPT-3.5, GPT-4, and Vicuna as benchmarks. Each model receives 85 unique jailbreak prompts. They generate 10 distinct variants per prompt. We test these rewritten prompts with 20 prohibited questions. This results a total number of 272,000 queries for the evaluation. We present the average query success rate in Table 6.5.

Table 6.5 demonstrates that MASTERKEY significantly outperforms other models in creating jailbreak prompts, using the query success rate as a metric. More specifically, MASTERKEY achieves an average success rate of 14.51% and 13.63% when measured against Bard and Bing Chat, respectively. To the best of our knowledge, this marks the first successful jailbreak for the two services. GPT-4 secures the second spot in our evaluation. As one of the top publicly available commercial LLMs, it showcases a notable performance. However, it falls short in contesting with MASTERKEY in the specialized field of generating jailbreak prompts, which indicates a potential area for enhancement in its fine-tuning GPT-3.5 performs with slightly less efficiency, primarily due to its process. smaller parameter size compared to GPT-4. This factor consequently restricts its ability to generate successful jailbreak prompts. Finally, Vicuna lags behind all other models, marking the least efficient performance. This can primarily be attributed to its sensitivity to disruptions from original prompts. For example, consider a prompt containing the sentence: "Now generate a sample/possible output of print(function("[INSERT_PROMPT_HERE]"))". In response to this, Vicuna merely prints "[INSERT_PROMPT_HERE]" rather than rewriting it. This limitation highlights a significant shortcoming in Vicuna's understanding and handling of the task of generating jailbreak prompts. The above findings underscore the critical role of domain-specific knowledge in the generation of successful jailbreak prompts.

We assess the impact of each jailbreak prompt generated by MASTERKEY. We do this by examining the jailbreak success rate for each prompt. This analysis gives us a glimpse into their individual performance. Our results indicate that the most effective jailbreak prompts account for 38.2% and 42.3% of successful jailbreaks for GPT-3.5 and GPT-4, respectively. On the other hand, for Bard and Bing Chat, only 11.2% and 12.5% of top prompts lead to successful jailbreak queries.



FIGURE 6.6: Average Query Success Rate Across LLM Chatbots for MAS-TERKEY, MASTERKEY-NO-FINETUNE, and MASTERKEY-NO-REWARD.

These findings hint that a handful of highly effective prompts significantly drive the overall jailbreak success rate. This observation is especially true for Bard and Bing Chat. We propose that this discrepancy is due to the unique jailbreak prevention mechanisms of Bard and Bing Chat. These mechanisms allow only a very restricted set of carefully crafted jailbreak prompts to bypass their defenses. This highlights the need for further research into crafting highly effective prompts.

6.7.3 Ablation Study (RQ4)

We carry out an ablation study to gauge each component's contribution to MAS-TERKEY's effectiveness. We create two variants for this study: MASTERKEY-NO-FINETUNE, and MASTERKEY-NO-REWARD. They are fine-tuned but lack reward-ranked fine-tuning. For the ablation study, each variant processes 85 jailbreak prompts. They generate 10 jailbreak variants for each. This approach helps us single out the effect of the components in question. We repeat the experiment five times. Then we assess the performances to gauge the omitted impact of each component. Figure 6.6 presents the result in terms of average query success rate.

From Figure 6.6, it is evident that MASTERKEY delivers superior performance compared to the other variants. Its success is attributable to its comprehensive methodology that involves both fine-tuning and reward-ranked feedback. This combination optimizes the model's understanding of context, leading to improved performance. MASTERKEY-NO-REWARD, which secures the second position in the study, brings into focus the significant role of reward-ranked feedback in enhancing a model's performance. Without this component, the model's effectiveness diminishes, as indicated by its lower ranking. Lastly, MASTERKEY-NO-FINETUNE, the variant that performs the least effectively in our study, underscores the necessity of fine-tuning in model optimization. Without the fine-tuning process, the model's performance noticeably deteriorates, emphasizing the importance of this step in the training process of large language models.

In conclusion, both fine-tuning and reward-ranked feedback are indispensable in optimizing the ability of large language models to generate jailbreak prompts. Omitting either of these components leads to a significant decrease in effectiveness, undermining the utility of MASTERKEY.

6.7.4 Cross-language Compatibility (RQ5)

To study the language compatibility of the MASTERKEY generated jailbreak prompts, we conduct supplementary evaluation on Ernie, which is developed by the leading Chinese LLM service provider Baidu [294]. This model supports simplified Chinese inputs with a limit on the token length of 600. To generate the input for Ernie, we translate the jailbreak prompts and questions into simplified Chinese and feed them to Ernie. Note that we only conducted a small experiment due to the rate limit and account suspension risks upon repeated jailbreak attempts. We finally sampled 20 jailbreak prompts from the experiment data with the 20 malicious questions.

Our experimental results indicate that the translated jailbreak prompts effectively compromise the Ernie chatbot. Specifically, the generated jailbreak prompts achieve an average success rate of 6.45% across the four policy violation categories. This implies that 1) the jailbreak prompts can work cross-language and 2) the model-specific training process can generate cross-model jailbreak prompts. These findings indicate the need for further research to enhance the resilience of various LLMs against such jailbreak prompts, thereby ensuring their safe and effective application across diverse languages. They also highlight the importance of developing robust detection and prevention mechanisms to ensure the integrity and security.
6.8 Mitigation Recommendation

To enhance jailbreak defenses, a comprehensive strategy is required. we propose several potential countermeasures that could bolster the robustness of LLM chatbots. Primarily, the ethical and policy-based alignments of LLMs must be solidified. This reinforcement increases their innate resistance to executing harmful instructions. Although the specific defensive mechanisms currently used are not disclosed, we suggest that supervised training [295] could provide a feasible strategy to strengthen such alignments. In addition, it is crucial to refine moderation systems and rigorously test them against potential threats. This includes the specific recommendation of incorporating input sanitization into system defenses, which could prove a valuable tactic. Moreover, techniques such as contextual analysis [296] could be integrated to effectively counter the encoding strategies that aim to exploit existing keyword-based defenses. Finally, it is essential to develop a comprehensive understanding of the model's vulnerabilities. This can be achieved through thorough stress testing, which provides critical insights to reinforce defenses. By automating this process, we ensure efficient and extensive coverage of potential weaknesses, ultimately strengthening the security of LLMs.

6.9 Related Work

6.9.1 Prompt Engineering and Jailbreaks in LLMs

Prompt engineering [297–300] plays an instrumental role in the development of language models, providing a means to significantly augment a model's ability to undertake tasks it has not been directly trained for. As underscored by recent studies [301–303], well-devised prompts can effectively optimize the performance of language models.

However, this powerful tool can also be used maliciously, introducing serious risks and threats. Recent studies [274–277, 304, 305] have drawn attention to the rise of "jailbreak prompts", ingeniously crafted to circumvent the restrictions placed on language models and coax them into performing tasks beyond their intended scope. Most of the traditional strategies fall into the category of fuzzing. For instance, [306] proposes the usage of greedy coordinate gradient (GCG) to fuzz the LLM and revise the prompt suffix following the gradient feedback from the model loss. [307] takes the blackbox approach and only relies on the LLM response as feedback to revise the prompt.

Unlike previous studies, which primarily underscore the possibility of such attacks and relies on fuzzing feedback, our research delves deeper. We not only devise and execute jailbreak techniques, but also abstract the jailbreak patterns into another LLM to complete the loop in an automated manner.

6.9.2 LLM Security and Relevant Attacks

Hallucination in LLMs. The phenomenon highlights a significant issue associated with the machine learning domain. Owing to the vast crawled datasets on which these models are trained, they can potentially generate contentious or biased content. These datasets, while large, may include misleading or harmful information, resulting in models that can perpetuate hate speech, stereotypes, or misinformation [308–312]. To mitigate this issue, mechanisms like RLHF (Reinforcement Learning from Human Feedback) [276, 313] have been introduced. These measures aim to guide the model during training, using human feedback to enhance the robustness and reliability of the LLM outputs, thereby reducing the chance of generating harmful or biased text. However, despite these precautionary steps, there remains a non-negligible risk from targeted attacks where such undesireable output are elicited, such as jailbreaks [274, 275] and prompt injections [314, 315]. These complexities underline the persistent need for robust mitigation strategies and ongoing research into the ethical and safety aspects of LLMs.

Prompt Injection. This type of attacks [314–317] constitutes a form of manipulation that hijacks the original prompt of an LLM, steering it towards malicious directives. The consequences can range from generation of misleading advice to unauthorized disclosure of sensitive data. LLM Backdoor [318–320] and model hijacking [321, 322] attacks can also be broadly categorized under this type of assault. Perez et al. [315] highlighted the susceptibility of GPT-3 and its dependent applications to prompt injection attacks, showing how they can reveal the application's underlying prompts.

Distinguishing our work, we conduct a systematic exploration of the strategies and prompt patterns that can initiate these attacks across a broader spectrum of realworld applications. In comparison, prompt injection attacks focus on altering the model's inputs with malicious prompts, causing it to generate misleading or harmful outputs, essentially hijacking the model's task. Conversely, jailbreak attacks aim to bypass restrictions imposed by service providers, enabling the model to produce outputs usually prevented.

6.9.3 Vulnerability Analysis for Traditional Web Applications

LLM chatbots are an emerging category of web applications. Various techniques have been proposed for detecting vulnerabilities or other flaws in web applications [263, 264, 323–330]. On the one hand, these techniques can be applied to detecting traditional types of vulnerabilities (e.g., SQL injection, XSS) in the web components of LLM chatbots. On the other hand, these techniques can inspire new approaches for detecting the new types of vulnerabilities (e.g., prompt injection, jailbreak) specific to LLM. We propose the time-based analysis of MASTERKEY that draws inspiration from time-based SQL injection attacks. In conclusion, combining traditional and LLM-centric approaches can establish a more comprehensive security strategy for LLM chatbots.

6.10 Conclusion

This study encompasses a rigorous evaluation of mainstream LLM chatbot services, revealing their significant susceptibility to jailbreak attacks. We introduce MAS-TERKEY, a novel framework to heat the arms race between jailbreak attacks and defenses. MASTERKEY first employs time-based analysis to reverse-engineer defenses, providing novel insights into the protection mechanisms employed by LLM chatbots. Furthermore, it introduces an automated method to generate universal jailbreak prompts, achieving an average success rate of 21.58% among mainstream chatbot services. These findings, together with our recommendations, are responsibly reported to the providers, and contribute to the development of more robust safeguards against the potential misuse of LLMs.

Chapter 7

Conclusion and Future Work

In this chapter, we first give a summary of the work conducted in this thesis and then discuss some future research directions based on our current results.

7.1 Conclusion

The challenge of security testing in human-interactive systems, particularly as they grow in complexity, forms the crux of this research.

This thesis embarked on a comprehensive journey, beginning with the exploration of Byzantine threats in Multi-Robot Systems (MRSs) using advanced fuzzing techniques. This initial phase involved a meticulous analysis of MRSs, identifying potential vulnerabilities and demonstrating the need for more sophisticated security approaches in these collaborative systems. The research then progressed to a detailed examination of robot operating systems, with a specific focus on ROS2. Here, the application of model checking revealed several critical security issues, underscoring the importance of thorough validation in such systems. The study not only identified vulnerabilities but also proposed robust defense mechanisms, offering practical solutions to enhance the security of robotic operating systems.

The exploration of web applications within this thesis marked a significant advancement in security testing approaches. By integrating human expertise directly into the testing process, the research recognized the limitations of purely automated methods in complex web environments. This innovative strategy leveraged the nuanced understanding and adaptability of human testers, enabling the identification of subtle vulnerabilities and irregular patterns that automated systems might overlook. The methodology demonstrated the potential for a more holistic approach to web application security, blending technological tools with human insight to enhance overall system robustness.

Turning to the realm of Large Language Models, the research ventured into relatively uncharted territory by addressing the security concerns of LLM-based chatbots. The study's groundbreaking methodology, which focused on understanding and overcoming the inherent 'black-box' nature of these AI systems, provided a novel perspective on security testing. The research unveiled key vulnerabilities and developed effective strategies to bypass the sophisticated defense mechanisms of these chatbots, paving the way for more secure and reliable AI-driven communication tools.

In conclusion, this thesis presents a comprehensive and innovative approach to security testing in human-interactive systems. It spans various domains, from robotics to AI-driven chatbots, addressing unique challenges and offering practical solutions. The research contributes significantly to the field, proposing strategies that blend traditional and novel testing methods, and setting a new benchmark for security in increasingly complex and interactive technological environments.

7.2 Future Work

Following my dissertation research, there is still a lot more to be explored in the future.

• Defense of Byzantine threats in multi-robot systems. The exploration of Byzantine threats in MRSs has opened avenues for future research in developing more sophisticated defense mechanisms. Future studies could focus on creating adaptive and predictive models that can preemptively identify and mitigate these threats in real-time, enhancing the resilience of MRSs against advanced attacks. Another prospective area involves exploring the integration of AI and machine learning techniques in detecting and responding to Byzantine behaviors. This could involve training models on vast datasets to recognize patterns indicative of such threats, thereby improving the overall security posture of MRSs.

- Security verification of robotic systems. Future work in this domain could delve into the development of more comprehensive and efficient verification tools that can handle the increasing complexity of modern robotic systems. Emphasis could be placed on creating scalable verification frameworks that can be easily adapted to different types of robotic systems. There is also a need for research into real-time verification methods that can provide ongoing security assurance in dynamic and unpredictable operational environments typical of robotic systems.
- Automated white-box API vulnerability identification. The existing solution of black-box based strategy in RESTful API testing is still not efficient enough. Advancements in automated white-box testing for APIs could focus on leveraging more advanced code analysis techniques, potentially integrating AI to predict and identify vulnerabilities based on historical data and patterns. Research could also explore the integration of continuous testing methods into the development lifecycle of APIs, ensuring that security is a constant consideration and that vulnerabilities are identified and addressed promptly.
- Large Language Model Security. The burgeoning field of LLM security necessitates the development of dynamic defense mechanisms capable of adapting to attacker strategies. Future research should focus on AI-driven defense systems that evolve with emerging attack patterns. With the rapid advancement of large language models, emerging issues such as fairness [331] and AI privacy [332] need to be comprehensively addressed. This includes understanding biases in LLM outputs and safeguarding user data processed by these models. Ethical implications and privacy concerns in LLM interactions are paramount, requiring a balanced approach. Future work should aim at developing robust ethical guidelines and privacy frameworks. This includes ensuring LLMs' interactions with users respect privacy norms and ethical standards, especially when handling sensitive information.
- Large Language Model For Security. The potential of LLMs in enhancing security tasks presents an exciting avenue for research. Future studies could explore the use of LLMs in areas like penetration testing, where they

can simulate complex attack scenarios, or in fuzzing, to generate sophisticated test cases. Additionally, the integration of LLMs in model checking could provide more intuitive and effective ways to identify system vulnerabilities. The application of LLMs in security tasks opens up possibilities for more intelligent and automated security solutions. Research should focus on harnessing the predictive and analytical capabilities of LLMs to enhance overall system security, thereby making security processes more efficient and comprehensive.

List of Publications

- <u>Gelei Deng</u>*, Yi Liu*, Yuekang Li, Kailong Wang, Ying Zhang, Zefeng Li, Haoyu Wang, Tianwei Zhang, Yang Liu. MASTERKEY: Automated Jailbreak Across Multiple Large Language Model Chatbots. In *The Network* and Distributed System Security Symposium (NDSS) 2024.
- <u>Gelei Deng</u>, Zhiyi Zhang, Yuekang Li, Yi Liu, Tianwei Zhang, Yang Liu, Guo Yu, Dongjin Wang. NAUTILUS: Automated RESTful API Vulnerability Detection. In 32nd USENIX Security Symposium (USENIX Security 23).
- <u>Gelei Deng</u>, Guowen Xu, Yuan Zhou, Tianwei Zhang, Yang Liu. On the (In) Security of Secure ROS2. In *Proceedings of the 2022 ACM SIGSAC Conference on Computer and Communications Security (ACM CCS 2022).*
- <u>Gelei Deng</u>, Yuan Zhou, Yuan Xu, Tianwei Zhang, Yang Liu, An investigation of byzantine threats in multi-robot systems, in *Proceedings of the 24th International Symposium on Research in Attacks, Intrusions and Defenses* (*RAID 21*).
- <u>Gelei Deng</u>, Yi Liu, Víctor Mayoral-Vilches, Peng Liu, Yuekang Li, Yuan Xu, Tianwei Zhang, Yang Liu, Martin Pinzger, Stefan Rass. Pentestgpt: An llm-empowered automatic penetration testing tool.
- Yi Liu^{*}, <u>Gelei Deng^{*}</u>, Zhengzi Xu, Yuekang Li, Yaowen Zheng, Ying Zhang, Lida Zhao, Tianwei Zhang, Yang Liu. Jailbreaking chatgpt via prompt engineering: An empirical study.
- Yi Liu^{*}, <u>Gelei Deng^{*}</u>, Yuekang Li, Kailong Wang, Tianwei Zhang, Yepang Liu, Haoyu Wang, Yan Zheng, Yang Liu. Prompt Injection attack against LLM-integrated Applications
- Yuan Xu, <u>Gelei Deng</u>, Tianwei Zhang, Han Qiu, Yungang Bao. Novel denial-of-service attacks against cloud-based multi-robot systems. In *Information Science*
- Yi Liu, Yuekang Li, <u>Gelei Deng</u>, Felix Juefei-Xu, Yao Du, Cen Zhang, Chengwei Liu, Yeting Li, Lei Ma, Yang Liu. ASTER: Automatic speech

recognition system accessibility testing for stutterers. In 2023 38th IEEE/ACM International Conference on Automated Software Engineering (ASE)

- Yuan Zhou, Hesuan Hu, <u>Gelei Deng</u>, Kun Cheng, Shang-Wei Lin, Yang Liu, Zuohua Ding. Distributed Motion Control for Multiple Mobile Robots Using Discrete-Event Systems and Model Predictive Control. In *IEEE Transactions* on Systems, Man, and Cybernetics: Systems.
- Yi Liu, Yuekang Li, <u>Gelei Deng</u>, Yang Liu, Ruiyuan Wan, Runchao Wu, Dandan Ji, Shiheng Xu, Minli Bao. Morest: model-based RESTful API testing with execution feedback. In *Proceedings of the 44th International Conference on Software Engineering*
- Xingshuo Han*, Kangqiao Zhao*, <u>Gelei Deng</u>, Yuan Xu, Tianwei Zhang.
 VisionGuard: A Unified Defense Framework against Physical Adversarial Attacks in Autonomous Driving Perception.
- Xingshuo Han, Yuan Zhou, <u>Gelei Deng</u>, Yuan Xu, Han Qiu, Tianwei Zhang. DMAFuzz: Dynamic Physical-world Attacks to Decision-making Module in Autonomous Driving Systems.
- Yuan Xu, Xingshuo Han, <u>Gelei Deng</u>, Yang Liu, Jiwei Li, Tianwei Zhang. SoK: Rethinking Sensor Spoofing Attacks against Robotic Vehicles from a Systematic View. in *IEEE European Symposium on Security and Privacy*, 2023.
- Guowen Xu, Xingshuo Han, <u>Gelei Deng</u>, Tianwei Zhang, Shengmin Xu, Anjia Yang, Hongwei Li. VerifyML: Obliviously Checking Model Fairness Resilient to Malicious Model Holder. in *IEEE Transaction Dependable Secure Computing*, 2023.
- Yisroel Mirsky, Ambra Demontis, Jaidip Kotak, Ram Shankar, <u>Gelei Deng</u>, Liu Yang, Xiangyu Zhang, Maura Pintor, Wenke Lee, Yuval Elovici, Battista Biggio. The threat of offensive ai to organizations. In *Computers & Security*.

Bibliography

- Soonjae Pyo, Jaeyong Lee, Kyubin Bae, Sangjun Sim, and Jongbaeg Kim. Recent progress in flexible tactile sensors for human-interactive systems: from sensors to advanced applications. *Advanced Materials*, 33(47):2005902, 2021.
 3
- [2] Jef Raskin. The humane interface: new directions for designing interactive systems. Addison-Wesley Professional, 2000. 11
- [3] Matthew L Bolton and Ellen J Bass. A method for the formal verification of human-interactive systems. In *Proceedings of the Human Factors* and Ergonomics Society Annual Meeting, volume 53, pages 764–768. SAGE Publications Sage CA: Los Angeles, CA, 2009. 3, 11
- [4] Open source robot operating system. http://www.ros.org/, 2019. 3, 12
- [5] ROS2. ROS 2 Foxy Fitzroy. https://docs.ros.org/en/foxy/Releases/ Release-Foxy-Fitzroy.html, 2020. 13, 74
- [6] Alessandro Farinelli, Luca Iocchi, and Daniele Nardi. Distributed on-line dynamic task assignment for multi-robot patrolling. Autonomous Robots, 41 (6):1321–1345, 2016. 3, 22
- [7] Filippo Ricca and Paolo Tonella. Analysis and testing of web applications. In Proceedings of the 23rd International Conference on Software Engineering. ICSE 2001, pages 25–34. IEEE, 2001. 3, 11
- [8] Yann LeCun, Yoshua Bengio, and Geoffrey Hinton. Deep learning. nature, 521(7553):436-444, 2015. 3
- [9] OpenAI. Introducing ChatGPT. https://openai.com/blog/chatgpt. 3, 11, 130
- [10] Stuart J Russell and Peter Norvig. Artificial intelligence a modern approach. London, 2010. 3
- [11] Lianmin Zheng, Wei-Lin Chiang, Ying Sheng, Siyuan Zhuang, Zhanghao Wu, Yonghao Zhuang, Zi Lin, Zhuohan Li, Dacheng Li, Eric. P Xing, Hao Zhang, Joseph E. Gonzalez, and Ion Stoica. Judging llm-as-a-judge with mt-bench and chatbot arena, 2023. 3, 155

- [12] Michael Sutton, Adam Greene, and Pedram Amini. Fuzzing: Brute Force Vulnerability Discovery. Addison-Wesley Professional, 2007. ISBN 0321446119. 3, 4, 15, 27
- [13] Hyungsub Kim, Muslum Ozgur Ozmen, Antonio Bianchi, Z Berkay Celik, and Dongyan Xu. Pgfuzz: Policy-guided fuzzing for robotic vehicles. In Network and Distributed System Security Symposium, 2021. 15, 55
- [14] Fute Shang, Buhong Wang, Tengyao Li, Jiwei Tian, and Kunrui Cao. Cpfuzz: Combining fuzzing and falsification of cyber-physical systems. *IEEE Access*, 8:166951–166962, 2020. 3, 4, 55
- [15] Taegyu Kim, Chung Hwan Kim, Junghwan Rhee, Fan Fei, Zhan Tu, Gregory Walkup, Xiangyu Zhang, Xinyan Deng, and Dongyan Xu. Rvfuzzer: Finding input validation bugs in robotic vehicles through control-guided testing. In *Proceedings of the 28th USENIX Conference on Security Symposium*, SEC'19, page 425–442, USA, 2019. USENIX Association. ISBN 9781939133069. 4, 15, 18, 55
- [16] Edmund M Clarke. Model checking. In Foundations of Software Technology and Theoretical Computer Science: 17th Conference Kharagpur, India, December 18–20, 1997 Proceedings 17, pages 54–56. Springer, 1997. 5
- [17] Gelei Deng, Guowen Xu, Yuan Zhou, Tianwei Zhang, and Yang Liu. On the (in) security of secure ros2. In Proceedings of the 2022 ACM SIGSAC Conference on Computer and Communications Security, pages 739–753, 2022.
- [18] Martin Davis, George Logemann, and Donald Loveland. A machine program for theorem-proving. Communications of the ACM, 5(7):394–397, 1962. 5
- [19] Donald W Loveland. Automated theorem proving: A logical basis. Elsevier, 2016. 5
- [20] Rae A Earnshaw. Virtual reality systems. Academic press, 2014. 11, 12
- [21] Peter A Hancock, Deborah R Billings, Kristin E Schaefer, Jessie YC Chen, Ewart J De Visser, and Raja Parasuraman. A meta-analysis of factors affecting trust in human-robot interaction. *Human factors*, 53(5):517–527, 2011. 11
- [22] Thomas B Sheridan. Human-robot interaction: status and challenges. Human factors, 58(4):525–532, 2016. 11
- [23] OWASP Juice-Shop Project. https://owasp.org/ www-project-juice-shop/, 2022. 11, 116, 117
- [24] Google. Bard. https://bard.google.com/?hl=en. 11, 130, 137

- [25] Jay Peters. The Bing AI bot has been secretly running GPT-4. https://www.theverge.com/2023/3/14/23639928/ microsoft-bing-chatbot-ai-gpt-4-llm. 11, 130
- [26] Oliver Kreylos. Environment-independent vr development. In International Symposium on Visual Computing, pages 901–912. Springer, 2008. 12
- [27] Leif P Berg and Judy M Vance. Industry use of virtual reality in product design and manufacturing: a survey. *Virtual reality*, 21:1–17, 2017. 12
- [28] Varun Kohli, Utkarsh Tripathi, Vinay Chamola, Bijay Kumar Rout, and Salil S Kanhere. A review on virtual reality and augmented reality use-cases of brain computer interface based applications for smart cities. *Microproces*sors and Microsystems, 88:104392, 2022. 12
- [29] Open Source Robotics Foundation. Open Robotics. https://www. openrobotics.org/, 2021. 12, 59, 61, 79
- [30] Dji onboard sdk. https://developer.dji.com/onboard-sdk/, 2020. 13, 19, 21
- [31] Ros pr2 package. http://wiki.ros.org/Robots/PR2/, 2020. 13, 19, 21
- [32] Fleets of drones could aid searches for lost hikers. http://news.mit.edu/ 2018/fleets-drones-help-searches-lost-hikers-1102/, 2018. 13
- [33] Mit creates control algorithm for drone swarms. https://techcrunch.com/ 2016/04/22/mit-creates-a-control-algorithm-for-drone-swarms/, 2016.
- [34] Who will control the swarm? https://platformlab.stanford.edu/news. php, 2016. 13
- [35] Hyungsub Kim, Muslum Ozgur Ozmen, Antonio Bianchi, Z. Berkay Celik, and Dongyan Xu. PGFUZZ: policy-guided fuzzing for robotic vehicles. In 28th Annual Network and Distributed System Security Symposium, NDSS 2021, virtually, February 21-25, 2021. The Internet Society, 2021. 15, 71, 73
- [36] Ros mrs application malicious communication identification script, 2021. URL https://github.com/GeleiDeng/RAID_2021_MRS_Fuzzing. 16
- [37] Pratap Tokekar, Joshua Vander Hook, David J. Mulla, and Volkan Isler. Sensor planning for a symbiotic uav and ugv system for precision agriculture. *IEEE Transactions on Robotics*, 32(6):1498–1511, 2016. 18
- [38] Mohammadreza Davoodi, Saba Faryadi, and Javad Mohammadpour Velni. A graph theoretic-based approach for deploying heterogeneous multi-agent systems with application in precision agriculture. *IEEE Transactions on Robotics*, 32(6):1498–1511, 2016.

- [39] Mohammadreza Davoodi, Javad Mohammadpour Velni, and Changying Li. Coverage control with multiple ground robots for precision agriculture. *Mechanical Engineering*, 140(06):S4–S8, 2018.
- [40] Patrick Nolan, Derek A. Paley, and Kenneth Kroeger. Multi-uas path planning for non-uniform data collection in precision agriculture. In *IEEE Aerospace Conference*, pages 1–12, 2017. 18
- [41] Michael E. Kepler and Daniel J. Stilwell. An approach to reduce communication for multi-agent mapping applications. In *IEEE/RJS International Conference on Intelligent RObots and Systems (IROS)*, pages 4814–4820, 2020. 18
- [42] Mahmoud Tavakoli, Gonçalo Cabrita, Ricardo Faria, Lino Marques, and Anibal T. de Almeida. Cooperative multi-agent mapping of three-dimensional structures for pipeline inspection applications. *International Journal of Robotics Research*, 31(12):1489–1503, 2012.
- [43] Wenbiao Han and Mohsen A. Jafari. Controller synthesis via mapping task sequence to petri nets in multi-agent collaboration applications. In *IEEE International Conference on Robotics and Automation (ICRA)*, pages 4312–4317, 2003.
- [44] Sebastian Thrun, Wolfram Burgard, and Dieter Fox. A real-time algorithm for mobile robot mapping with applications to multi-robot and 3d mapping. In *IEEE International Conference on Robotics and Automation (ICRA)*, pages 321–328. IEEE, 2000.
- [45] Franciszek Seredynski. Competitive coevolutionary multi-agent systems: The application to mapping and scheduling problems. Journal of Parallel and Distributed Computing, 47(1):39–57, 1997. 18
- [46] Tenda Okimoto, Tony Ribeiro, Damien Bouchabou, and Katsumi Inoue. Mission oriented robust multi-team formation and its application to robot rescue simulation. In *International Joint Conference on Artificial Intelligence (IJ-CAI)*, pages 454–460, 2016. 18
- [47] Yugang Liu and Goldie Nejat. Multirobot cooperative learning for semiautonomous control in urban search and rescue applications. *Journal of Field Robotics*, 33(4):512–536, 2016.
- [48] Karl Muecke and Brian Powell. A distributed, heterogeneous, targetoptimized operating system for a multi-robot search and rescue application. *IEAAIE*, (2):266–275, 2011. 18
- [49] CG Leela Krishna and Robin R Murphy. A review on cybersecurity vulnerabilities for unmanned aerial vehicles. In *IEEE International Symposium on Safety, Security and Rescue Robotics*, pages 194–199, 2017. 18

- [50] Vishal Dey, Vikramkumar Pudi, Anupam Chattopadhyay, and Yuval Elovici. Security vulnerabilities of unmanned aerial vehicles and countermeasures: An experimental study. In *International Conference on VLSI Design and International Conference on Embedded Systems*, pages 398 – 403, Pune, India, 2018.
- [51] Drew Davidson, Hao Wu, Rob Jellinek, Vikas Singh, and Thomas Ristenpart. Controlling uavs with sensor input spoofing attacks. In USENIX Workshop on Offensive Technologies, pages 221–231, 2016. 55
- [52] Yunmok Son, Hocheol Shin, Dongkwan Kim, Young-Seok Park, Juhwan Noh, Kibum Choi, Jungwoo Choi, and Yongdae Kim. Rocking drones with intentional sound noise on gyroscopic sensors. In USENIX Security Symposium (USENIX Security 15), pages 881–896, 2015. 18, 55
- [53] Fan Fei, Zhan Tu, Ruikun Yu, Taegyu Kim, Xiangyu Zhang, Dongyan Xu, and Xinyan Deng. Cross-layer retrofitting of uavs against cyber-physical attacks. In *IEEE International Conference on Robotics and Automation*, pages 550–557, 2018. 18
- [54] Hongjun Choi, Wen-Chuan Lee, Yousra Aafer, Fan Fei, Zhan Tu, Xiangyu Zhang, Dongyan Xu, and Xinyan Xinyan. Detecting attacks against robotic vehicles: A control invariant approach. In ACM Conference on Computer and Communications Security, pages 801–816, 2018. 18
- [55] Davide Quarta, Marcello Pogliani, Mario Polino, Federico Maggi, Andrea Maria Zanchettin, and Stefano Zanero. An experimental security analysis of an industrial robot controller. In *IEEE Symposium on Security and Privacy*, pages 268–286, 2017. 18, 55
- [56] Cesar Cerrudo and Lucas Apa. Hacking robots before skynet. Technical report, IOActive, Inc, 2017. 18, 24, 55
- [57] Bernhard Dieber, Benjamin Breiling, Sebastian Taurer, Severin Kacianka, Stefan Rass, and Peter Schartner. Security for the robot operating system. *Robotics and Autonomous Systems*, 98:192–203, 2017. 18, 24, 55
- [58] Wikipedia. Byzantine fault Wikipedia, the free encyclopedia, 2021. URL https://en.wikipedia.org/wiki/Byzantine_fault. 18
- [59] Global Times. Mainframe malfunction causes dozens of drones to crash into building in sw china. https://www.globaltimes.cn/page/202101/ 1214165.shtml, 2021. 18
- [60] Zohir Bouzid, Maria Gradinariu Potop-Butucaru, and Sébastien Tixeuil. Byzantine convergence in robot networks: The price of asynchrony. In Tarek Abdelzaher, Michel Raynal, and Nicola Santoro, editors, *Principles of Distributed Systems*, pages 54–70, Berlin, Heidelberg, 2009. Springer Berlin Heidelberg. ISBN 978-3-642-10877-8. 18, 54

- [61] Cédric Auger, Zohir Bouzid, Pierre Courtieu, Sébastien Tixeuil, and Xavier Urbain. Certified impossibility results for byzantine-tolerant mobile robots. In Teruo Higashino, Yoshiaki Katayama, Toshimitsu Masuzawa, Maria Potop-Butucaru, and Masafumi Yamashita, editors, *Stabilization, Safety,* and Security of Distributed Systems, pages 178–190, Cham, 2013. Springer International Publishing. 54
- [62] Anisur Rahaman Molla, Kaushik Mondal, and William K. Moses Jr au2. Efficient dispersion on an anonymous ring in the presence of weak byzantine robots, 2020. 54
- [63] Volker Strobel, Eduardo Castelló Ferrer, and Marco Dorigo. Managing byzantine robots via blockchain technology in a swarm robotics collective decision making scenario. In Proceedings of the 17th International Conference on Autonomous Agents and MultiAgent Systems, AAMAS '18, page 541–549, Richland, SC, 2018. International Foundation for Autonomous Agents and Multiagent Systems. 18, 54
- [64] Max Ammann, Lucca Hirschi, and Steve Kremer. Dy fuzzing: formal dolevyao models meet cryptographic protocol fuzz testing. Cryptology ePrint Archive, 2023. 18
- [65] D. Dolev and A. Yao. On the security of public key protocols. *IEEE Transactions on Information Theory*, 29(2):198–208, 1983. doi: 10.1109/TIT.1983. 1056650.
- [66] Open source robot operating system, 2020. URL https://index.ros.org/. 19, 31
- [67] Ros abb package. http://wiki.ros.org/abb/, 2020. 19
- [68] Isai Roman-Ballesteros and Carlos F Pfeiffer. A framework for cooperative multi-robot surveillance tasks. In *Electronics, Robotics and Automotive Mechanics Conference (CERMA'06)*, volume 2, pages 163–170, 2006. 19, 24, 35
- [69] Joaquin López, Diego Pérez, Enrique Paz, and Alejandro Santana. Watchbot: A building maintenance and surveillance system based on autonomous robots. *Robotics and Autonomous Systems*, 61(12):1559–1571, 2013.
- [70] Kelin Jose and Dilip Kumar Pratihar. Task allocation and collision-free path planning of centralized multi-robots system for industrial plant inspection using heuristic methods. *Robotics and Autonomous Systems*, 80:34–42, 2016. 35, 91
- [71] Robert Reid, Andrew Cann, Calum Meiklejohn, Liam Poli, Adrian Boeing, and Thomas Braunl. Cooperative multi-robot navigation, exploration, mapping and object detection with ros. In 2013 IEEE Intelligent Vehicles Symposium (IV), pages 1083–1088, 2013. 24, 36, 91

- [72] Stuart Golodetz, Tommaso Cavallari, Nicholas A Lord, Victor A Prisacariu, David W Murray, and Philip HS Torr. Collaborative large-scale dense 3d reconstruction with online inter-agent pose optimisation. *IEEE Transactions* on Visualization and Computer Graphics, 24(11):2895–2905, 2018. 36
- [73] Juan C. Elizondo-Leal, Gabriel Ramírez-Torres, and Gregorio Toscano Pulido. Multi-robot exploration and mapping using self biddings and stop signals. In Alexander Gelbukh and Eduardo F. Morales, editors, *MICAI 2008: Advances in Artificial Intelligence*, pages 615–625, Berlin, Heidelberg, 2008. Springer Berlin Heidelberg. 24, 32, 37
- [74] Marc Berhault, He Huang, Pinar Keskinocak, Sven Koenig, Wedad Elmaghraby, Paul Griffin, and Anton Kleywegt. Robot exploration with combinatorial auctions. In *Proceedings 2003 IEEE/RSJ International Conference* on Intelligent Robots and Systems (IROS 2003), volume 2, pages 1957–1962. IEEE, 2003.
- [75] Craig Tovey, Michail G Lagoudakis, Sonal Jain, and Sven Koenig. The generation of bidding rules for auction-based robot coordination. In *Multi-Robot Systems. From Swarms to Intelligent Automata Volume III*, pages 3–14. Springer, 2005.
- [76] Flavio Cabrera-Mora and Jizhong Xiao. A flooding algorithm for multirobot exploration. *IEEE Transactions on Systems, Man, and Cybernetics, Part B* (Cybernetics), 42(3):850–863, 2012.
- [77] Abdelfetah Hentout, Abderraouf Maoudj, Nesrine Kaid-Youcef, Djamila Hebib, and Brahim Bouzouia. Distributed multi-agent bidding-based approach for the collaborative mapping of unknown indoor environments by a homogeneous mobile robot team. *Journal of Intelligent Systems*, 29(1): 84–99, 2020. 37
- [78] Wolfram Burgard, Mark Moors, Cyrill Stachniss, and Frank E Schneider. Coordinated multi-robot exploration. *IEEE Trans. Robot.*, 21(3):376–386, 2005. 24, 36, 38
- [79] Weihua Sheng, Qingyan Yang, Jindong Tan, and Ning Xi. Distributed multirobot coordination in area exploration. *Robotics and Autonomous Systems*, 54(12):945–955, 2006.
- [80] Jing Yuan, Yalou Huang, Tong Tao, and Fengchi Sun. A cooperative approach for multi-robot area exploration. In 2010 IEEE/RSJ International Conference on Intelligent Robots and Systems, pages 1390–1395. IEEE, 2010.
- [81] Andrew J Smith and Geoffrey A Hollinger. Distributed inference-based multirobot exploration. Autonomous Robots, 42(8):1651–1668, 2018. 38
- [82] Liu Yang. Swarm robot ros sim. https://github.com/yangliu28/swarm_ robot_ros_sim, 2020. 39

- [83] Hai Zhu, Jelle Juhl, Laura Ferranti, and Javier Alonso-Mora. Distributed multi-robot formation splitting and merging in dynamic environments. In 2019 International Conference on Robotics and Automation (ICRA), pages 9080–9086. IEEE, 2019.
- [84] Guannan Li, David St-Onge, Carlo Pinciroli, Andrea Gasparri, Emanuele Garone, and Giovanni Beltrame. Decentralized progressive shape formation with robot swarms. Autonomous Robots, 43(6):1505–1521, 2019.
- [85] Hanlin Wang and Michael Rubenstein. Shape formation in homogeneous swarms using local task swapping. *IEEE Transactions on Robotics*, 36(3): 597–612, 2020. 19, 39
- [86] N. Koenig and A. Howard. Design and use paradigms for gazebo, an opensource multi-robot simulator. In 2004 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), volume 3, pages 2149–2154 vol.3, 2004. 20, 35, 43, 80
- [87] Turtlebot official website, 2021. URL https://www.turtlebot.com/. 20, 44, 82
- [88] Ros index package list, 2020. URL https://index.ros.org/packages/. 21, 64
- [89] Veerajagadheswar Prabakaran, Mohan Rajesh Elara, Thejus Pathmakumar, and Shunsuke Nansai. Floor cleaning robot with reconfigurable mechanism. *Automation in Construction*, 91:155–165, 2018. ISSN 0926-5805. 22
- [90] N Vimal Kumar and C Selva Kumar. Development of collision free path planning algorithm for warehouse mobile robot. *Procedia computer science*, 133:456–463, 2018. 22
- [91] Valerio Digani, Lorenzo Sabattini, and Cristian Secchi. A probabilistic eulerian traffic model for the coordination of multiple agvs in automatic warehouses. *IEEE Robotics and Automation Letters*, 1(1):26–32, 2016. 22
- [92] Juan Jesús Roldán, Elena Peña-Tapia, Pablo Garcia-Aunon, Jaime Del Cerro, and Antonio Barrientos. Bringing adaptive and immersive interfaces to real-world multi-robot scenarios: Application to surveillance and intervention in infrastructures. *IEEE Access*, 7:86319–86335, 2019. 22
- [93] David Portugal, Luca Iocchi, and Alessandro Farinelli. A ROS-Based Framework for Simulation and Benchmarking of Multi-robot Patrolling Algorithms, pages 3–28. Springer International Publishing, Cham, 2019. 22
- [94] Dieter Fox, Jonathan Ko, Kurt Konolige, Benson Limketkai, Dirk Schulz, and Benjamin Stewart. Distributed multirobot exploration and mapping. *Proceedings of the IEEE*, 94(7):1325–1339, 2006. 23

- [95] Frank E Schneider, Dennis Wildermuth, and Hans-Ludwig Wolf. Elrob and eurathlon: Improving search rescue robotics through real-world robot competitions. In 2015 10th International Workshop on Robot Motion and Control (RoMoCo), pages 118–123, 2015. 23
- [96] Laura Barnes, MaryAnne Fields, and Kimon Valavanis. Unmanned ground vehicle swarm formation control using potential fields. In 2007 Mediterranean Conference on Control Automation, pages 1–8, 2007. 23
- [97] Hongliang Guo, Yan Meng, and Yaochu Jin. Swarm robot pattern formation using a morphogenetic multi-cellular based self-organizing algorithm. In 2011 IEEE International Conference on Robotics and Automation, pages 3205– 3210, 2011. 23
- [98] Nicholas DeMarinis, Stefanie Tellex, Vasileios P Kemerlis, George Konidaris, and Rodrigo Fonseca. Scanning the internet for ROS: A view of security in robotics research. In *International Conference on Robotics and Automation*, pages 8517–8521, Montreal, QC, Canada, 2019. IEEE. 24
- [99] Jarrod R. Mcclean and Charles Farrar. A preliminary cyber-physical security assessment of the robot operating system (ros). In *Proceedings of SPIE*, pages 874110–1 – 874110–8, 2013. 24, 58, 60
- [100] Robot vulnerability database (rvd). https://github.com/aliasrobotics/ RVD/, 2020. 24, 64
- [101] Victor Mayoral Vilches, Lander Usategui San Juan, Bernhard Dieber, Unai Ayucar Carbajo, and Endika Gil-Uriarte. Introducing the robot vulnerability database (RVD). CoRR, abs/1912.11299, 2019. URL http: //arxiv.org/abs/1912.11299. 24
- [102] Keywhan Chung, Xiao Li, Peicheng Tang, Zeran Zhu, Zbigniew T. Kalbarczyk, Ravishankar K. Iyer, and Thenkurussi Kesavadas. Smart malware that uses leaked control data of robotic applications: The case of raven-ii surgical robots. In 22nd International Symposium on Research in Attacks, Intrusions and Defenses (RAID 2019), pages 337–351, Chaoyang District, Beijing, September 2019. USENIX Association. ISBN 978-1-939133-07-6. 25, 55
- [103] Víctor Mayoral-Vilches, Martin Pinzger, Stefan Rass, Bernhard Dieber, and Endika Gil-Uriarte. Can ros be used securely in industry? red teaming rosindustrial, 2020. 25
- [104] Ros 2 robotic systems threat model. https://design.ros2.org/articles/ ros2_threat_model.html, 2020. 25
- [105] Jyotirmoy V Deshmukh, Alexandre Donzé, Shromona Ghosh, Xiaoqing Jin, Garvit Juniwal, and Sanjit A Seshia. Robust online monitoring of signal temporal logic. *Formal Methods in System Design*, 51(1):5–30, 2017. 27

- [106] Dejan Ničković and Tomoya Yamaguchi. Rtamt: Online robustness monitors from stl. In International Symposium on Automated Technology for Verification and Analysis, pages 564–571. Springer, 2020. 27, 28, 34
- [107] Dariush Forouher, Jan Hartmann, and Erik Maehle. Data flow analysis in ros. In ISR/Robotik 2014; 41st International Symposium on Robotics, pages 1-6, 2014. 30
- [108] Sebastian Buck, Richard Hanten, C. Robert Pech, and Andreas Zell. Synchronous dataflow and visual programming for prototyping robotic algorithms. In Weidong Chen, Koh Hosoda, Emanuele Menegatti, Masahiro Shimizu, and Hesheng Wang, editors, *Intelligent Autonomous Systems 14*, pages 911–923, Cham, 2017. Springer International Publishing. ISBN 978-3-319-48036-7. 30
- [109] Donghwa Jeong and Kiju Lee. Inchbot: A novel swarm microrobotic platform. In 2013 IEEE/RSJ International Conference on Intelligent Robots and Systems, pages 5565–5570, 2013. 39, 44
- [110] Zhi Yan, Luc Fabresse, Jannik Laval, and Noury Bouraqadi. Metrics for performance benchmarking of multi-robot exploration. In 2015 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pages 3407–3414, 2015. 40
- [111] Jiří Hörner. Map-merging for multi-robot system, 2016. URL https://is. cuni.cz/webapps/zzp/detail/174125/. 42
- [112] Zhi Yan, Luc Fabresse, Jannik Laval, and Noury Bouraqadi. Team size optimization for multi-robot exploration. In In Proceedings of the 4th International Conference on Simulation, Modeling, and Programming for Autonomous Robots (SIMPAR 2014), pages 438–449, Bergamo, Italy, October 2014. 42
- [113] Ros-Visualization. Ros rviz 3d visualizer. URL https://github.com/ ros-visualization/rviz. 43
- [114] Raspberry Pi. URL https://www.raspberrypi.org/. 51
- [115] Turtlebot3 lds-01 laser sensor, 2021. URL https://emanual.robotis.com/ docs/en/platform/turtlebot3/appendix_lds_01/. 51
- [116] David Saldana, Amanda Prorok, Shreyas Sundaram, Mario FM Campos, and Vijay Kumar. Resilient consensus for time-varying networks of dynamic agents. In 2017 American Control Conference (ACC), pages 252–258, 2017. 53
- [117] Zohir Bouzid, Maria Gradinariu Potop-Butucaru, and Sébastien Tixeuil. Byzantine-resilient convergence in oblivious robot networks. In Vijay Garg, Roger Wattenhofer, and Kishore Kothapalli, editors, *Distributed Computing* and Networking, pages 275–280, Berlin, Heidelberg, 2009. Springer Berlin Heidelberg. ISBN 978-3-540-92295-7. 54

- [118] Ichiro Suzuki and Masafumi Yamashita. Distributed anonymous mobile robots: Formation of geometric patterns. SIAM J. Comput., 28:1347–1363, 1999. 54
- [119] Paola Flocchini, Giuseppe Prencipe, Nicola Santoro, and Peter Widmayer. Distributed coordination of a set of autonomous mobile robots. In *Proceedings* of the IEEE Intelligent Vehicles Symposium, pages 480–485. IEEE, 2000. 54
- [120] Igor Zikratov, Oleg Maslennikov, Ilya Lebedev, Aleksandr Ometov, and Sergey Andreev. Dynamic trust management framework for robotic multiagent systems. In Olga Galinina, Sergey Balandin, and Yevgeni Koucheryavy, editors, *Internet of Things, Smart Spaces, and Next Generation Networks* and Systems, pages 339–348, Cham, 2016. Springer International Publishing. ISBN 978-3-319-46301-8. 54
- [121] Marcello Pogliani, Federico Maggi, Marco Balduzzi, Davide Quarta, and Stefano Zanero. Detecting insecure code patterns in industrial robot programs. In Proceedings of the 15th ACM Asia Conference on Computer and Communications Security, pages 759–771, 2020. 55
- [122] Daniel S Fowler, Jeremy Bryans, Siraj Ahmed Shaikh, and Paul Wooderson. Fuzz testing for automotive cyber-security. In 2018 48th Annual IEEE/IFIP International Conference on Dependable Systems and Networks Workshops (DSN-W), pages 239–246. IEEE, 2018. 55
- [123] Guanpeng Li, Yiran Li, Saurabh Jha, Timothy Tsai, Michael Sullivan, Siva Kumar Sastry Hari, Zbigniew Kalbarczyk, and Ravishankar Iyer. Av-fuzzer: Finding safety violations in autonomous driving systems. In 2020 IEEE 31st International Symposium on Software Reliability Engineering (ISSRE), pages 25–36. IEEE, 2020. 55
- [124] Jia Cheng Han and Zhi Quan Zhou. Metamorphic fuzz testing of autonomous vehicles. In Proceedings of the IEEE/ACM 42nd International Conference on Software Engineering Workshops, pages 380–385, 2020.
- [125] Tommaso Dreossi, Daniel J Fremont, Shromona Ghosh, Edward Kim, Hadi Ravanbakhsh, Marcell Vazquez-Chanlatte, and Sanjit A Seshia. Verifai: A toolkit for the formal design and analysis of artificial intelligence-based systems. In *International Conference on Computer Aided Verification*, pages 432–442. Springer, 2019. 55
- [126] Nils Ole Tippenhauer, Christina Pöpper, Kasper Bonne Rasmussen, and Srdjan Capkun. On the requirements for successful gps spoofing attacks. In ACM Conference on Computer and Communications Security (CCS), pages 75–86, 2011. 55
- [127] NTyler Nighswander, Brent M. Ledvina, Jonathan Diamond, Robert Brumley, and David Brumley. Gps software attacks. In ACM Conference on Computer and Communications Security (CCS), pages 450–461, 2012.

- [128] Kexiong (Curtis) Zeng, Shinan Liu, Yuanchao Shu, Dong Wang, Haoyu Li, Yanzhi Dou, Gang Wang, and Yaling Yang. All your gps are belong to us: Towards stealthy manipulation of road navigation systems. In USENIX Security Symposium (USENIX Security 18), pages 1527–1544, 2018.
- [129] Jon S Warner and Roger G Johnston. A simple demonstration that the global positioning system (gps) is vulnerable to spoofing. *Journal of Security Administration*, 25(2):19–27, 2002.
- [130] Seong-Hun Seo, Byung-Hyun Lee, Sung-Hyuck Im, and Gyu-In Jee. Effect of spoofing on unmanned aerial vehicle using counterfeited gps signal. *Journal* of Positioning, Navigation, and Timing, 4(2):57–65, 2015. 55
- [131] Hocheol Shin, Dohyun Kim, Yujin Kwon, and Yongdae Kim. Illusion and dazzle: Adversarial optical channel exploits against lidars for automotive applications. In International Workshop on Cryptographic Hardware and Embedded Systems (CHES), pages 445–467, 2017. 55
- [132] Yulong Cao, Chaowei Xiao, Benjamin Cyr, Yimeng Zhou, Won Park, Sara Rampazzi, Qi Alfred Chen, Kevin Fu, and Z. Morley Mao. Adversarial sensor attack on lidar-based perception in autonomous driving. In ACM Conference on Computer and Communications Security (CCS), pages 2267–2281, 2019. 55
- [133] Timothy Trippel, Ofir Weisse, Wenyuan Xu, Peter Honeyman, and Kevin Fu. Walnut: Waging doubt on the integrity of mems accelerometers with acoustic injection attacks. In 2017 IEEE European symposium on security and privacy (EuroS&P), pages 3–18. IEEE, 2017. 55
- [134] Yazhou Tu, Zhiqiang Lin, Insup Lee, and Xiali Hei. Injected and delivered: Fabricating implicit control over actuation systems by spoofing inertial sensors. In USENIX Security Symposium (USENIX Security 18), pages 1545– 1562, 2018. 55
- [135] Kun Cheng, Yuan Zhou, Bihuan Chen, Rui Wang, Yuebin Bai, and Yang Liu. Guardauto: A decentralized runtime protection system for autonomous driving. *IEEE Transactions on Computers*, 70(10):1569–1581, 2020. 58
- [136] Jarrod McClean, Christopher Stull, Charles Farrar, and David Mascareñas. A preliminary cyber-physical security assessment of the Robot Operating System (ROS). In Robert E. Karlsen, Douglas W. Gage, Charles M. Shoemaker, and Grant R. Gerhart, editors, Unmanned Systems Technology XV, volume 8741, pages 341 – 348. International Society for Optics and Photonics, SPIE, 2013. 58, 60
- [137] Sean Rivera and Radu State. Securing robots: An integrated approach for security challenges and monitoring for the robotic operating system (ros). In 2021 IFIP/IEEE International Symposium on Integrated Network Management (IM), pages 754–759. IEEE, 2021. 58, 60

- [138] Petra Mazdin, Michal Barciś, Hermann Hellwagner, and Bernhard Rinner. Distributed task assignment in multi-robot systems based on information utility. In 2020 IEEE 16th International Conference on Automation Science and Engineering (CASE), pages 734–740, 2020. 58
- [139] Andrea Testa, Andrea Camisa, and Giuseppe Notarstefano. Choirbot: A ros 2 toolbox for cooperative robotics. *IEEE Robotics and Automation Letters*, 6(2):2714–2720, 2021.
- [140] Yuan Xu, Gelei Deng, Tianwei Zhang, Han Qiu, and Yungang Bao. Novel denial-of-service attacks against cloud-based multi-robot systems. *Information Sciences*, 576:329-344, 2021. ISSN 0020-0255. URL https://www. sciencedirect.com/science/article/pii/S002002552100654X. 89
- [141] Misook Kim, SangGyu Kim, ByoungYoul Song, Young-sook Jeong, and Hong Seong Park. Study on requirements of cloud-based environments for easy development of ros modules. In 2021 18th International Conference on Ubiquitous Robots (UR), pages 48–51, 2021.
- [142] Shreyas Gokhale. Jdemultibot: Multi-robot exercises for robotics academy in ros2. 2020.
- [143] Agata Barciś, Michał Barciś, and Christian Bettstetter. Robots that sync and swarm: A proof of concept in ros 2. In 2019 International Symposium on Multi-Robot and Multi-Agent Systems (MRS), pages 98–104, 2019. 58
- [144] Amazon Web Services. AWS Robotics. https://aws.amazon.com/ robomaker/, 2021. 58, 59, 65, 82
- [145] irobot vacuums. https://www.irobot.com/, 2019. 58
- [146] Edmund M Clarke Jr, Orna Grumberg, Daniel Kroening, Doron Peled, and Helmut Veith. Model checking. 2018. 59
- [147] Edmund M. Clarke, William Klieber, Miloš Nováček, and Paolo Zuliani. Model Checking and the State Explosion Problem, pages 1–30. Springer Berlin Heidelberg, 2012. 59
- [148] Fabian Yamaguchi, Nico Golde, Daniel Arp, and Konrad Rieck. Modeling and discovering vulnerabilities with code property graphs. In 2014 IEEE Symposium on Security and Privacy, pages 590–604, 2014. 59, 67, 68
- [149] Ros 2 robotic systems threat model. https://design.ros2.org/articles/ ros2_threat_model.html, 2021. 59, 71, 73
- [150] Adam Barth, Dan Boneh, and Brent Waters. Privacy in encrypted content distribution using private broadcast encryption. In Giovanni Di Crescenzo and Avi Rubin, editors, *Financial Cryptography and Data Security*, pages 52–64, Berlin, Heidelberg, 2006. Springer Berlin Heidelberg. ISBN 978-3-540-46256-9. 60, 77, 83, 84, 93

- [151] On the (in)security of secure ros2. https://sites.google.com/view/ secure-sros2, 2022. 60, 68, 69, 74, 79, 82, 86, 87, 89
- [152] Rafael R. Teixeira, Igor P. Maurell, and Paulo L.J. Drews. Security on ros: analyzing and exploiting vulnerabilities of ros-based systems. In 2020 Latin American Robotics Symposium (LARS), 2020 Brazilian Symposium on Robotics (SBR) and 2020 Workshop on Robotics in Education (WRE), pages 1-6, 2020. 60
- [153] DDS Foundation. Data Distribution Services. https://www. dds-foundation.org/, 2020. 61
- [154] Gerardo Pardo-Castellote, Bert Farabaugh, and Rick Warren. An introduction to dds and data-centric communications. *Real-Time Innovations*, 2005. 61
- [155] Ros2 node to participant mapping, 2020. URL https://design.ros2.org/ articles/Node_to_Participant_mapping.html. 62
- [156] Object Management Group. DDS security [Online]. https://www.omg.org/ spec/DDS-SECURITY/1.1/PDF, 2018. 62
- [157] Michael Myers, Rich Ankney, Ambarish Malpani, Slava Galperin, and Carlisle Adams. X. 509 internet public key infrastructure online certificate status protocol-ocsp. 1999. 62
- [158] Joseph Salowey, Abhijit Choudhury, and David McGrew. Aes galois counter mode (gcm) cipher suites for tls. *Request for Comments*, 5288, 2008. 63
- [159] ROS2. SROS2. https://github.com/ros2/sros2, 2021. 63, 64, 72
- [160] Ken Conley. Ros command line tool: rostopic. http://library.isr.ist. utl.pt/docs/roswiki/rostopic.html, 2011. 63, 78
- [161] Hakan Lindqvist. Mandatory access control. Master's thesis in computing science, Umea University, Department of Computing Science, SE-901, 87, 2006. 64
- [162] Gelei Deng, Yuan Zhou, Yuan Xu, Tianwei Zhang, and Yang Liu. An investigation of byzantine threats in multi-robot systems. In 24th International Symposium on Research in Attacks, Intrusions and Defenses, pages 17–32, 2021. 64
- [163] Shengtuo Hu, Qi Alfred Chen, Jiachen Sun, Yiheng Feng, Z. Morley Mao, and Henry X. Liu. Automated discovery of denial-of-service vulnerabilities in connected vehicle protocols. In 30th USENIX Security Symposium (USENIX Security 21), pages 3219–3236, August 2021. ISBN 978-1-939133-24-3. 73, 93

- [164] Qi Alfred Chen, Yucheng Yin, Yiheng Feng, Z Morley Mao, and Henry X Liu. Exposing congestion attack on emerging connected vehicle based traffic signal control. In 25th Annual Network and Distributed System Security Symposium (NDSS), 2018. 64
- [165] Yuan Xu, Tianwei Zhang, and Yungang Bao. Analysis and mitigation of function interaction risks in robot apps. 24th International Symposium on Research in Attacks, Intrusions and Defenses, 2021. 65
- [166] Cloud robotics core: Kubernetes, federation, app management [online]. https://googlecloudrobotics.github.io/core/, 2022. 65
- [167] Jun Sun, Yang Liu, Jin Song Dong, and Chunqing Chen. Integrating specification and programs for system modeling and verification. In 2009 Third IEEE International Symposium on Theoretical Aspects of Software Engineering, pages 127–135, 2009. 65, 69
- [168] Alireza Souri, Amir Masoud Rahmani, Nima Jafari Navimipour, and Reza Rezaei. A symbolic model checking approach in formal verification of distributed systems. *Human-centric Computing and Information Sciences*, 9(1), 2019. doi: 10.1186/s13673-019-0165-x. 69
- [169] Huiquan Zhu, Jing Sun, Jin Song Dong, and Shang-Wei Lin. From verified model to executable program: The pat approach. *Innovations in Systems* and Software Engineering, 12(1):1–26, 2015. 69
- [170] Charles Antony Richard Hoare. Communicating sequential processes. Communications of the ACM, 21(8):666–677, 1978. 69
- [171] Peter Chen, Marjon Dean, Don Ojoko-Adams, Hassan Osman, Lilian Lopez, Nick Xie, and Nancy Mead. System quality requirements engineering (square) methodology: Case study on asset management system. page 326, 12 2004.
 71
- [172] Amir Pnueli. The temporal logic of programs. In 18th Annual Symposium on Foundations of Computer Science (sfcs 1977), pages 46–57, 1977. 71
- [173] Jun Sun, Yang Liu, and Jin Song Dong. Model checking csp revisited: Introducing a process analysis toolkit. In Tiziana Margaria and Bernhard Steffen, editors, *Leveraging Applications of Formal Methods, Verification and Validation*, pages 307–322, Berlin, Heidelberg, 2008. Springer Berlin Heidelberg. 72
- [174] Krzysztof R. Apt and Dexter Kozen. Limits for automatic verification of finite-state concurrent systems. Inf. Process. Lett., 22(6):307–309, 1986. 73
- [175] Syed Hussain, Omar Chowdhury, Shagufta Mehnaz, and Elisa Bertino. Lteinspector: A systematic approach for adversarial testing of 4g lte. In Network and Distributed Systems Security (NDSS) Symposium 2018, 2018. 73

- [176] Ros metrics, 2022. URL https://metrics.ros.org/rosdistro_ rosdistro.html. 74
- [177] ROS2. ROS 2 Galatic Geochelone. https://docs.ros.org/en/galactic/ index.html, 2021. 74
- [178] ROS2. ROS 2 Foxy Elusor. https://docs.ros.org/en/eloquent/index. html, 2019. 74
- [179] ROS2. ROS 2 Rolling. https://docs.ros.org/en/rolling/index.html, 2022. 76
- [180] Ros-Visualization. Ros-visualization/RQT_GRAPH. https://github.com/ ros-visualization/rqt_graph, 2020. 77
- [181] Object Management Group. The real-time publish-subscribe protocol (rtps) dds - omg [online]. https://www.omg.org/spec/DDSI-RTPS/2.3/Beta1/ PDF, 2018. 77
- [182] ROS2. Set rtps_protection_kind to encrypt by default pull request #171 · ROS2/SROS2. https://github.com/ros2/sros2/pull/171, 2020. 77
- [183] Murph Finnicum and Samuel T King. Building secure robot applications. In 6th USENIX Workshop on Hot Topics in Security, HotSec'11, 2011. 77
- [184] ROS2. SROS2 Project Sample Policies. https://github.com/ros2/sros2/ tree/master/sros2/sros2/policy, 2020. 77
- [185] OpenRMF. Open-rmf/rmf demos: Demonstrations of the openrmf software [online], 2022. URL https://github.com/open-rmf/rmf_demos. 78, 79
- [186] CVE-2019-19625 Detail. https://nvd.nist.gov/vuln/detail/ CVE-2019-19625, 2019. 81
- [187] Víctor Vilches, Gorka Olalde, Xabier Baskaran, Alejandro Cordero, Lander Juan, Endika Gil-Uriarte, Odei Urabain, and Laura Kirschgens. Aztarna, a footprinting tool for robots, 12 2018. 81
- [188] Key-whan Chung, Xiao Li, Peicheng Tang, Zeran Zhu, Zbigniew T. Kalbarczyk, Ravishankar K. Iyer, and Thenkurussi Kesavadas. Smart malware that uses leaked control data of robotic applications: The case of raven-ii surgical robots. In 22nd International Symposium on Research in Attacks, Intrusions and Defenses, RAID 2019, Chaoyang District, Beijing, China, September 23-25, 2019, 2019. 81
- [189] Eduard Fosch-Villaronga and Tobias Mahler. Cybersecurity, safety and robots: Strengthening the link between cybersecurity and safety in the context of care robots. *Computer Law & Security Review*, 41:105528, 2021. 81

- [190] Bernd Carsten Stahl and Mark Coeckelbergh. Ethics of healthcare robotics: Towards responsible research and innovation. *Robotics and Autonomous Sys*tems, 86:152–161, 2016. 81
- [191] Ryan Shah and Shishir Nagaraja. Privacy with surgical robotics: Challenges in applying contextual privacy theory. CoRR, abs/1909.01862, 2019. URL http://arxiv.org/abs/1909.01862. 81
- [192] AWS-Robotics. Aws-Robotics/AWS RoboMaker Small Warehouse World, 2021. URL https://github.com/aws-robotics/ aws-robomaker-small-warehouse-world. 82
- [193] JdeRobot. JdeRobot: Open toolkit for developing Robotics applications. https://jderobot.github.io/, 2020. 82
- [194] Shashank Agrawal and Melissa Chase. Fame: Fast attribute-based message encryption. In Proceedings of the 2017 ACM SIGSAC Conference on Computer and Communications Security, CCS '17, page 665–682, New York, NY, USA, 2017. Association for Computing Machinery. ISBN 9781450349468. 83
- [195] Brent Waters. Ciphertext-policy attribute-based encryption: An expressive, efficient, and provably secure realization. In Dario Catalano, Nelly Fazio, Rosario Gennaro, and Antonio Nicolosi, editors, *Public Key Cryptography – PKC 2011*, pages 53–70, Berlin, Heidelberg, 2011. Springer Berlin Heidelberg. ISBN 978-3-642-19379-8.
- [196] Jie Chen, Romain Gay, and Hoeteck Wee. Improved dual system abe in prime-order groups via predicate encodings. In Elisabeth Oswald and Marc Fischlin, editors, Advances in Cryptology - EUROCRYPT 2015, pages 595– 624, Berlin, Heidelberg, 2015. Springer Berlin Heidelberg. 83
- [197] Hwimin Kim, Dae-Kyoo Kim, and Alaa Alaerjan. Abac-based security model for dds. *IEEE Transactions on Dependable and Secure Computing*, 2021. 83
- [198] Romain Gay. A new paradigm for public-key functional encryption for degree-2 polynomials. In *Public Key Cryptography (1)*, pages 95–120, 2020. 84
- [199] Sanjam Garg and Mohammad Hajiabadi. Trapdoor functions from the computational diffie-hellman assumption. In Annual International Cryptology Conference, pages 362–391. Springer, 2018. 84
- [200] Wouter Castryck, Jana Sotáková, and Frederik Vercauteren. Breaking the decisional diffie-hellman problem for class group actions using genus theory. In Annual International Cryptology Conference, pages 92–120. Springer, 2020. 84
- [201] B. Blanchet. An efficient cryptographic protocol verifier based on prolog rules. In *Proceedings. 14th IEEE Computer Security Foundations Workshop*, 2001., pages 82–96, 2001. 88

- [202] Elena Basan, Mikhail Medvedev, and Stanislav Teterevyatnikov. Analysis of the impact of denial of service attacks on the group of robots. In 2018 International Conference on Cyber-Enabled Distributed Computing and Knowledge Discovery (CyberC), pages 63–638, 2018. 89
- [203] ApexAI ROS2 Performance Test [Online]. https://gitlab.com/ApexAI/ performance_test, 2021. 89
- [204] Apex ai: The vehicle os company. https://www.apex.ai/, 2022. 89
- [205] Supporting navigation in multi-robot systems through delay tolerant network communication. *IFAC Proceedings Volumes*, 42(22):25–30, 2009. ISSN 1474-6670. 1st IFAC Workshop on Networked Robotics. 90
- [206] Yuya Maruyama, S. Kato, and Takuya Azumi. Exploring the performance of ros2. 2016 International Conference on Embedded Software (EMSOFT), pages 1–10, 2016. 91
- [207] Yizhe Zhang, Lianjun Li, Michael Ripperger, Jorge Nicho, Malathi Veeraraghavan, and Andrea Fumagalli. Gilbreth: A conveyor-belt based pickand-sort industrial robotics application. In 2018 Second IEEE International Conference on Robotic Computing (IRC), pages 17–24, 2018. 92
- [208] Ronald W Ritchey and Paul Ammann. Using model checking to analyze network vulnerabilities. In Proceeding 2000 IEEE Symposium on Security and Privacy. S&P 2000, pages 156–165. IEEE, 2000. 93
- [209] Dang Tu Nguyen, Chengyu Song, Zhiyun Qian, Srikanth V Krishnamurthy, Edward JM Colbert, and Patrick McDaniel. Iotsan: Fortifying the safety of iot systems. In Proceedings of the 14th International Conference on emerging Networking EXperiments and Technologies, pages 191–203, 2018.
- [210] Zuohua Ding, Yuan Zhou, and MengChu Zhou. Stability analysis of switched fuzzy systems via model checking. *IEEE Transactions on Fuzzy Systems*, 22 (6):1503–1514, 2014.
- [211] Zuohua Ding, Yuan Zhou, Mingyue Jiang, and MengChu Zhou. A new class of petri nets for modeling and property verification of switched stochastic systems. *IEEE Transactions on Systems, Man, and Cybernetics: Systems*, 45(7):1087–1100, 2014. 93
- [212] Renato Carvalho, Alcino Cunha, Nuno Macedo, and André Santos. Verification of system-wide safety properties of ROS applications. In 2020 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pages 7249–7254, 2020. 93
- [213] Chi Hu, Wei Dong, Yonghui Yang, Hao Shi, and Ge Zhou. Runtime verification on hierarchical properties of ROS-based robot swarms. *IEEE Transactions on Reliability*, 69(2):674–689, 2019. 93

- [214] Yanan Liu, Yong Guan, Xiaojuan Li, Rui Wang, and Jie Zhang. Formal analysis and verification of DDS in ROS2. In 2018 16th ACM/IEEE International Conference on Formal Methods and Models for System Design (MEMOCODE), pages 1–5, 2018. 93, 94
- [215] Amit Sahai and Brent Waters. Fuzzy identity-based encryption. In Ronald Cramer, editor, Advances in Cryptology – EUROCRYPT 2005, pages 457– 473, 2005. 94
- [216] John Bethencourt, Amit Sahai, and Brent Waters. Ciphertext-policy attribute-based encryption. In 2007 IEEE Symposium on Security and Privacy (SP '07), pages 321–334, 2007. 94
- [217] Vipul Goyal, Abhishek Jain, Omkant Pandey, and Amit Sahai. Bounded ciphertext policy attribute based encryption. In Luca Aceto, Ivan Damgård, Leslie Ann Goldberg, Magnús M. Halldórsson, Anna Ingólfsdóttir, and Igor Walukiewicz, editors, Automata, Languages and Programming, pages 579– 591, Berlin, Heidelberg, 2008. Springer Berlin Heidelberg. ISBN 978-3-540-70583-3.
- [218] Allison Lewko, Tatsuaki Okamoto, Amit Sahai, Katsuyuki Takashima, and Brent Waters. Fully secure functional encryption: Attribute-based encryption and (hierarchical) inner product encryption. In Henri Gilbert, editor, Advances in Cryptology – EUROCRYPT 2010, pages 62–91, Berlin, Heidelberg, 2010. Springer Berlin Heidelberg. ISBN 978-3-642-13190-5. 94
- [219] Shucheng Yu, Cong Wang, Kui Ren, and Wenjing Lou. Attribute based data sharing with attribute revocation. In *Proceedings of the 5th ACM Sympo*sium on Information, Computer and Communications Security, ASIACCS '10, page 261–270, New York, NY, USA, 2010. Association for Computing Machinery. ISBN 9781605589367. 94
- [220] Ruffin White, Henrik I. Christensen, Gianluca Caiazza, and Agostino Cortesi. Procedurally provisioned access control for robotic systems. In 2018 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), 2018. 94
- [221] Yi Liu. Restinfer: automated inferring parameter constraints from natural language restful API descriptions. In Abhik Roychoudhury, Cristian Cadar, and Miryung Kim, editors, Proceedings of the 30th ACM Joint European Software Engineering Conference and Symposium on the Foundations of Software Engineering, ESEC/FSE 2022, Singapore, Singapore, November 14-18, 2022, pages 1816–1818. ACM, 2022. doi: 10.1145/3540250.3559078. URL https://doi.org/10.1145/3540250.3559078. 96
- [222] Yi Liu. Restcluster: Automated crash clustering for restful API. In 37th IEEE/ACM International Conference on Automated Software Engineering, ASE 2022, Rochester, MI, USA, October 10-14, 2022, pages 198:1–198:3.

ACM, 2022. doi: 10.1145/3551349.3559511. URL https://doi.org/10.1145/3551349.3559511.96

- [223] Cloud apis, 2021. https://cloud.google.com/apis. 96
- [224] Azure rest api reference, 2021. https://learn.microsoft.com/enus/rest/api/azure/. 96
- [225] Rest api handbook, 2021. https://developer.wordpress.org/rest-api/. 96
- [226] The state of api security q1 2021, 2021. https://www.securitymagazine.com/articles/94509-of-organizations-hadapi-security-incident-last-year. 96
- [227] Google cloud penetration testing: A complete guide, 2021. https://www.getastra.com/blog/security-audit/google-cloud-penetrationtesting/. 96
- [228] Penetration testing, 2021. https://learn.microsoft.com/enus/azure/security/fundamentals/pen-testing. 96
- [229] The OWASP Zed Attack Proxy (ZAP). https://www.zaproxy.org/, journal=OWASP ZAP, 2021. 96, 98, 116
- [230] w3af: Open Source Web Application Security Scanner. https://w3af.org/, 2018. 96, 98, 116
- [231] François Gauthier, Behnaz Hassanshahi, Ben Selwyn-Smith, Trong Nhan Mai, Maximilian Schlüter, and Micah Williams. Backrest: A model-based feedback-driven greybox fuzzer for web applications. ArXiv, abs/2108.08455, 2021. 96
- [232] OWASP API Security project. https://owasp.org/ www-project-api-security/. 96
- [233] How to test a REST API. https://docs.rapid7.com/appspider/ how-to-test-a-rest-api/, 2018. 96
- [234] V. Atlidakis, P. Godefroid, and M. Polishchuk. Restler: Stateful rest api fuzzing. In 2019 IEEE/ACM 41st International Conference on Software Engineering (ICSE), pages 748–758, 2019. doi: 10.1109/ICSE.2019.00083. 97, 98, 108, 112, 116, 126
- [235] E. Viglianisi, M. Dallago, and M. Ceccato. Resttestgen: Automated blackbox testing of restful apis. In 2020 IEEE 13th International Conference on Software Testing, Validation and Verification (ICST), pages 142–152, Los Alamitos, CA, USA, oct 2020. IEEE Computer Society. doi: 10. 1109/ICST46399.2020.00024. URL https://doi.ieeecomputersociety. org/10.1109/ICST46399.2020.00024. 126

- [236] Y. Liu, Y. Li, G. Deng, Y. Liu, R. Wan, R. Wu, D. Ji, S. Xu, and M. Bao. Morest: Model-based restful api testing with execution feedback. In 2022 IEEE/ACM 44th International Conference on Software Engineering (ICSE), pages 1406-1417, Los Alamitos, CA, USA, 2022. IEEE Computer Society. doi: 10.1145/3510003.3510133. URL https://doi.ieeecomputersociety. org/10.1145/3510003.3510133. 97, 98, 108, 112, 116
- [237] OpenAPI. OpenAPI Specification. https://swagger.io/specification/, 2020. 97, 99
- [238] Microsoft. Cloud computing services: Microsoft azure, 2022. URL https: //azure.microsoft.com/. 98, 111
- [239] Atlassian. Confluence: Your remote-friendly team workspace. https://www. atlassian.com/software/confluence, 2022. 98, 125
- [240] Roy Thomas Fielding and Richard N. Taylor. Architectural Styles and the Design of Network-Based Software Architectures. PhD thesis, 2000. AAI9980887. 99
- [241] Fun & Flexible Software for Online Communities, Teams, and Groups. https://buddypress.org/, 2021. 99
- [242] Fast, Secure Managed WordPress Hosting. https://wordpress.com/, 2021. 99
- [243] MITRE. CVE Project by MITRE. https://cve.mitre.org/cve/search_ cve_list.html, 2021. 101
- [244] Offensive Security. Exploit Database. https://www.exploit-db.com/, 2021. 101
- [245] MITRE. Common Weakness Enumeration (CWE). https://cwe.mitre. org/index.html, 2021. 101
- [246] MITRE. CVE-2021-21389, 2021. https://cve.mitre.org/cgibin/cvename.cgi?name=CVE-2021-21389. 103
- [247] D.P. Lopresti. Robust retrieval of noisy text. In Proceedings of the Third Forum on Research and Technology Advances in Digital Libraries,, pages 76-85, 1996. doi: 10.1109/ADL.1996.502518. 109
- [248] Vladimir I. Levenshtein. Binary codes capable of correcting deletions, insertions, and reversals. Soviet physics. Doklady, 10:707–710, 1965. 109
- [249] OpenAPI Specification Version 3. https://swagger.io/specification/. 111
- [250] Fuzzdb-Project. FuzzDB: Dictionary of attack patterns and primitives for black-box application fault injection and resource discovery.
 =https://github.com/fuzzdb-project/fuzzdb, 2022. 116

- [251] Manuel Leithner, Bernhard Garn, and Dimitris E. Simos. Hydra: Feedbackdriven black-box exploitation of injection vulnerabilities. *Information and Software Technology*, 140:106703, 2021. ISSN 0950-5849. doi: https://doi. org/10.1016/j.infsof.2021.106703. 116
- [252] Bernhard Garn, Jovan Zivanovic, Manuel Leithner, and Dimitris E Simos. Combinatorial methods for dynamic gray-box sql injection testing. Software Testing, Verification and Reliability, 32(6):e1826, 2022. 116
- [253] OWASP. Nodegoat: The owasp nodegoat project. https://github.com/ OWASP/NodeGoat, 2021. 116, 117
- [254] erev0s. Erev0s/VAmPI: Vulnerable REST API with OWASP top 10 Vulnerabilities for Security Testing. https://github.com/erev0s/VAmPI, 2020. 117
- [255] Anonymous Submission Website for Nautilus. https://sites.google.com/ view/nautilus-testing. 117, 119, 120, 123
- [256] SEO Panel: World's first SEO Control Panel for Multiple Websites. https: //www.seopanel.org/, 2022. 117
- [257] Navigate CMS. https://www.navigatecms.com/en/home, 2022. 117
- [258] Gila cms. https://gilacms.com/, 2022. 117, 124
- [259] András Vargha and Harold D. Delaney. A critique and improvement of the cl common language effect size statistics of mcgraw and wong. *Journal of Educational and Behavioral Statistics*, 25(2):101–132, 2000. doi: 10.3102/ 10769986025002101. 118
- [260] Atlassian. Atlassian: Software development and collaboration tools. https://www.atlassian.com/, 2022. 125
- [261] Yan Zheng, Yi Liu, Xiaofei Xie, Yepang Liu, Lei Ma, Jianye Hao, and Yang Liu. Automatic web testing using curiosity-driven reinforcement learning. In 43rd IEEE/ACM International Conference on Software Engineering, ICSE 2021, Madrid, Spain, 22-30 May 2021, pages 423–435. IEEE, 2021. doi: 10. 1109/ICSE43902.2021.00048. URL https://doi.org/10.1109/ICSE43902. 2021.00048. 126
- [262] Andrea Arcuri. Restful api automated test case generation with evomaster. *ACM Trans. Softw. Eng. Methodol.*, 28(1), January 2019. ISSN 1049-331X. doi: 10.1145/3293455. URL https://doi.org/10.1145/3293455. 126
- [263] Bernardo Guimaraes and Miroslav Stampar. sqlmap: Automatic SQL injection and database takeover tool. https://sqlmap.org/, 2022. 126, 163
- [264] s0md3v. S0md3v/xsstrike: Most advanced xss scanner. https://github. com/s0md3v/XSStrike, 2022. 126, 163

- [265] Yan Shoshitaishvili, Michael Weissbacher, Lukas Dresel, Christopher Salls, Ruoyu Wang, Christopher Kruegel, and Giovanni Vigna. Rise of the hacrs: Augmenting autonomous cyber reasoning systems with human assistance. In Proceedings of the 2017 ACM SIGSAC Conference on Computer and Communications Security, CCS '17, page 347–362, New York, NY, USA, 2017. Association for Computing Machinery. ISBN 9781450349468. doi: 10.1145/ 3133956.3134105. URL https://doi.org/10.1145/3133956.3134105. 127
- [266] Cornelius Aschermann, Sergej Schumilo, Ali Abbasi, and Thorsten Holz. Ijon: Exploring deep state spaces via fuzzing. In 2020 IEEE Symposium on Security and Privacy (SP), pages 1597–1612, 2020. doi: 10.1109/SP40000.2020. 00117. 127
- [267] Iz Beltagy, Kyle Lo, and Arman Cohan. Scibert: A pretrained language model for scientific text. In *EMNLP*, 2019. 130
- [268] Ying Zhang, Wenjia Song, Zhengjie Ji, Danfeng, Yao, and Na Meng. How well does llm generate security tests?, 2023.
- [269] Gelei Deng, Yi Liu, Víctor Mayoral-Vilches, Peng Liu, Yuekang Li, Yuan Xu, Tianwei Zhang, Yang Liu, Martin Pinzger, and Stefan Rass. Pentestgpt: An llm-empowered automatic penetration testing tool, 2023. 130
- [270] Enkelejda Kasneci, Kathrin Sessler, Stefan Küchemann, Maria Bannert, Daryna Dementieva, Frank Fischer, Urs Gasser, Georg Groh, Stephan Günnemann, Eyke Hüllermeier, Stepha Krusche, Gitta Kutyniok, Tilman Michaeli, Claudia Nerdel, Jürgen Pfeffer, Oleksandra Poquet, Michael Sailer, Albrecht Schmidt, Tina Seidel, Matthias Stadler, Jochen Weller, Jochen Kuhn, and Gjergji Kasneci. Chatgpt for good? on opportunities and challenges of large language models for education. *Learning and Individual Differences*, 103:102274, 2023. 130
- [271] Anees Merchant. How Large Language Models are Shaping the Future of Journalism. https://www.aneesmerchant.com/personal-musings/ how-large-language-models-are-shaping-the-future-of-journalism.
- [272] Sung Kim. Writing a Film Script Using AI OpenAI ChatGPT. https://medium.com/geekculture/ writing-a-film-script-using-ai-openai-chatgpt-e339fe498fc9.
- [273] Ann Yuan, Andy Coenen, Emily Reif, and Daphne Ippolito. Wordcraft: Story writing with large language models. In *IUI*, page 841–852, 2022. 130
- [274] Yi Liu, Gelei Deng, Zhengzi Xu, Yuekang Li, Yaowen Zheng, Ying Zhang, Lida Zhao, Tianwei Zhang, and Yang Liu. Jailbreaking chatgpt via prompt engineering: An empirical study, 2023. 130, 135, 137, 138, 151, 161, 162

[275]	Haoran Li, Da	di Guo, We	ei Fan,	Mingshi	Xu, Jie	e Huang,	Fanpu	Meng,	and
	Yangqiu Song.	Multi-step	Jailbr	eaking P	rivacy A	Attacks or	n Chat	GPT, 2	023.
	135, 137, 162								

- [276] Yotam Wolf, Noam Wies, Yoav Levine, and Amnon Shashua. Fundamental limitations of alignment in large language models. *arXiv preprint*, 2023. 162
- [277] Murray Shanahan, Kyle McDonell, and Laria Reynolds. Role-play with large language models. *arXiv preprint*, 2023. 130, 161
- [278] OpenAI. Creating safe AGI that benefits all of humanity. https://openai. com, . 130
- [279] OpenAI. Moderation OpenAI API. https://platform.openai.com/docs/ guides/moderation, 130, 133, 137
- [280] Baidu. ERNIE Titan LLM. https://gpt3demo.com/apps/ erinie-titan-llm-baidu. 132, 137
- [281] Anthropic. Introducing Claude. https://www.anthropic.com/index/ introducing-claude. 133
- [282] Google. Google AI Principles. URL https://ai.google/responsibility/ principles/. 133
- [283] Microsoft. URL https://www.bing.com/new/termsofuse. 133, 137
- [284] OpenAI. API to Prevent Prompt Injection & Jailbreaks. https://community.openai.com/t/ api-to-prevent-prompt-injection-jailbreaks/203514/2. 135
- [285] Jailbreak chat. https://www.jailbreakchat.com/. 138, 152
- [286] OpenAI. https://platform.openai.com/examples. 142
- [287] Israel Cohen, Yiteng Huang, Jingdong Chen, Jacob Benesty, Jacob Benesty, Jingdong Chen, Yiteng Huang, and Israel Cohen. Pearson correlation coefficient. Noise reduction in speech processing, pages 1–4, 2009. 142
- [288] DN Lawley. A Generalization of Fisher's Z Test. Biometrika, 30(1/2):180– 187, 1938. 145
- [289] Friedrich L. Bauer. Cæsar Cipher, pages 180–180. Springer US, Boston, MA, 2011. ISBN 978-1-4419-5906-5. doi: 10.1007/978-1-4419-5906-5_162. URL https://doi.org/10.1007/978-1-4419-5906-5_162. 150
- [290] Xuanqi Liu and Zhuotao Liu. LLMs Can Understand Encrypted Prompt: Towards Privacy-Computing Friendly Transformers, 2023. 150

- [291] Hugo Touvron, Thibaut Lavril, Gautier Izacard, Xavier Martinet, Marie-Anne Lachaux, Timothée Lacroix, Baptiste Rozière, Naman Goyal, Eric Hambro, Faisal Azhar, Aurelien Rodriguez, Armand Joulin, Edouard Grave, and Guillaume Lample. Llama: Open and efficient foundation language models, 2023. 153
- [292] Rohan Taori, Ishaan Gulrajani, Tianyi Zhang, Yann Dubois, Xuechen Li, Carlos Guestrin, Percy Liang, and Tatsunori B Hashimoto. Stanford alpaca: An instruction-following llama model, 2023. 153
- [293] Wei-Lin Chiang, Zhuohan Li, Zi Lin, Ying Sheng, Zhanghao Wu, Hao Zhang, Lianmin Zheng, Siyuan Zhuang, Yonghao Zhuang, Joseph E. Gonzalez, Ion Stoica, and Eric P. Xing. Vicuna: An open-source chatbot impressing gpt-4 with 90%* chatgpt quality, March 2023. URL https://lmsys.org/blog/ 2023-03-30-vicuna/. 155, 156
- [294] Baidu. Ernie. https://yiyan.baidu.com/welcome. 160
- [295] Wangmeng Xiang, Chao Li, Yuxuan Zhou, Biao Wang, and Lei Zhang. Language Supervised Training for Skeleton-based Action Recognition, 2022. 161
- [296] Thijs Van Ede, Hojjat Aghakhani, Noah Spahn, Riccardo Bortolameotti, Marco Cova, Andrea Continella, Maarten van Steen, Andreas Peter, Christopher Kruegel, and Giovanni Vigna. Deepcase: Semi-supervised Contextual Analysis of Security Events. In *IEEE S&P*, pages 522–539, 2022. 161
- [297] JD Zamfirescu-Pereira, Richmond Y Wong, Bjoern Hartmann, and Qian Yang. Why Johnny Can't Prompt: How Non-AI Experts Try (and fail) to Design LLM Prompts. In CHI, pages 1–21, 2023. 161
- [298] Yongchao Zhou, Andrei Ioan Muresanu, Ziwen Han, Keiran Paster, Silviu Pitis, Harris Chan, and Jimmy Ba. Large Language Models are Human-level Prompt Engineers. arXiv preprint, 2022.
- [299] Reid Pryzant, Dan Iter, Jerry Li, Yin Tat Lee, Chenguang Zhu, and Michael Zeng. Automatic Prompt Optimization with Gradient Descent and Beam Search. arXiv preprint, 2023.
- [300] Xinyi Hou, Yanjie Zhao, Yue Liu, Zhou Yang, Kailong Wang, Li Li, Xiapu Luo, David Lo, John Grundy, and Haoyu Wang. Large language models for software engineering: A systematic literature review, 2023. 161
- [301] Jonas Oppenlaender, Rhema Linder, and Johanna Silvennoinen. Prompting AI Art: An Investigation into the Creative Skill of Prompt Engineering. *arXiv preprint*, 2023. 161
- [302] Jules White, Quchen Fu, Sam Hays, Michael Sandborn, Carlos Olea, Henry Gilbert, Ashraf Elnashar, Jesse Spencer-Smith, and Douglas C Schmidt. A Prompt Pattern Catalog to Enhance Prompt Engineering with ChatGPT. arXiv preprint, 2023.

- [303] Laria Reynolds and Kyle McDonell. Prompt programming for large language models: Beyond the few-shot paradigm. In *CHI EA*, 2021. 161
- [304] Abhinav Rao, Sachin Vashistha, Atharva Naik, Somak Aditya, and Monojit Choudhury. Tricking LLMs into Disobedience: Understanding, Analyzing, and Preventing Jailbreaks. arXiv preprint, 2023. 161
- [305] Wai Man Si, Michael Backes, Jeremy Blackburn, Emiliano De Cristofaro, Gianluca Stringhini, Savvas Zannettou, and Yang Zhang. Why So Toxic?: Measuring and Triggering Toxic Behavior in Open-Domain Chatbots. In CCS, pages 2659–2673, 2022. 161
- [306] Andy Zou, Zifan Wang, Nicholas Carlini, Milad Nasr, J. Zico Kolter, and Matt Fredrikson. Universal and transferable adversarial attacks on aligned language models, 2023. URL https://arxiv.org/abs/2307.15043. 162
- [307] Jiahao Yu, Xingwei Lin, Zheng Yu, and Xinyu Xing. Gptfuzzer: Red teaming large language models with auto-generated jailbreak prompts, 2024. URL https://arxiv.org/abs/2309.10253. 162
- [308] Emily M. Bender, Timnit Gebru, Angelina McMillan-Major, and Shmargaret Shmitchell. On the Dangers of Stochastic Parrots: Can Language Models Be Too Big? In FAccT, pages 610–623. 162
- [309] Weiwei Sun, Zhengliang Shi, Shen Gao, Pengjie Ren, Maarten de Rijke, and Zhaochun Ren. Contrastive Learning Reduces Hallucination in Conversations. arXiv preprint, 2022.
- [310] Potsawee Manakul, Adian Liusie, and Mark JF Gales. Selfcheckgpt: Zeroresource black-box hallucination detection for generative large language models. *arXiv preprint*, 2023.
- [311] Nick McKenna, Tianyi Li, Liang Cheng, Mohammad Javad Hosseini, Mark Johnson, and Mark Steedman. Sources of Hallucination by Large Language Models on Inference Tasks. arXiv preprint, 2023.
- [312] Samuel Gehman, Suchin Gururangan, Maarten Sap, Yejin Choi, and Noah A. Smith. RealToxicityPrompts: Evaluating Neural Toxic Degeneration in Language Models. In *EMNLP*, pages 3356–3369, 2020. 162
- [313] Marco Ramponi. The Full Story of Large Language Models and RLHF. https://www.assemblyai.com/blog/ the-full-story-of-large-language-models-and-rlhf. 162
- [314] Kai Greshake, Sahar Abdelnabi, Shailesh Mishra, Christoph Endres, Thorsten Holz, and Mario Fritz. Not what you've signed up for: Compromising Real-World LLM-Integrated Applications with Indirect Prompt Injection. In arXiv preprint, 2023. 162
- [315] Fábio Perez and Ian Ribeiro. Ignore Previous Prompt: Attack Techniques For Language Models. In NeurIPS ML Safety Workshop, 2022. 162
- [316] Giovanni Apruzzese, Hyrum S. Anderson, Savino Dambra, David Freeman, Fabio Pierazzi, and Kevin A. Roundy. "Real Attackers Don't Compute Gradients": Bridging the Gap between Adversarial ML Research and Practice. In SaTML, 2023.
- [317] Yi Liu, Gelei Deng, Yuekang Li, Kailong Wang, Tianwei Zhang, Yepang Liu, Haoyu Wang, Yan Zheng, and Yang Liu. Prompt injection attack against llm-integrated applications, 2023. 162
- [318] Eugene Bagdasaryan and Vitaly Shmatikov. Spinning Language Models: Risks of Propaganda-As-A-Service and Countermeasures. In S&P, pages 769–786. IEEE, 2022. 162
- [319] Zhiyuan Zhang, Lingjuan Lyu, Xingjun Ma, Chenguang Wang, and Xu Sun. Fine-mixing: Mitigating Backdoors in Fine-tuned Language Models. In *EMNLP*, pages 355–372, 2022.
- [320] Kai Mei, Zheng Li, Zhenting Wang, Yang Zhang, and Shiqing Ma. NO-TABLE: Transferable Backdoor Attacks Against Prompt-based NLP Models. In ACL, 2023. 162
- [321] Ahmed Salem, Michael Backes, and Yang Zhang. Get a Model! Model Hijacking Attack Against Machine Learning Models. In *NDSS*, 2022. 162
- [322] Wai Man Si, Michael Backes, Yang Zhang, and Ahmed Salem. Two-in-One: A Model Hijacking Attack Against Text Generation Models. arXiv preprint, 2023. 162
- [323] Gelei Deng, Zhiyi Zhang, Yuekang Li, Yi Liu, Tianwei Zhang, Yang Liu, Guo Yu, and Dongjin Wang. NAUTILUS: Automated RESTful API Vulnerability Detection. In USENIX Security Symposium, 2023. URL https: //api.semanticscholar.org/CorpusID:260777640. 163
- [324] Yi Liu, Yuekang Li, Gelei Deng, Yang Liu, Ruiyuan Wan, Runchao Wu, Dandan Ji, Shiheng Xu, and Minli Bao. Morest: Model-based RESTful API Testing with Execution Feedback. 2022 IEEE/ACM 44th International Conference on Software Engineering (ICSE), pages 1406–1417, 2022. URL https://api.semanticscholar.org/CorpusID:248392251.
- [325] Vaggelis Atlidakis, Patrice Godefroid, and Marina Polishchuk. RESTler: Stateful REST API Fuzzing. 2019 IEEE/ACM 41st International Conference on Software Engineering (ICSE), pages 748–758, 2019. URL https://api. semanticscholar.org/CorpusID:85465006.
- [326] Emanuele Viglianisi, Michael Dallago, and Mariano Ceccato. RESTTEST-GEN: Automated Black-Box Testing of RESTful APIs. 2020 IEEE 13th International Conference on Software Testing, Validation and Verification

(*ICST*), pages 142-152, 2020. URL https://api.semanticscholar.org/CorpusID:221107287.

- [327] Ying Zhang, Md Mahir Asef Kabir, Ya Xiao, Danfeng Yao, and Na Meng. Automatic detection of java cryptographic api misuses: Are we there yet? *IEEE Transactions on Software Engineering*, 49(1):288–303, 2023. doi: 10. 1109/TSE.2022.3150302.
- [328] Yi Liu. RESTInfer: Automated Inferring Parameter Constraints from Natural Language RESTful API Descriptions. In Proceedings of the 30th ACM Joint European Software Engineering Conference and Symposium on the Foundations of Software Engineering, ESEC/FSE 2022, page 1816–1818, New York, NY, USA, 2022. Association for Computing Machinery. ISBN 9781450394130. doi: 10.1145/3540250.3559078. URL https://doi.org/10. 1145/3540250.3559078.
- [329] Yi Liu, Yuekang Li, Yang Liu, Ruiyuan Wan, Runchao Wu, and Qingkun Liu. Morest: Industry practice of automatic restful api testing. In Proceedings of the 37th IEEE/ACM International Conference on Automated Software Engineering, ASE '22, New York, NY, USA, 2023. Association for Computing Machinery. ISBN 9781450394758. doi: 10.1145/3551349.3559498. URL https://doi.org/10.1145/3551349.3559498.
- [330] Kailong Wang, Junzhe Zhang, Guangdong Bai, Ryan Ko, and Jin Song Dong. It's Not Just the Site, It's the Contents: Intra-domain Fingerprinting Social Media Websites Through CDN Bursts. In 30th The Web Conference (WWW), 2021. doi: 10.1109/ICECCS2018.2018.00011. 163
- [331] Yingji Li, Mengnan Du, Rui Song, Xin Wang, and Ying Wang. A survey on fairness in large language models. arXiv preprint arXiv:2308.10149, 2023. 167
- [332] Yifan Yao, Jinhao Duan, Kaidi Xu, Yuanfang Cai, Eric Sun, and Yue Zhang. A survey on large language model (llm) security and privacy: The good, the bad, and the ugly. arXiv preprint arXiv:2312.02003, 2023. 167